# 1987 MINNESOTA NEW FIRMS STUDY:

BIOII

An Exploration of New Firms and Their Economic Contributions

> Paul Reynolds and Brenda Miller

#### 1987 MINNESOTA NEW FIRMS STUDY:

An Exploration of New Firms and Their Economic Contributions

by

Paul Reynolds and Brenda Miller

Institutional Sponsor:

Center for Urban and Regional Affairs

Funding Provided by:

Metropolitan Council of the Twin Cities Minnesota Chamber of Commerce and Industry City of Saint Paul Saint Paul Port Authority State of Minnesota AVTI State of Minnesota Community Colleges State of Minnesota Department of Trade and Economic Development State of Minnesota State Planning Agency University of Minnesota Center for Urban and Regional Affairs University of Minnesota Extension Service A publication of the Center for Urban and Regional Affairs, 330 Hubert H. Humphrey Center, 301 19th Avenue S., Minneapolis, Minnesota 55455.

The content of this report is the responsibility of the authors and is not necessarily endorsed by CURA or by those providing funding for the study.

1988

Publication No. CURA 88-1

This report is not copyrighted. Permission is granted for reproduction of all or part of the material, except that reprinted with permission from other sources. Acknowledgement would, however, be appreciated and CURA would like to receive two copies of any material thus reproduced.

#### Edited by Mary Byers

<u>NOTE</u>: The Methodological Appendix, a separate document, provides the details of all procedures and the organization of the data file. A copy is available in the CURA library for examination.

### Contents

		Page
Figures and Tables	· ·	v
Acknowledgments		ix
Executive Summary		xi
Chapter 1. Why Study New Firms?		1
Understanding Economic Contributions Structural Change in Minnesota's Economy Sources of Employment, Structural Change Variation across Industries Variation across Regions Variation among New Firms Overview: Sources of New Firm Contributions		1 3 4 5 9 12 13
Chapter 2. New Firm Emergence		15
New Firm Births Birth Rate Variation across Minnesota Regions Regional Characteristics and New Firm Birth Rates Conclusion and Implications	· ,	15 17 20 22
Chapter 3. Predicting New Firm Contributions		25
New Firms as Equipotential Economic Entities Factors Associated with Contributions Multivariate Prediction of Contributions Development Types of New Firms Four Development Types Development Type and Sales History Development Type and Contributions Development Type and Minnesota Region Export Focus of New Firms Three Export Foci Export Focus and Contributions Export Focus and Minnesota Regions Growth and Export Focus of New Firms Growth, Export Focus, and Contributions Growth, Export Focus, and Minnesota Regions		25 26 30 31 32 33 34 36 36 38 41 41 43
Summary and Implications		46

	Page
Chapter 4. Start-up Problems: Nature and Current Status	49
Start-up Problem Dimensions	49
Variation among Minnesota Regions	51
Start-up Problems and Societal Contributions	52
Solution of Start-up Problems	53
Summary	53
Chapter 5. Minnesota Infrastructure	55
Local Infrastructure Features: Importance and Satisfaction	56
Minnesota Region and Infrastructure Satisfaction	60
Type of New Firm and Reactions to Infrastructure	61
Reaction to Potential Government Services	63
Summary	68
Chapter 6. Educational Institutions and New Firms	71
Educational Attainment of New Firm Principals	71
Employee Education	75
Educational Service Use and Contributions	77
Summary	78
Chapter Appendices	80
Chapter 7. Summary and Implications	91
Major Findings	91
Implications	94
Unresolved Issues	94
Appendices	
Appendix A. Definitions	97
Appendix B. Estimating New Firms	101
Appendix C. Research Procedure	109
Appendix D. Questionnaire	115
Appendix E. General Dimensions of New Firms Study:	
Constituent Items	131
Appendix F. New Firm Contributions: Associations with	
Selected Factors	137

#### Figures and Tables

Figures

Page

2

3

4

6

8

9

11

12

17

18

19

32

33

34

35

35

37

38

39

40

41

42

43

44

45

46

51

60

71

#### 1.1 Basic Economic Model 1.2 Employment Changes in Minnesota by Industry, 1975-85 1.3 Establishment Changes in Minnesota by Industry, 1975-85 1.4 Total New Firm Contributions by Industry 1.5 Percent of New Firm Jobs by Industry Sector, 1986 1.6 Minnesota Development Regions 1.7 Minnesota Region and Total New Firm Contributions 1.8 Percent of New Firm Jobs by Minnesota Region, 1986 2.1 Minnesota Regions and Total Annual New Firm Births (as estimated over 1979-84) 2.2 Minnesota Regions and New Firm Birth Rates Per 1,000 Establishments (as estimated over 1979-84) 2.3 Market Emphasis and Variation in Birth Rates 3.1 Development Type by Industry, 1979-84 3.2 Development Type and Sales History, 1979-84 3.3 Development Type and Total Contributions, 1985-86 3.4A Development Type by Minnesota Region, as of 1986 3.4B Development Type Within Minnesota Region, as of 1986 3.5A Export Orientation by Industry Sector 3.5B Minnesota New Firms: Export Status by Industry Sector 3.6 Export Focus and Percentage Contributions 3.7 Export Focus and Total Contributions 3.8 Export Focus and Minnesota Regions, 1986 3.9A Industry and Export Growth Patterns, 1986 3.9B Growth, Export Focus, and Industry, 1986 3.10 Growth, Export Focus, and Total Contributions 3.11A Growth and Export Focus by Minnesota Regions 3.11B All Minnesota New Firms by Growth and Export Focus 4.1 Start-Problem Severity for Minnesota Regions 5.1 Minnesota Region and Infrastructure Satisfaction 6.1 Principal's Education and Percent of Total Contributions

C.1 County Business Patterns (CBP) Establishment Census, Dun's Marketing Identifier (DMI) File Count, and New Firm Sample Compared, by Industry Sector 111

Page

## Tables

$1.1 \\ 1.2$	Minnesota Jobs and Personal Income, 1978-86 Average New Firm Contributions by Industry	1 8
1.3	Average New Firm Contributions by Region	10
1.4	Distribution of Jobs and Sales among New Firms	13
2.1	Estimates of Minnesota New Firm Births	16
2.2	Factors Associated with Export Potential New Firm Birth Rates	21
2.3	Factors Associated with Local Market New Firm Birth Rates	22
3.1	Selected Factors Related to New Firm Contributions	27
3.2	Multivariate Factors Predicting Contributions	29
3.3	Development Types of New Firms	31
3.4	Development Type and Average Contributions, 1985-86	33
3.5	Types of Export Foci	37
3.6	Export Focus and Average Contributions	38
3.7	Growth Patterns and Export Focus, in 1986	42
3.8	Growth, Export Focus, and Average Contributions	44
4.1	Start-up Problems: Ranked by Severity	50
4.2	Start-up Problems and Societal Contributions	52
4.3	Start-up Problem Solution and Contributions	53
5.1	Reasons for New Firm Location	56
5.2	Importance of Local Infrastructure Features	57
5.3	Satisfaction with Local Infrastructure Features	58
5.4	Satisfaction with and Importance of Infrastructure Features	59
5.5	Export, Growth Status, and Infrastructure Importance,	ć1
5 6	by New Firm Type Evenent Crowth Status, and Infrastructure Satisfaction	01
5.0	Export Growth Status, and Infrastructure Satisfaction,	()
57	Dy New Film Type Interest in and Value of Potential Concernment Services	60
5.8	Export Crowth Status and Interest in and Value of Petential	64
5.0	Covernment Services, by New Firm Type	66
5.9	New Firm Interest in Government Services	67
6 1	Principal's Education and Average New Firm Contributions	70
6 2	Principal's Educational Attainment and New Firm Industry	7/
6 3	Principal's Education and New Firm Development	75
64	Principal's Education and Market Orientation	75
65	Employee Skills Training or Education Difficult to Find	15
0.5	Provided by New Firms	76
6 6	Educational Institutions and Appropriate Training or Retraining	70
6.7	Use of Educational Programs and New Firm Contributions	77

A6.1	Difficult-to-find Employee Skills: Detailed Occupations	80
A6.2	Training or Education Provided by New Firms: Detailed	90
	Decupations	02
A0.3	University: Detailed Occupations	85
A6.4	Retraining or Additional Education Appropriate for Four-Year	
	Colleges: Detailed Occupations	86
A6.5	Retraining or Additional Education Appropriate for Community	
	Colleges: Detailed Occupations	87
A6.6	Retraining or Additional Education Appropriate for Vocational/	
	Technical Institutions: Detailed Occupations	88
A.1	Classification of Firm Economic Emphasis	99
B.1	Initial DMI Sample and Estimates of DMI Population: Minnesota	
	Excluding St. Paul	104
B.2	Initial DMI Sample and Estimates of DMI Population: St. Paul	
	Only	105
B.3	Developing Estimates of Total New Firm Populations	106
B.4	Developing Estimates of New Firm Populations by Minnesota	
	Region	107
C.1	Summary of Results of Data Collection Procedure	114
_		
E.1	Infrastructure Importance Dimensions	132
E.2	Start-up Problem Dimensions	133
E.3	Current Management Focus Dimensions	134
E.4	Competitive Strategy Dimensions	135
E.5	Existing and Proposed Government Services Dimensions	136
F.1	New Firm Characteristics and Average Contributions	138
F.2	Correlations Based on Transformed Contributions	139

Page



#### Acknowledgements

#### We extend our sincere thanks to:

THE RESPONDENTS: Over 1,120 men and women took the time from the management of their new firms to complete a rather elaborate questionnaire or to participate in a detailed phone interview; many responded to follow-up calls to clarify or expand original responses. Their cooperation was indispensable.

THE ADVISORY COMMITTEE: A number of individuals, experienced in research and its relationship to policy issues, contributed to the project through service on an advisory committee. Representing the financial sponsors, this group made contributions throughout the design and conduct of the project. (The final composition is listed here; members replaced during the project are shown in parentheses.)

Metropolitan Council of the Twin Cities Minnesota Chamber of Commerce and Industry City of Saint Paul

Saint Paul Port Authority State of Minnesota AVTI

State of Minnesota Community Colleges State of Minnesota Department of Trade and Economic Development State of Minnesota State Planning Agency University of Minnesota Center for Urban and Regional Affairs University of Minnesota Extension Service Charles Ballentine Betsy Kremser (John Olson) Warren Hanson (Alan Emory) Mara O'Neill Jim Terrell Melvin E. Johnson Bill Stock Douglas Easterling

Brian Zucker Alan Robinette Will Craig Tom Scott Wilbur Maki Gordon Rose John Sem

- THE RESEARCH STAFF: Special thanks to the many who helped prepare the mailings, complete the interviews, and code the results: Pat Johnson, Karen Kedrowski, Kelly Lynch, Christy Moore, John Paulette, Jeff Pilcher, Marjie Schmidt, Julie Segal, Brian Seedorf, and Jon Springer. Enhanced contributions were made by Cathy Davis, Laura Griffin, and Tamara Halbritter. Brian Harmon and Terry Schmit of the Department of Sociology Data Center provided creative solutions to an endless number of computer problems, always with good humor and infinite patience.
- SPECIAL THANKS ARE DUE TO OTHER STAFF: Shirley Bennett, administrator of CURA, who made a major contribution by keeping track of the diverse formal arrangements with three types of sponsors (University of Minnesota sponsors, Minnesota state agency sponsors, and other government agencies and private nonprofit sponsors). Her experience, care, imagination, and good nature in preventing financial chaos were a major contribution to the project. Thanks also to Mary Byers, who edited this manuscript with great care, and to Chris McKee, principal secretary at CURA, who prepared the final text and sorted through all sorts of last minute problems for us.



#### EXECUTIVE SUMMARY

The 12,000 to 28,500 new firms founded in Minnesota between 1979 and 1984 and active in 1986 were responsible for 6-14 percent of the 1986 jobs and 5-12 percent of the 1985 personal income. Further, the jobs provided by the new firms accounted for 42-99 percent of the net increase in Minnesota jobs between 1978 and 1986 and 12-29 percent of the net increase in personal income. Jobs provided by new firms represent a larger proportion of existing jobs in Greater Minnesota than in the Minneapolis-St. Paul region.

Such findings have led to greater interest by governments in strategies that may enhance the contributions of new firms. Three general emphases are possible: (1) improve the "business climate," (2) focus on new firms with a high potential for providing contributions, and (3) combine the two strategies.

This study considers the mechanisms behind the contributions of new firms in some detail. Most fundamental, these contributions reflect both the average (typical) contributions of new firms and the number founded, conveniently expressed as the annual birth rate. It is estimated that from 5,000 to 10,000 new firms are initiated in Minnesota annually. New firms in industries with substantial export potential (manufacturing, distributive services [wholesale], and business services) tend to be founded in regions with more mid-career adults, a greater proportion of college-educated adults, and higher per capita income, for example, the Twin Cities. The founding of firms in local market industries (construction, retail, consumer services) shows a lesser relationship to regional demographics; their emergence tends to reflect a turnover or replacement process.

There is substantial variation in the contributions of new firms: jobs provided range from none to over three hundred (average of nine); 1985 sales range from none to over \$25 million (average of \$538,000); and 1985 exports range from none (for 80 percent of the new firms) to over \$10 million (average of \$106,000). Predicting new firm contributions is possible, but it is not particularly affected by firm age (years since first sale). New firms in retail and consumer services were a major source of new jobs; new firms in manufacturing, distributive services (wholesale), and business services were equally important as sources of out-of-state exports.

The major factor that affects contributions is the growth rate established in the first twenty-four months after the first sale. One-third of the new firms have a high growth pattern, and these firms account for over twothirds of all jobs, four-fifths of all sales, and nine-tenths of all exports.

Those starting new firms are relatively satisfied with the local context, regardless of the state region. One notable exception is the universal dissatisfaction with taxes, which all consider very important. There is broad interest in government services, but different firms indicated strong interest in a different set of such services. Both the community college and the area vocational/technical institute (AVTI) system are seen as a useful source of educational and training assistance; graduates of these institutions have started new firms that are major sources of contributions.

The analysis of this data suggests that programs designed to locate and assist the small proportion of high-growth, export-oriented new firms may be the most efficient and effective. General attempts to improve the "business climate" may be both expensive and too diffuse to be effective.

#### Chapter 1

#### WHY STUDY NEW FIRMS?

Between 12,000 and 28,500 new firms, established in Minnesota between 1979 and 1984, survived into 1986.<sup>1</sup> In 1986 they contributed between 6 percent and 14 percent of all jobs and from 5 percent to 12 percent of all personal income. Their contributions were equal to 42-99 percent of the net growth in jobs between 1978 and 1986, and 12-29 percent of the net growth in personal income.<sup>2</sup>

	Total All Private Nonfarm Sectors ( <u>excludes farms and governments</u> )			Total Contributions of New F: )	
			Net	Minimum	Maximum
	1978	1986	Growth	(N = 12, 140)	(N = 28, 499).
Jobs (thousands)	1,646	1,928	282	119	280
Salaries, wages, proprietors' in- come (millions)	19,206	33,345	14,139	1,749	4,105

Table 1.1. MINNESOTA JOBS AND PERSONAL INCOME, 1978-86

These contributions are from new firms that did not exist as economic entities prior to 1979. None are branches or subsidiaries of established business firms. This report is intended to provide an improved understanding of the processes that lead to the economic and social contributions provided by new--not always small--firms.

These contributions are not uniform: there is substantial variation between industries, location in the state, and among new firms themselves. New firms may be separate, but are not equal in terms of their contributions (jobs, sales, and exports).

#### UNDERSTANDING ECONOMIC CONTRIBUTIONS

One strategy has dominated attempts to understand and explain economic outputs (goods/services, jobs). This is to consider the factors of production (land,

<sup>&</sup>lt;sup>1</sup> The basis for the range of estimates in the number of new firms is discussed in Appendix B.

<sup>&</sup>lt;sup>2</sup> Data on Minnesota taken from Tables SA25 and SA5 provided by the Regional Economic Information System, Bureau of Economic Analysis, U.S. Department of Commerce, April 1987.

labor, and capital) as they relate to outputs (goods, services, jobs, taxes). This model is presented in Figure 1.1. Regardless of the number of input variables or the sophistication of the quantitative model, the intermediate processes (indicated by "?") are deemphasized. The precise causal mechanisms by which the factors of production are combined to produce jobs and goods/ services are usually not specified in this strategy.

One purpose of this project is to supplement, not replace, this widely used model of economic growth by enhancing the understanding of the entities-business firms--that are the intervening economic processes.

Figure 1.1. BASIC ECONOMIC MODEL



In the United States, this "intervening mechanism"--the major source of jobs, goods, and services--consists primarily of private business firms. In Minnesota alone, about 100,000 individual establishments are providing goods/ services and jobs.<sup>3</sup>

Three general strategies for promoting economic development can be considered. One is to focus on reducing the costs of production or improving the quality of inputs. For example, a subsidy to lower interest rates for all borrowers (business and consumers) or improved education to enhance worker skills. A second strategy is to assist selected entities that make up the intervening mechanisms, such as a subsidized loan to a specific firm that may, in turn, increase employment and exports. A third approach would be to combine these two strategies.

The basic intent of this report is to explore the development and contributions of new firms. This, in turn, may help guide the formulation of policies to promote economic growth.

<sup>&</sup>lt;sup>3</sup> Establishments are defined as operating entities at a single location providing a good or service. New firms are almost universally single establishments. Existing firms may consist of multiple establishments. See Appendix A for a more complete discussion.

#### STRUCTURAL CHANGE IN MINNESOTA'S ECONOMY

The employment structure in Minnesota, as in many parts of the United States, is undergoing considerable change. Fluctuations in the number of jobs and establishments<sup>4</sup> are viewed with considerable alarm in those industries with major losses.

Less widely known is employment expansion in a variety of areasexpansions that have kept the job pool constant. The changes in Minnesota's employment for all industry sectors from 1975 to 1985 are presented in Figure 1.2.<sup>5</sup> The percentages were derived from the total number of full- and parttime jobs, which increased steadily from 1.9 million (1975) to 2.2 million (1980) and finally to 2.4 million (1985). The largest decreases were in distributive services, government, and farm employment (but not agricultural service); the largest increases were seen in business services, health/ educational/social services, and consumer services. Other industry sectors show little or no change.



Figure 1.2. EMPLOYMENT CHANGES IN MINNESOTA BY INDUSTRY, 1975-85

- <sup>4</sup> This refers to stand-alone economic entities, not always the entire legally \_ defined business organization.
- <sup>5</sup> Based on Table SA25 provided by the Regional Economic Information System, Bureau of Economic Analysis, U.S. Department of Commerce, April 1987. Includes full- and part-time employment.

A comparable analysis of the distribution of establishments is presented in Figure 1.3.<sup>6</sup> This reflects only nonfarm, private-sector business entities. The number of establishments has been steadily increasing; a total of 77,000 is recorded for 1975, 85,000 for 1980, and 104,000 for 1985. The pattern of industry-sector distribution is similar to that of the distribution of jobs. Relative increases were seen in the proportion of business service and health/ educational/social service establishments, and relative decreases were seen in retail, distributive services, and construction establishments. The other industry sectors--manufacturing, consumer services, agricultural services, and mining--reflect little change. Most of these shifts are observed nationwide.



Figure 1.3. ESTABLISHMENT CHANGES IN MINNESOTA BY INDUSTRY, 1975-85

#### SOURCES OF EMPLOYMENT, STRUCTURAL CHANGE

There are several mechanisms that may account for changes in jobs in any area (city, county, state, etc.). They include:

Birth:	Firms or establishments are creat	ed.	
Expansion:	Existing firms or establishments	increase i	n size.
Move in:	Existing firms or establishments	move into	a region.

<sup>&</sup>lt;sup>6</sup> This is based on the data from *County Business Patterns* and includes only business entities that have reported paid employees. The procedures for counting establishments were changed between 1982 and 1983, which may increase the proportion of temporary or seasonal establishments.

Death:	Firms	or	establishments	disappear (die).
Contractions:	Firms	or	establishments	shrink (become smaller).
Move out:	Firms	or	establishments	move out of a region.

There is little disagreement over the existence of these different processes; the major discussions concern the relative importance of each for the net change in jobs, establishments, and firms.

Recent research has suggested that a major source of new jobs is the creation of new firms and the expansion of existing small firms. However, the lack of reliable data on firms, particularly births and deaths, makes this a difficult area to study. Most economic information is gathered on inputs (interest rates, labor costs, etc.) and outputs (jobs, gross national product, etc.) rather than the characteristics of the intervening mechanisms. New firm births and small firm expansions are estimated to account for 40-80 percent of net new jobs.<sup>7</sup>

In addition to the creation of new jobs, new firms are of further interest as an indicator of the future directions of the economy. The future structural form makes its first appearance not as a planner's romantic vision but rather in the nature of new firms being established. Perhaps the best evidence that new economic opportunities are developing is the establishment of new entities (firms) to provide goods and services to underserved or unserved markets.

#### VARIATION ACROSS INDUSTRIES

The variation in social contributions--jobs, sales, and exports--of new firms in the major industries is presented in Figure 1.4. The industry sectors are defined in a slightly different way than in Figures 1.2 and 1.3. Transportation and utility establishments are combined with those in wholesale and labeled "distributive services." A separate category of "business (producer) services" is created from finance, insurance, real estate and relevant categories from services (as usually defined). Other categories within services (hotel and lodging, personal services, repair services, amusements) are classified as "consumer services."<sup>8</sup>

<sup>&</sup>lt;sup>7</sup> The higher estimates are usually associated with the analysis of David L. Birch, *The Job Generation Process* (Cambridge, Mass: MIT Program on Neighborhood and Regional Change, 1979); the lower estimates reflect the analysis of Catherine Armington and Marjorie Olde, "Sources of Job Growth: A New Look at the Small Business Role," *Economic Development Quarterly*, 6 (Fall 1982): 3-7.

<sup>&</sup>lt;sup>°</sup> These categories are described in more detail in Appendix A.



Figure 1.4. TOTAL NEW FIRM CONTRIBUTIONS BY INDUSTRY

300

0

-6-

Dist Serv

Serv

Bus

Soc Serv

Ed, ŕ

Manuf

Constr

Mining

Agri Serv

 $\bigotimes$ 

Cons Serv

Retail

Further, a range of estimates of the aggregate contributions is provided, reflecting the problems in estimating the total number of new firms.<sup>9</sup> The greatest range in estimates is seen in producer services, retail, and consumer services. These are industry sectors in which a private credit-rating service (Dun and Bradstreet) is most likely to miss a significant proportion of new firms, perhaps because they need less financial support (loans, credit, or capital) than those in manufacturing and distributive services.

Despite the range of estimates, the aggregate contributions of new firms vary substantially by industry. Retail and consumer service new firms are a major source of new jobs, although construction, manufacturing, distributive service, and business service new firms also make substantial contributions. New firms in five industry sectors are a major source of new sales: distributive services, construction, manufacturing, business services, and retail. Agricultural service and consumer service new firms also make significant contributions. The primary sources of out-of-state exports are manufacturing, distributive service, and business service new firms. Modest exports are provided by agriculture service, construction, and consumer service new firms.

The significance of new firms in relation to existing firms is determined by the proportion of all current jobs provided by new firms. This is presented, for both high and low job estimates, in Figure 1.5. The industry sectors fall into three groups: new firm jobs account for 10-30 percent of all jobs in construction, agricultural services, and consumer services; 5-15 percent of all jobs in retail, mining, distributive services, and business services; and 2-5 percent of all jobs in manufacturing and health, education, and social services.

This difference in aggregate or total contributions is due, in part, to the differences in the average contributions of different industry sectors. The average new firm contributions--jobs, sales, and exports--are given for each industry listed in Table 1.2. This represents the basic data gathered in the 1986 survey of Minnesota new firms.

<sup>&</sup>lt;sup>9</sup> In this analysis, two estimates of the number of new firms are used. The first is based on the estimated population of new establishments in the lists of Dun's Marketing Services, reduced by 50 percent, the percentage found not to be active in the phone verification of Dun's listings. The second is this figure with a correction factor for the new firms missed by Dun's field staff. These two figures provide a range of estimates. These ranges vary substantially across industry sectors, reflecting the differential effectiveness of Dun's procedures in capturing new firms. Coverage of new firms tends to be more timely if large amounts of credit are required to initiate the new firm. See Appendix B for details.



Figure 1.5. PERCENT OF NEW FIRM JOBS BY INDUSTRY SECTOR, 1986

Table 1.2. AVERAGE NEW FIRM CONTRIBUTIONS BY INDUSTRY

ut stag af inte al unit identities	No. in Weighted Sample <sup>a</sup>	1986 Jobs	1985 Sales (\$1,000)	1985 Out-of-state Exports (\$1,000)
All responding firms	1,119	9.0	514	106
Agriculture services	20	7.8	1,181	334
Mining	100	28.4	813	0
Construction	109	9.8	540	55
Manufacturing	143	10.3	61/	210
Distributive services	188	8.3	766	215
Business services	180	8.1	462	112
Retail	321	9.1	455	31
Consumer services	128	7.7	148	4
Health, education, social services	30	15.0	325	44
Total providing data		1,103	918	916

<sup>a</sup> Weighted to represent the total number of establishments in Dun's Marketing Identifier file as of October 1986.

<sup>b</sup> Analyses of industries do not include mining.

Two patterns are significant. First, there is variation among the averages for different industries: industry averages for 1986 new firm jobs range from eight to fifteen; industry averages for 1985 new firm sales from \$148,000 to \$1,181,000; and industry averages for 1985 new firm exports from \$4,000 and less to \$334,000. Second, this variation may be the greatest for exports, because average exports are substantial (over \$100,000/year) in only four industry areas: agriculture, manufacturing, distributive services, and business services. The average for agriculture, however, is enhanced by a single grain-handling firm with unusually high sales and exports-one risk of small samples.

#### VARIATION ACROSS REGIONS

Minnesota has been divided into thirteen development regions as shown in Figure 1.6. The average contributions of new firms in each region are presented in Table 1.3. The variation is substantial, as great as across industry sectors: 1986 employment varies from an average of four to seventeen; 1985 sales from an average of \$185,000 to \$819,000; 1985 out-of-state exports from \$2,000 to \$212,000. This reflects the existing economic structure and, in turn, the types of new firms that emerge in different regions.





-9-

		N7		1005	1985
		NO. IN	1000	1985	Out-oi-state
		Weighted	1986	Sales	Exports
		Sample <sup>a</sup>	Jobs	(\$1,000)	(\$1,000)
A11	Responding Firms	1,119	9.0	514	106
1	Northwest	10	8.7	166	52
2	Headwaters	9	4.2	168	40
3	Arrowhead	58	7.3	326	47
4	West Central	40	8.0	358	116
5	Region Five	29	12.2	940	385
6W	Six West	8	6.4	123 <sup>b</sup>	14
6E	Six East	16	11.8	675	115
7W	Central Minnesota	42	10.2	1,097	109
7E	East Central	13	5.2	376	4
8	Southwest	21	5.9	261	11
9	Region Nine	35	6.4	343	64
10	Southeastern	57	7.2	312	74
11	Metropolitan Council	781	9.9	571	112

#### Table 1.3. AVERAGE NEW FIRM CONTRIBUTIONS BY REGION

<sup>a</sup> Weighted to represent the total number of establishments in Dun's Marketing Identifier file as of October 1986.

<sup>b</sup> Excludes a single grain-handling firm with reported sales in excess of \$10 million.

The estimated total contributions of jobs, sales, and exports for each region are presented in Figure 1.7. As with the analysis by industry sector, the variation in estimates reflects ambiguity regarding the actual number of new firm births.

There is not, however, much uncertainty about the region receiving the greatest absolute benefits from new firms. New firms in the Twin Cities metropolitan region provide jobs, sales, and out-of-state exports roughly twice those of the rest of the state combined. This reflects, more than anything else, the most significant factor affecting new firm foundings (or births): they are created in regions with high levels of economic activity and a high density of business entities.

These differences point to variation in both industry sector and growth patterns typical of new firms in each region.  $^{10}$ 

<sup>10</sup> The range of estimates in Table 1.3 reflects the same adjustments discussed in footnote 9 related to Figure 1.4 and reviewed in Appendix B.





The relative significance of new firms for the regional economies is indicated by the proportion of all existing jobs provided by new firms (see Figure 1.8). Most dramatic is the clear difference between the Twin Cities area and the remaining twelve Greater Minnesota regions: the proportion of jobs accounted for by new firms is *higher* in every nonmetropolitan region. They are particularly significant for regions in the central portion of the state. And 20-40 percent of existing jobs in Region Five are provided by new firms. This may reflect either economic growth or the dominance of certain industries with high volatility--construction, retail, consumer services--in these Greater Minnesota regions.



Figure 1.8. PERCENT OF NEW FIRM JOBS BY MINNESOTA REGION, 1986

#### VARIATION AMONG NEW FIRMS

Variations in contributions among new firms are even more striking than the differences among industry sectors or Minnesota regions. As shown in Table 1.4, 1986 jobs vary from zero to over 300; 1985 sales range from zero (for new firms not yet selling a product or service) to over \$20 million. Exports are not shown but although three-fourths of the new firms have no exports at all, the largest reported 1985 out-of-state exports of \$15 million.

1986 Jobs		1985 Sales (\$1,000)	
(n = 1, 102)	New Firms (%)	(n = 921)	New Firms (%)
0-4	46	<100	33
5-9	. 26	101-250	26
10-14	11	251-500	16
15-24	8	501-1,000	13
25-49	5	1,001-2,500	7
50-74	2	2,501-5,000	3
75-99	1	5,001-8,600	1
100-303	1	8,601-30,000	1

Table 1.4. DISTRIBUTION OF JOBS AND SALES AMONG NEW FIRMS

A major objective of the following analysis is to further understanding of the sources of this substantial variation in new firm contributions (jobs, sales, and exports).

#### OVERVIEW: SOURCES OF NEW FIRM CONTRIBUTIONS

Two separate factors affect the total contributions made by new firms:

- The number initiated--or born.
- The job, sales, and exports of the typical new firm.

The remainder of this report will address these two topics. Chapter 2 explores the factors associated with differences in the initiation (birth rates) of new firms across the thirteen regions in Minnesota. Chapter 3 examines the prediction and understanding of the variation in new firm contributions.

Chapter 4 reviews the start-up problems reported by the new firms, the extent to which there is variation across the regions of Minnesota. Chapter 5 examines the reactions of new firms to both their immediate local context and services that might be provided by state, regional, or local government agencies. Chapter 6 attends to the educational attainment of those starting new firms and the educational needs associated with the jobs created by new firms.

Chapter 7 summarizes the major points of this report and explores selected policy implications.

Data for analysis are taken from federal government statistics or from a special survey conducted for this project. A sample of new firms was drawn from the Dun's Marketing Identifier (DMI) file--based on credit ratings compiled by Dun and Bradstreet's financial services. A phone verification of these listings determined that only 50 percent qualified as autonomous, ongoing, new firms. A sixteen-page questionnaire was sent to qualified firms; those that did not return the questionnaire after three mailings were contacted for a phone interview. Three-fourths of eligible firms provided the critical information requested by the survey. A more complete summary of the research procedure is given in Appendix C of this report; the questionnaire is included in Appendix D. Full details of the research procedure are available in the Methodological Appendix, a separate document.

High confidence that the sample is representative of all new firms in Minnesota is justified by the close match of the sample with the best available censuses of business establishments; the careful screening of firms that were not new, not autonomous start-ups, and not active; and the high response rate.

#### Chapter 2

#### NEW FIRM EMERGENCE

Before a new firm can make a contribution--provide jobs, sales, or exports--it must be created. The full impact of new firms is substantially affected by their total number. This chapter will first consider estimates of the total number of new firms established each year in Minnesota. While there is substantial variation in the incidence of firm creation across regions, there is also regional variation in the size of human and organizational populations. Therefore, the following analysis will present an across-region comparison based on the annual rate of new firm foundings as a proportion of existing business entities.

Fluctuations in the annual birth rate are substantially less than the variation in absolute births. Nonetheless, there are still considerable differences among the thirteen regions of Minnesota. Some regions are more fertile for new firm births than others. This leads to a preliminary exploration of selected factors that may be related to variations in region fertility. These are, in a sense, the characteristics of a socioeconomic system that can affect outputs.

There is a major interest in the contributions of new firms to exports. The analysis in Chapter 1, particularly Figure 1.4, indicated that new firm contributions to out-of-state exports came almost entirely from three industries. Manufacturing, distributive services, and business services will be treated and referred to as "export potential" industries. Not all firms in these industries export; firms that <u>do</u> export, however, are quite likely to be in one of these industries. Firms in export-oriented industries are also more likely to involve more complex, sophisticated products, services, or internal technology. These may require more educated personnel and greater capital investment.

In the other industry sectors, total export contributions are quite small. Dominated by new firms in construction, retail, and consumer services, they will be referred to as "local market."

#### NEW FIRM BIRTHS

There is no public census of new firm births, or deaths, for Minnesota.<sup>1</sup> As a result, it is necessary to use available data to estimate new firm births. The estimates used for this analysis are based on the data provided by Dun's Marketing Identifier (DMI) files. They are summarized in Table 2.1; details are given in Appendix B.

<sup>&</sup>lt;sup>1</sup> This need not be the case, for the state has this information as part of the data collected with unemployment insurance payments. Other states, e.g., Mississippi, have used this information to provide annual descriptions of establishment births and deaths.

	New Firm Births (1979-84)	Average Annual Births	Annual Birth Rate (per 1,000) <sup>a</sup>	
Original Dun's Marketing Identifier count	24,293	4,048	42	
Corrected for new firms missed by Dun & Bradstreet field staff <sup>b</sup>	56,999	9,500	98	•
Corrected for listings that are not new, autonomous, and still in business <sup>C</sup>	28,499	4,750	49	

Table 2.1. ESTIMATES OF MINNESOTA NEW FIRM BIRTHS

<sup>a</sup> Assumes 97,292 establishments.

- <sup>b</sup> David Birch and Susan McCracken, "The Small Business Share of Job Creation: Lessons Learned from the Use of a Longitudinal File" (Cambridge, Mass.: MIT Program on Neighborhood and Regional Change, mimeo., March 1983).
- <sup>c</sup> Based on results of phone verification interviews completed in Minnesota; 50 percent estimated as ongoing, new firms that started autonomously.

Other analyses of new firm births provide confidence that these are reasonable estimates. A study completed on new firms filing unemployment insurance in Minnesota found that the 1977 birth rate of establishments was 91 per thousand 1976 establishments.<sup>2</sup> Data on Mississippi for 1984 indicate a birth rate of 116 per thousand establishments existing in 1983.<sup>3</sup> Analysis of entries into the unemployment insurance file in Tennessee yielded an average of 111 entries per thousand existing establishments over a six-year period (1980-85).<sup>4</sup>

Applying the annual birth rate reported for Mississippi (116/1,000) to Minnesota would provide an upper estimate of 11,286 new firms each year. A

<sup>&</sup>lt;sup>2</sup> John Tauzell, "Survival of Minnesota New Businesses: 1977-1980." Review of Labor and Economic Conditions, Minnesota Department of Economic Security, 9 (August 1982): 10-17. For 1977 there were 7,105 new filings; County Business Patterns lists 77,937 Minnesota establishments for 1976.

<sup>&</sup>lt;sup>3</sup> Mississippi Employment Security Commission, Labor Market Information Department, "Mississippi's Business Population: Births, Deaths & Changes in Ownership--1984," August 1985. Excluding public administration establishments, they report 5,587 new listings appeared in 1984; 48,340 existed in 1983.

 <sup>&</sup>lt;sup>4</sup> William F. Fox et al., Entries and Exits of Firms in the Tennessee Economy: 1980-1985, publication number E01-1490-010-87 (Knoxville: University of Tennessee, Center for Business and Economic Research, May 1987).

minimum estimate would be the uncorrected counts based on Dun's Market Identifier file, about 4,000 or 40 per thousand per year.

As a conservative estimate of new firm births, the count derived from the DMI file (a total of 24,293 for six years or an average of 4,048 each year) is used in the following analysis.<sup>5</sup> It is the basis for the estimate of the annual number of new firms to emerge in each of the thirteen regions of Minnesota, as presented in Figure 2.1.

Of the new firm births in Minnesota, 53 percent occurred in the Minneapolis-St. Paul (Twin Cities) metropolitan region. Perhaps more striking is that two-thirds of the new firms in the three export potential industries emerged in this region.



Figure 2.1. MINNESOTA REGIONS AND TOTAL ANNUAL NEW FIRM BIRTHS (AS ESTIMATED OVER 1979-84)

#### BIRTH RATE VARIATION ACROSS MINNESOTA REGIONS

The total number of new firms initiated in the thirteen regions of Minnesota varies by a factor of 50 (from 40 to 2,000). This and the dramatic variation in economic activity among these regions suggest a standardized measure--a birth rate. The birth rate for new firms is estimated for each region by dividing the estimate of the annual number of new firms by the number of existing establishments (per 1,000).<sup>6</sup> New firm birth rates are presented in

It is very close to the best estimate of 53,784 after the two corrections are completed (see Appendix B for details).

Based on the count of establishments in County Business Patterns for 1983.

Figure 2.2 for two types of industry sectors--local market and those with export potential. The range of annual birth *rates* is considerably narrower (a factor of 2.0) than the range of absolute births (a factor of 50).

Export industries

Local market





Combined with the knowledge that virtually no new firms are "move-ins," this would suggest that the established industries are a major source of new firms in any region. New firms do not migrate, they are created locally.<sup>7</sup> The presence of a particular industry may be important as a source of new firms in that or complementary industries.<sup>8</sup>

Despite the relative reduction in variation, there is still substantial interregion variation in birth rates. The average (per 1,000) and range of annual birth rates are as follows:

	Average	Range
All industries	81	54-116
Export potential	36	21-53
Local market	45	31-67

<sup>&</sup>lt;sup>7</sup> See the discussion at the beginning of Chapter 4.

<sup>&</sup>lt;sup>8</sup> The background of those that start new firms increases confidence in this interpretation: most new firm start-up teams are dominated by those with substantial business experience, and most of this experience is in the same industry.

This raises an important question regarding the relationship of birth rates in these different types of industries: "Are export potential birth rates systematically related to local market birth rates?" If highly related, the same policies can be adopted to promote new firm births in all industries. If unrelated, it suggests different policies may be required to stimulate different types of new firms.

The correlations between three estimates of new firm births--all firms, those with export potential, and local market--are provided for the thirteen Minnesota regions in Figure 2.3. Despite the small number of units for analysis (thirteen regions), two important patterns are seen:

- There is a modest statistically significant relationship between the birth rates of export potential and local market new firms.
- The all-industry birth rate appears to reflect the local market new firm birth rate more than the export potential new firm birth rate.

The first of these suggests that the processes that lead to the initiation of new firms in export-potential and local-market industries may be different. Different variables may be involved in predictions of variations of the two different birth rates. Different policies and procedures will perhaps be necessary to affect birth rates in different industry sectors.



Figure 2.3. MARKET EMPHASIS AND VARIATION IN BIRTH RATES

- 1. Numbers indicate Pearson Product Moment Correlations among all thirteen Minnesota regions.
- 2. Export potential new firms include those with primary classification (SIC codes) as manufacturing, distributive services (mostly wholesale), and producer (business) services. All others are placed in the local market category.
- 3. New firm (establishment) estimates are based on data provided by Dun's Marketing Services for firms established 1979-84 as of October 1985. A base count of total firms (establishments) has been developed from information reported in *County Business Patterns* (U.S. Department of Commerce) for Minnesota for 1983.
- 4. Two-tailed test of statistical significance: \* p < 0.05; \*\*\* 0.001 p.

#### REGIONAL CHARACTERISTICS AND NEW FIRM BIRTH RATES

Understanding the characteristics associated with interregion variations in new firm births involves, first, specifying the variables to be considered. It is clear that the number could be quite large. Based on the analysis of the start-up team characteristics in a 1984 pilot study of new firms,<sup>9</sup> and the recurring comment that the unemployed tend to start many new firms, four factors were selected to represent quite different characteristics of each region for a preliminary analysis:<sup>10</sup>

General Feature	Operational Indicator .	
Availability of jobs	Unemployment rate, 1980	
Individual income	Per capita income, 1979	
Pool of potential start-up team members	Percentage of adults 25-39 years old, 1980	
Educational attainment	Percentage with high school experience, 1980 Percentage with college experience, 1980	

Indicators for all characteristics were computed for each region by aggregating data on counties provided for the 1980 census. For per capita income this was based on 1979 income data; for all other variables it was current as of 1980. This was early in the six-year period for which new firm births were estimated, overlapping with 1979 and 1980, preceding the other four years, 1981-84. This provides some confidence that the regional characteristics preceded the new firm births, and that these regional characteristics were related to processes causing new firm births.<sup>11</sup>

The simplest measures of association between these region characteristics and variation in export potential new firm birth rates are presented in Table 2.2.

The highest levels of association with the birth rate of export potential new firms involve social, life-course characteristics rather than economic variables. The larger the percentage of midcareer adults and those with education beyond high school, the higher the birth rates of export potential

<sup>&</sup>lt;sup>9</sup> Paul D. Reynolds and Steve West, New Firms in Minnesota: Their Contributions to Employment and Exports, Start-up Problems, and Current Status (Minneapolis: University of Minnesota, Center for Urban and Regional Affairs, 1985).
<sup>10</sup> A complete portrayal of the socioeconomic character of a multicounty region

<sup>&</sup>lt;sup>10</sup> A complete portrayal of the socioeconomic character of a multicounty region would involve a much wider range of variables. This was beyond the capacity of the available resources.

<sup>11</sup> These associations may, however, be spurious. More research involving a wider range of regions is required to investigate this factor as thoroughly as it deserves.

new firms. The unemployment rate actually has a negative correlation with the birth rate of export potential new firms.

Table 2.2. FACTORS ASSOCIATED WITH EXPORT POTENTIAL NEW FIRM BIRTH RATES

	Pearson Product
	Correlation
Percentage of population	+0 64**
Percentage of adults over 25 with 16 or more years of education	th +0.44
Percentage of adults over 25 with 12 or more years of education	th +0.42
Per capita income (1979)	+0.37
Unemployment rate	-0.24
	Percentage of population between 25 and 39 years old Percentage of adults over 25 with 16 or more years of education Percentage of adults over 25 with 12 or more years of education Per capita income (1979) Unemployment rate

Notes:

1. All data on development regions were taken from 1980 U.S. Census.

2. Two-tailed tests of statistical significance: \*\* 0.01 .

The negative correlation between the unemployment rate and the annual birth rates of export potential new firms suggests that starting export potential new firms does not seem to be a reaction to problems in locating a job.<sup>12</sup> Export potential new firms may emerge in regions with expanding economies, where the unemployment rate tends to be low.

The association between these same variables and the birth rates of local market new firms is presented in Table 2.3.

Here, the patterns are quite different. Of primary importance is that none of the associations is statistically significant. This suggests that the founding of new firms in local market industries is a general, widespread phenomenon that is not affected by the regional features included in this analysis. Two regional characteristics have a relatively higher association with new firm births, namely, the presence of young adults and a higher unemployment rate. This suggests that where young adults cannot find jobs, they may start new firms with a local market orientation.

<sup>&</sup>lt;sup>12</sup> This is consistent with the career transitions reported by those starting new firms: less than 10 percent say they were unemployed prior to the new firm start-up.

		Pearson Product Correlation
Presence of young adults	Percentage of population between 25 and 39 years old	+0.27
Availability of jobs	Unemployment rate	+0.25
Advanced education	Percentage of adults over 25 16 or more years of education	with n -0.06
Public (high school) education	Percentage of adults over 25 12 or more years of education	with 1 -0.09
Personal income	Per capita income (1979)	-0.16

Table 2.3. FACTORS ASSOCIATED WITH LOCAL MARKET NEW FIRM BIRTH RATES

Note:

1. All data on development regions were taken from 1980 U.S. Census.

2. No correlations are statistically significant at 0.05 level.

In short, starting a construction firm, retail shop, or customer service facility may be a response to poor employment opportunities. Those starting new businesses may live in regions with contracting economies, as reflected in low incomes and higher unemployment rates--regions with an absence of new firms in export potential industries.<sup>13</sup>

Although tentative, this analysis is consistent with the initial finding, the low relationship between birth rates of local market and export potential new firms. Different regional characteristics appear related to the birth rates of the different types of new firms.

#### CONCLUSION AND IMPLICATIONS

The findings from this analysis, factors associated with variations in new firm births in the thirteen Minnesota regions, include:  $^{14}$ 

<sup>&</sup>lt;sup>13</sup> An analysis of new firm births in ten Pennsylvania regions (two major urban areas) found stronger support for this interpretation of the factors affecting local market new firm births. Paul Reynolds and Steve Freeman, 1986 Pennsylvania New Firm Survey Volume Two (Washington, D.C.: Appalachian Regional Commission, 1987).

Appalachian Regional Commission, 1997.
14 A major caution is associated with this analysis: the number of units of analysis is small (only thirteen regions were involved), and extreme values with such a small sample may lead to unreliable correlations--relations that are difficult to replicate.

- Evidence that the same factors are not equally significant in explaining the births of local market and export potential new firms.
- Evidence that education and life course or career stage are of major significance in births of export potential new firms; unemployment rates are not.
- Suggestive evidence that higher unemployment may lead to the founding of new construction, retail, and consumer service firms.

These findings, albeit preliminary, have implications for public policy. First, new firm births in different industry sectors may be unrelated (or have only a very modest relationship). There may be a time lag--not explored in this analysis--between the expansion of new firms in one industry and expansion in other industries in the same area.

Second, the most critical factors in the birth rates of new firms may be difficult to affect with public policy-at least in a short period of time. Most important is the presence of established firms in a given industry as the source of trained individuals in midcareer, those responsible for initiating new firms in export potential industries. It is these college-educated, experienced individuals between 30 and 49 who appear to be absent from rural regions. This may reflect the rural-urban migrations that have been occurring for decades.

On the other hand, an increased awareness of the population characteristics of a region can be helpful in guiding public programs to encourage new firms. The programs can be tailored to the types of firms (or industries) prevalent in the region and of interest to the indigenous population. After all, virtually all new firms in a given region are created by established residents.
# Chapter 3

### PREDICTING NEW FIRM CONTRIBUTIONS

The contributions (jobs, sales, exports) provided by new firms reflect both the number of firms initiated and their individual contributions. There is substantial variation in the contributions of individual new firms. The average number of new jobs provided in 1986 was 9.1, but the range was from zero to over 300. Average 1985 sales were \$538,000, but the range was from zero to \$29 million. Average 1985 exports were \$106,000, but the range was from zero to \$12 million; only one in three had any exports at all. Clearly, new firms are not equal in their contributions; this chapter will explore the sources of these variations.

There are two ways of thinking about new firms and how their contributions might vary:

- As equal in potential for contributions, with variations related to stage of development, technology, or differences in the special characteristics or situation of individual firms; or
- As different types, each with a distinctive or characteristic potential for contributions.

If the first model is useful, then it should be possible to identify those new firm characteristics associated with the level of development and, in turn, the level of contributions. Policies to promote economic growth could then focus on factors critical to the development of all firms.

The second model leads to the search for typologies of new firms and the relative contributions of each type. If accurate, it would provide information that would allow the identification of the types of new firms with the greatest potential for contributions. Programs could then target limited resources to those firms most likely to foster economic growth.

This chapter will begin with multifactor predictions of new firm contributions, based on data collected using the 1986 Minnesota New Firm Survey. This is presented as an "equipotential model" of new firms as sources of economic development. Analysis of this model is followed by a discussion of typologies of new firms based on, first, their developmental patterns; second, their export focus; and, third, a combination of the two. Implications for promoting regional economic growth are considered in the summary.

#### NEW FIRMS AS EOUIPOTENTIAL ECONOMIC ENTITIES

Attempting to predict the contributions of new firms, as economic entities with equivalent potential, is relatively straightforward. It involves the use of standard analysis procedures (correlations, multiple regression analysis) to determine which characteristics of individual new firms are related to current contributions (jobs, sales, and exports).

-25-

# Factors Associated with Contributions

The data collected in the 1986 Minnesota New Firm Survey have been consolidated into 142 variables that might be related to new firm contributions.<sup>1</sup> Measures of association between the complete list and the statistical significance of each factor are presented in Appendix F. The highest correlation is 0.60; the majority are statistically significant; and the patterns vary for different types of contributions (jobs, sales, and exports).

As a preliminary step, variables that had correlations of about  $0.20^2$  or higher for any one of the three types of contributions (jobs, sales, exports) were chosen for special attention. Then the attempt was made to include the most significant general factors developed from multiple-item inventories regarding conditions that might precede contributions: start-up problems, current management focus, and strategic (competitive) emphasis.<sup>3</sup> These variables are presented in Table 3.1.

The substantial variation from one column to the next is important. For example, variables highly correlated with jobs (first-year employment; low initial proportion of managers and professionals) are not those most highly correlated with sales (first-year sales, annual sales growth). This suggests that different factors and processes are related to the different contributions.

#### Multivariate Prediction of Contributions

Virtually all the factors in Table 3.1 have statistically significant relationships with the contributions (jobs, sales, and exports). But this does not provide information on the importance or strength of the relationship or the predictions available from combinations of variables. These estimates require more sophisticated analyses, such as are provided by multiple regression procedures.

<sup>&</sup>lt;sup>1</sup> This includes responses to five multiple-item inventories: 20 local infrastructure characteristics; 43 start-up problems; 25 aspects of current management focus; 14 strategic or competitive emphasis; and reactions to 31 services that might be provided by local, regional, or state government agencies. They were consolidated into general dimensions using factor analysis techniques. Details are provided in Appendix E.

<sup>&</sup>lt;sup>2</sup> Note that with such a large sample, virtually every correlation was statistically significant.

<sup>&</sup>lt;sup>3</sup> The latter two, management emphasis and strategic focus, may not be causes of current contributions. Measured at the same time as contributions, they may be associated with variations in jobs, sales, and exports.

	Pearson	Correlati	on with
	1986	1985	1985
	Jobs	Sales	Exports
· · · · · · · · · · · · · · · · · · ·			
Interest costs, % of total	10**	12**	19**
Taxes costs, % of total	15***	21***	12*
Sales within MN. 1985	.34***	.50***	.23***
Sales, adjoining MN, 1985	.23***	.33***	.47***
Sales, rest of U.S., 1985	.20***	.30***	. 50***
Sales, abroad, 1985	.10**	.11***	.19***
Informal finan. resources: personal savings	.26***	.23***	.23***
Total informal financial resources	.32***	.30***	.30***
Start problems, av. severity: employees	.21***	.15**	.27**
Start problems, av. severity: financing	.19***	.17***	.29***
Start problems, av. severity: government	.23***	.20***	.34***
Infrastructure importance: labor	.32***	.21***	.02
Management focus: planning	21***	18***	21***
Management focus: marketing	15***	17***	24***
Management focus: prin. involvement	09*	12**	20**
Management focus: product/service	.02	.03	.20**
Management focus: experience	06	15***	19**
First-year employment	.33***	.30***	.28***
Average sales growth, all years	.30***	.47***	.45***
Average sales growth, third year	.41***	.54***	.48***
Size of the start team	.27***	.21***	.15***
Average number hours worked/week	.09**	.21***	.04
Age range of start team	.16***	.21***	.16***
Percent males on the start team	.08**	.21***	.23***
Percent start team working 50+ hours/week	.09**	.24***	.08
Av. interest, train/ed. programs	.31***	.18***	.12*
Av. interest, tech/growth programs	.24***	.19***	.31***
Av. interest, capital/financing programs	.22***	.18***	.19***
1st formal finan: work cap line of credit	.25***	.26***	.26***
lst formal finan: inventory term loans	.12***	.20***	.14***
1986 formal finan: work cap/line of credit	.17***	.18***	.17***
Count of start problems: products/markets	.19***	.17***	.13***
Count of start problems: employees	.37***	.32***	.37***
Count of start problems: planning	.22***	.10*	.11*
Count of start problems: financing	.22***	.24***	.27***
Count of start problems: government	.26***	.23***	.28***
Count of start problems: total	.35***	.33***	. 30***
Start problems, partly solved: employees	.27***	.12**	. 20***
Start problems, fully solved: financing	.12**	.21***	.05

# Table 3.1. SELECTED FACTORS RELATED TO NEW FIRM CONTRIBUTIONS

Notes:

1. Low value indicates increased emphasis.

2. Statistical significance: \* 0.01 \*\*\* p < 0.000</pre>

3. All correlations are based on normally distributed log 10 transformations

of jobs, sales, or exports.

٢.

For the multiple regression analysis, the variables listed in Table 3.1 were supplemented with three that indicated the presence or absence of a characteristic: starting out as a corporation, gestation period starting with sales, and gestation period starting with hiring.<sup>4</sup>

As the emphasis is on predicting future contributions of new firms, variables were chosen that would reflect their situation or status in the first one to two years of start-up. That is, a variable like "fourth-year sales" was <u>not</u> used to predict "1985 sales," for "1985 sales" would be the "fourth-, fifth-, or sixth-year sales" for a proportion of the sample.

A standard multiple regression using a stepwise procedure was utilized for the analysis.<sup>5</sup> The procedure adds, tests, and subtracts variables in an attempt to find the smallest set of variables that wll have the greatest predictive power. The procedure is designed to retain only those variables that make a statistically significant contribution to the prediction (of jobs or sales or exports, as the case may be). The omission of variables is often as significant as their inclusion.

The major factors included in the regression analysis for the three predictive equations are presented in Table 3.2. "Std Beta" refers to the weights adopted by the procedure for the best possible predictions of, in this case, the jobs, sales, and exports of the new firms. A larger "beta weight" (ignoring the sign) indicates that the factor has a larger independent influence on the contribution (jobs, sales, and exports). Only variables that would improve the predictions are listed; exclusion implies it did not provide a statistically significant increase in predictive power.<sup>6</sup>

One indication that critical predictive variables have been included in the analysis is a high multiple correlation and estimate of explained variance. In this case the results are quite encouraging. The percentage of explained variance for the three measures of social contributions are as follows:<sup>7</sup>

<sup>&</sup>lt;sup>4</sup> The period of time in which major start-up events occurred was called the gestation period. It could be initiated by any of four events: personal commitment by the start-up team members, formal financial support, sales, or hiring. The gestation period averaged six months, but ranged from one to over sixty months.

<sup>&</sup>lt;sup>5</sup> "Regression," Chapter 35 in SPSS, Inc., <u>User's Guide: SPSS-X</u>, 2nd ed. (New York: McGraw-Hill, 1986).

<sup>&</sup>lt;sup>6</sup> The log transformations of the contributions (jobs, sales, and exports) were normally distributed and used as dependent variables for the analysis. There were no significant departures from normality among the residuals. The required assumptions for high confidence in the results appear to have been met.

<sup>&</sup>lt;sup>7</sup> Note, however, that only (?) 342 firms with any exports are included in the last regression, excluding (?) 75 percent of the firms in the sample.

	Total Firms <u>in Analysis</u>	Multiple <u>Correlation</u>	Variance Explained (%)
Jobs	1,101	0.74	55
Sales	917	0.65	42
Exports	301	0.65	42

This high explained variance suggests it is possible to have confidence that the most important of the 142 variables were among those selected for the multiple regression. From these 42 most important variables, only 10 were included in the prediction of sales; 10 in the prediction of jobs; and 4 in the prediction of exports. It also suggests that it is unlikely that critical variables have been omitted.

Table	3.2.	MULTIVARIATE	FACTORS	PREDICTING	CONTRIBUTIONS
-------	------	--------------	---------	------------	---------------

	Std Beta	Variable Change to Increase Contribution
1004 - 1		
1986 Jobs		
PROFFP	-0.228	Proportion of managers, professionals first year
ISUPT	0.277	Total start-up problems solved
VB9	0.184	Total informal, prestart financing
XOF15	0.632	Total employment in first year
NOSKLGP	-0.471	Proportion of current employees unskilled
MF2	-0.156 <sup>a</sup>	Current management focus on marketing
IMP4	0.171	Infrastructure importance of labor
USAMT	0.165	1985 sales outside of Minnesota
ASALGR1	-0.189	Average sales growth years 1-2
VIN1	0.160	Initial formal working capital loan
1985 Sales		
ASALGR1	0.188	Average sales growth years 1-2
ISUP2	0.194	Count of major start-up problems with employees
BV9	0.190	Total informal, prestart financing
HARDWK	0.122	Percentage start-up team working 40-60 hours/week
PFS4	0.189	Percentage of financial start-up problems solved
BOYPC	0.141	Percentage of males on start-up team
PCMAT	0.146	Percentage of material costs
IMP4	0.141	Infrastructure importance of labor
ISUR4	0.133	Count of major start-up problems with financing
YR1R	0.129	First-year sales
1985 Exports		
YR1R	0.332	First-year sales
SAMPSIC2	0.303	Manufacturing, distributive, or business service industry
ISUP2	0.257	Count of major start-up problems with employees
SUP9	0.248	Average severity local government start-up problems

<sup>a</sup> Lower value indicates increased emphasis.

Providing interpretations as to why each variable might be important in increasing jobs, sales, or exports is an interesting and complex challenge. For current jobs provided, measures of the initial size of the firm (firstyear employment, total informal financing, initial working capital loan) as well as the growth pattern (average sales growth in first year) are useful predictors. Others involve the number of start-up problems (and the extent to which they are solved) and some indication of the new firm's structure (fewer managers, staff professionals the first year), as well as the firm's management emphasis (on marketing), concerns (with the local labor supply), and the scope of the market for its goods and services (nationwide).

For current sales, both first-year sales and average sales growth in the first year are of significance. Higher sales are associated with more concerns with personnel (major start-up problems with employees, increased importance of local labor supply) and financing (start-up problems with financing, amount of initial informal financing). The start-up teams are dominated by males and have a high level of work involvement (forty to sixty hours per week).<sup>8</sup>

Analysis of the factors associated with exports is more restricted, for only 27 percent of the new firms report out-of-state exports in 1985. Nonetheless, only four variables have a significant impact on the predictions: the average first-year sales and one major class of start-up problems (initial problems with employees and labor). A critical factor is the industry of the new firm. Being in an export-oriented industry--manufacturing, distributive service, or business service--is significant, as are reports of start-up problems involving local government (which includes concerns regarding liability insurance and coping with government regulations).

These predictive variables are marked by an absence of factors that could be affected--either easily or at all--through public policy or programs. The need for suitable personnel (which might be affected by appropriate public education and training programs) and problems with financial support both appear as predictive variables, but their predictive influence is modest compared to the other variables. This encourages other ways of thinking about new firms and their contributions.

#### DEVELOPMENT TYPES OF NEW FIRMS

If all new firms were similar, varying only in development or maturity, any number of factors might be correlated with "growth" and contributions. Age is among the most obvious; it seems reasonable to expect older firms to be larger and make greater contributions. Further, age should be most significant in the early development of new firms, when their contributions might easily double or triple in a few years. Despite this restriction on the range of ages--the oldest firms had their first sales in 1979, six years before 1985-the range is great enough for analysis.

<sup>8</sup> But not an extremely high work involvement, more than sixty hours per week.

Age does not have a dramatic relation to the growth and contributions of new firms however. The correlation of firm age with jobs is 0.06; sales, 0.12; and exports, 0.02. Although age-jobs and age-sales correlations are statistically significant, the proportion of total variance accounted for by age is 1 percent or less. The age of a new firm does not help predict its contributions in jobs, sales, and exports.

#### Four Development Types

A different perspective on new firm contributions is to identify specific types of new firms and consider their individual features. One strategy for classifying new firms is to examine developmental patterns, focusing on two aspects: initial sales (high and low) and average annual growth (high and low).<sup>9</sup> Specifically:

- Start-up year sales: Classifying firms as starting with high or low first-year sales (at least \$250,000; or less than \$250,000).
- Annual sales growth: Classifying firms as slow or fast growing (average annual growth of more or less than \$100,000 per year).

The result is a fourfold classification of new firms as presented in Table 3.3.  $^{10}\,$ 

```	Average Annual (first yea		
· · · · · · · · · · · · · · · · · · ·	Low (< \$100,000)	High (> \$100,000)	Totals
First-year sales (start-up year) High: \$250,000 or more	7	10	17
Low: Less than \$250,000	66	17	83
Total	73	27	100

Table 3.3. DEVELOPMENT TYPES OF NEW FIRMS, IN PERCENTS (N = 676)

Two-thirds of the new firms have low first-year sales <u>and</u> low annual growth. About one in four has high annual growth, with the greater proportion

<sup>&</sup>lt;sup>9</sup> Average annual growth is highly correlated, in excess of 0.80, with com-

<sup>10</sup> pound annual growth rates. 10 These divisions are used for several reasons: they are just past the average value for each distribution; they are easy to remember and discuss; they were used in two other studies (1984 in Minnesota; 1986 in Pennsylvania) and thus facilitate comparison. However, this phenomenon is quite robust, and minor changes in the cut points would not affect the major patterns.

of high-growth firms coming from low-start firms. The smallest proportion of firms (one in fourteen) have high initial sales and low annual growth.

The distribution of these firms across industry sectors is presented in Figure 3.1. Low-start, low-growth firms are the most common type of new firm in every industry sector.<sup>11</sup> It is true, however, that low-start, low-growth new firms are less prevalent in manufacturing, distributive services, and construction than in other industries.



Figure 3.1. Development Type by Industry, 1979-84

# Development Type and Sales History

The purpose of this typology was to determine if there were some useful way to predict the developmental pattern or sales histories of new firms. Sales histories are presented in Figure 3.2.<sup>12</sup> The dramatic differences between the low-start-up, low-growth new firms and the high-growth new firms suggest that this typology has some value.

<sup>&</sup>lt;sup>11</sup> The small number of agriculture and mining firms, less than ten, is inadequate for much confidence in the patterns for these industries.

<sup>&</sup>lt;sup>12</sup> The number of firms available for the analysis drops steadily from the youngest to the oldest new firms. The number of firms of each age is indicated along the bottom axis.



Figure 3.2. DEVELOPMENT TYPE AND SALES HISTORY, 1979-84

Years Since Start-up

# Development Type and Contributions

The average 1985-86 contributions associated with the four developmental types are presented in Table 3.4. The average high-start, high-growth firm provides five times the jobs, twenty-one times the sales, and forty-two times the exports of the average low-start, low-growth firm.

		1985	1985	
	1986 Jobs	Sales (\$1,000)	Exports (\$1,000)	
High				
Low start, high growth	26.3	3,152	716	
High start, low growth	24.9	1,144	238 44	
Low start, low growth	5.3	150	17	

Table 3.4. DEVELOPMENT TYPE AND AVERAGE CONTRIBUTIONS, 1985-86

As only a minority of new firms have high-growth developmental patterns, it is possible their total contributions are small. The total contributions of the different types of new firms in the sample are presented in Figure 3.3. Fast-growing new firms--24 percent of all new firms--are responsible for 80 percent of sales and exports and 60 percent of the jobs provided by new firms. Because this sample is representative of all new firms emerging in Minnesota, confidence that this pattern would be found in all new firms should be high.



Figure 3.3. DEVELOPMENT TYPE AND TOTAL CONTRIBUTIONS, 1985-86

# Development Type and Minnesota Region

The occurrence of the different types of new firms in the thirteen regions of Minnesota is presented in Figure 3.4A. While some high-growth firms are present in every region, they are more prevalent in Region 11. As described in Chapter 2, this is the same region with a higher birth rate of firms in export potential industries (manufacturing, distributive services, and business services). (This is discussed further in sections C and D of this chapter.)

The proportion of new firms of each type emerging within each region is presented in Figure 3.4B. Low-start-up, low-growth new firms account for more than half of the new firms in every region, yet high-growth firms are also found in many of these regions. Only Region 2 (Headwaters) has no high-growth new firms in this sample. Across regions, the percentage of high-growth (low or high start-up) new firms varies from zero to 44 percent.



Figure 3.4A. DEVELOPMENT TYPE BY MINNESOTA REGION, AS OF 1986

-35-

# EXPORT FOCUS OF NEW FIRMS

For those concerned with regional economic growth, exporting firms provide additional advantages. The increased income can be considered a net economic contribution to the region, reflected in a direct, overall increase in jobs for the regional economy. There may be an indirect or multiplier effect, as well, namely, additional local market jobs created for each export-related job.

However, Minnesota exports may not constitute a net increase in jobs for the entire economic system--regional, national, or international. Unless the total number of jobs in the system increases, exports may represent only job displacement from one state or region to another.

Despite the controversy over net job creation versus job displacement, export-oriented new firms are clearly linked with external economic systems. Those national or international systems form the context for a regional economy.

The export orientation of new firms was a topic of some interest. The following discussion concerns the classification of new firms in terms of their export emphasis, the relative contributions of each such firm, and the regions where they have emerged.

#### Three Export Foci

The new firms were classified on the basis of their exports into one of three categories (see Table 3.5). In essence, a firm could be in an export category on the basis of either the volume of sales or the relative emphasis of total sales.<sup>13</sup>

There is substantial variation across industry sectors in the presence of export-oriented new firms, as presented in Figure 3.5A. The typical new firm in every industry is oriented toward an intrastate or local market. But three industries have a relatively high proportion of national and regional exporting new firms: manufacturing, distributive services (mostly wholesale), and business services.

The representation of industries among new firms with different export orientations is indicated in Figure 3.5B. Over 90 percent of national exporting new firms are in manufacturing, distributive services, and business services. These three industries include 65 percent of regional exporting new firms; another 25 percent are in construction and retail.

<sup>13</sup> Export sales volume was the major factor affecting the classification: 77 percent of the national export firms had \$100,000 or more in 1985 national sales and 77 percent of the regional export firms had \$100,000 in 1985 regional sales.

Table 3.5. TYPES OF EXPORT FOCI

	Percent All New Firms
National Exporter 50 percent or more 1985 sales outside Minnesota region (Minnesota and adjacent states); or \$100,000 or more in 1985 sales outside Minnesota region	10
Regional Exporter 50 percent or more 1985 sales outside Minnesota or \$100,000 or more in 1985 sales outside Minnesota	6
Intrastate Focus More than 50 percent of sales within Minnesota and less than \$100,000 sales outside Minnesota	84





New firms computing a replanal of mathemal warher have over twice the amployees and finst times the vales of introstate or local mathet new firmes National experting firms have over three these the total apports of regimes expecting new firms, decemen of the way the firms were classified, it is to be expecting new firms the average expects of intrastate new firms are simply dif



Figure 3.5B. MINNESOTA NEW FIRMS: EXPORT STATUS BY INDUSTRY SECTOR

# Export Focus and Contributions

Table 3.6 presents the average contributions (jobs, sales, and exports) of export-oriented new firms.

	Percent All New Firms	1986 Jobs	1985 Sales (\$1,000)	1985 Exports (\$1,000)
National exporters	10	14	1 451	849
Regional exporters	6	13	1,463	282
Intrastate focus	84	6	361	4

Table 3.6. Export Focus and Average Contributions

New firms targeting a regional or national market have over twice the employees and four times the sales of intrastate or local market new firms. National exporting firms have over three times the total exports of regional exporting new firms. Because of the way the firms were classified, it is to be expected that the average exports of intrastate new firms are almost nil. The total contributions of new firms with different export orientations are presented in Figure 3.6. Although export-oriented new firms make up only 16 percent of all new firms, they account for 45 percent of total sales, 27 percent of new jobs, and--to be expected--97 percent of all exports.



Figure 3.6. EXPORT FOCUS AND PERCENTAGE CONTRIBUTIONS

The contributions of exporting new firms in different industries vary substantially, as presented in Figure 3.7. National and regional exporting new firms account for over half the contributions in manufacturing, distributive services, and business services.

The same pattern can be described differently. Major contributions from exporting firms are provided by new firms in manufacturing, distributive services, and business services.





# Export Focus and Minnesota Regions

The emergence pattern of new firms, including those that export, varies across the thirteen regions of Minnesota (see Figure 3.8). The majority of exportoriented new firms are concentrated in a few regions with over 60 percent of all exporting new firms emerging in the Twin Cities metropolitan area.





#### GROWTH AND EXPORT FOCUS OF NEW FIRMS

Both export status and growth (development) patterns are associated with greater average and total new firm contributions. It is of some importance to determine the extent to which these are the same firms. Analysis of this issue starts with a two-way classification of new firms: by annual sales growth (high vs. low) and export focus (all exporting vs. local market).<sup>14</sup> The resulting distribution is presented in Table 3.7.

The typical new firm (66 percent of the sample), has a low annual growth and is oriented toward local markets. Over half of the high-growth new firms also concentrate on local markets. Of the exporting new firms, however, twothirds are high growth (12 percent of all new firms are high-growth exporters). Low-growth exporters account for only 7 percent (roughly one in fourteen) of new firms.

<sup>&</sup>lt;sup>14</sup> The classification schemes are the same as those presented in Tables 3.3 and 3.5.

The relationship of export-growth patterns to industry is presented in Figure 3.9A and B.

spite nachlogo	SALES	SALES DESTINATIONS .		
and the second second	Local Markets	Out of State	Total	10
Growth pattern				
High	15	12	27	
Low	66	7	73	
Total	81	19	100	

Table 3.7 GROWTH PATTERNS AND EXPORT FOCUS, IN 1986 (IN PERCENTS)

Note: Based on 648 new firms in sample.



Figure 3.9A. INDUSTRY AND EXPORT GROWTH PATTERNS, 1986

The Figure 3.9A indicates, as might be expected, that low-growth, localmarket new firms are dominant in three major industries (retail, consumer services, and construction); slightly more than half in business services; and less than half in the combination of manufacturing and distributive services. Figure 3.9B indicates that 90 percent of high-growth exporting new firms come from three industries (manufacturing, distributive services, and business services). However, among high-growth new firms (local market or exporting), five of the major industries are well represented (there are few from consumer or health, education, and social services).



# Figure 3.9B. GROWTH, EXPORT FOCUS, AND INDUSTRY, 1986

#### Growth, Export Focus, and Contributions

The average contribution for each growth-export type is presented in Table 3.8. The low-growth, local-market new firms have substantially smaller average contributions. The major differences in average jobs and sales are between the high- and low-growth categories, rather than the difference in export orientations. The average 1986 jobs and 1985 sales for high-growth new firms are 18 and \$1,732,000 respectively; for low-growth new firms the averages are 6 and \$227,000.<sup>15</sup> Low-growth exporting new firms account for substantially greater average 1985 exports than other low-growth new firms (\$171,000 vs. \$4,000), but average 1986 jobs and 1985 sales are comparable to those of other low-growth new firms.

 $<sup>^{15}</sup>$  Based on the average of categories, not weighted by the number of firms.

and the set of the first	Percent	A KXY	1985	1985
	all Nov Firms	1986	Sales	Exports
The should be a series	New PILLIS	0005	(91,000)	(91,000)
High growth, exporting	12	21.1	2,272	933
Low growth, exporting	7	6.5	261	171
High growth, local market	15	14.0	1,193	10
Low growth, local market	66	6.4	192	4

Table 3.8. Growth, Export Focus, and Average Contributions

The aggregate or total contributions of new firms classified by growth and export focus are presented in Figure 3.10. The high-growth firms dominate every type of contribution: they represent 28 percent of all new firms, and they provide 70 percent of total jobs, and 85 percent of total sales and exports. The remaining major source of exports is the low-growth exporting firms: they account for 7 percent of all new firms and provide 15 percent of the exports.



Figure 3.10. GROWTH, EXPORT FOCUS, AND TOTAL CONTRIBUTIONS

#### Growth, Export Focus, and Minnesota Regions

The proportion of all new firms with different export-growth emphases emerging in the thirteen Minnesota regions is presented in Figure 3.11A. Local market, low-growth new firms dominate in every region. The proportion of exportoriented new firms is more varied. As Figure 3.11B indicates it is clear that export-oriented new firms are being established throughout the state. Most, however, are established in the Twin Cities region.

# 100 Export, high growth Percentage Each Type Local, high growth 80 Export, low growth 60 Local, low growth 40 20 0 9-Region Nine 0-Southeastern 5-Region Five 6W-Six West 7W-Central MN 8-Southwest 11-Met Council 2-Headwaters 6E-Six East 7E-E Central 3-Arrowhead 4-W Central -Northwest

Figure 3.11A. GROWTH AND EXPORT FOCUS BY MINNESOTA REGIONS

The second strategy was to classify new firms in terms of critical characteristical developmental parterne (average annual value growth) and argent orientation (amphanis on sub-of-state sales). The results were work strating, for the average bigh-growth new firm was substantially internet the the everage low-growth new firm, and, of even more significance, they ware tappenables for the majority of all contributions about two-thirds of the high-growth new firms were export offented, and secondented for feur-firthe s high-growth and firms were export offented, and secondented for feur-firthe s all seconds.



Figure 3.11B. ALL MINNESOTA NEW FIRMS BY GROWTH AND EXPORT FOCUS

#### SUMMARY AND IMPLICATIONS

Two major factors affect the contributions (jobs, sales, and exports) of new firms: their number and size. The substantial variation in the contributions of new firms led to attempts to explain, and predict, their potential roles.

Two strategies were used to understand and explain the differences in the average contributions of new firms. The first assumed that all new firms were equal in their potential. It involved attention to a number of variables (142, reduced to 42) considered useful for predicting variations in jobs, sales, and exports. While the capacity to predict variation in jobs and sales was high, the number of variables was large and few seemed obvious candidates for a policy intervention that would have a major impact.

The second strategy was to classify new firms in terms of critical characteristics: developmental patterns (average annual sales growth) and export orientation (emphasis on out-of-state sales). The results were quite striking, for the average high-growth new firm was substantially larger than the average low-growth new firm, and, of even more significance, they were responsible for the majority of all contributions. About two-thirds of the high-growth new firms were export oriented, and accounted for four-fifths of all exports.

A larger percentage of high-growth and export-oriented new firms emerged in the major urban region, the Minneapolis-St. Paul metropolitan area. However, a number of high-growth and export-oriented new firms have emerged throughout the state.

Policy strategies for enhancing new firms and, in turn, economic development would include:

- Improving the economic (or business) climate for all new firms.
- Attempting to provide specific assistance for those new firms with the greatest potential for societal contributions.
- A combination of these two approaches.

Implementation of any universal strategy is likely to be expensive for two reasons. (1) A large number of new firms would be eligible for service: 5,000-10,000 in Minnesota every year. (2) The analysis of factors related to new firm contributions has failed to disclose any with major effects, i.e., none that can be expected substantially to improve jobs, sales, or exports. In sum, improving the overall "business climate" may be difficult and have only modest effects on new firm contributions.

The alternate strategy--providing specific assistance to selected highpotential new firms--can be implemented at any level. Only a few highpotential new firms, or a larger proportion of high-growth firms, might receive assistance. A key feature of such a selective strategy would be the identification of high-potential new firms to receive attention.

Developing procedures for identifying high-potential new firms should be Possible. Two results from this project are encouraging. First, high annual sales growth--the key indicator--is obvious early in the life of a new firm; usually within twenty-four months of initial sales. Second, the procedures used in this project located and identified such new firms using readily available public data. If it can be done for research purposes, it can be done for a public program.

Furthermore, programs focusing on new firms with both high growth and a national export orientation should give careful attention to three major industries (manufacturing, distributive services, and business services). High-growth regional exporters are also found in construction (perhaps in border communities) and retail. High-growth, local market new firms are to be found in most industry sectors, the major exception being consumer services.

#### Chapter 4

# START-UP PROBLEMS: NATURE AND CURRENT STATUS

The respondents, who were involved in starting the new firms, were asked to identify major start-up problems from a list of forty-one; for each problem they mentioned, an opportunity was provided to indicate whether it was currently solved.

The results are presented in Table 4.1, ranked in descending order. The average ranking for each problem is also provided; the correlation with the percentage indicating the problem was "major" is extremely high.

The percentage currently solved is also presented. The less frequently a problem was said to be "major," the more likely it is considered currently solved.

The problems at the top of the list tend to involve financial management, locating employees, and obtaining external financial support. As these problems were presented in groups and clearly overlap, analysis was completed to see if they could be classified into major categories. Not only could they be placed into nine major groups, but these groups were very similar to those developed in other comparable surveys of new firms and their problems.

# START-UP PROBLEM DIMENSIONS

The major dimensions, reflecting the major types of problems, that emerged from the factor analysis were as follows:

<u>General Label</u>	No. Items in Factor	Average % Major Problem	Average Severity	Average % Fully Solved	
P:	<b>,</b> .	0.6	0.00	20	
rinancial support	4	36	2.06	39	
Marketing strategy	11	23	2.01	25	
Manage cash	4	36	1.99	38	
Government relations	3	25	1.96	41	
Develop/implement plan	6	15	1.92	25	
Personnel/cohesion	6	20	1.91	31	
Site	2	17	1.86	73	
Access	3	9	1.65	47	
Infrastructure	2	6	1.50	60	
Unions	1	6	1.44	39	
_					

Table 4.1. START-UP PROBLEMS: RANKED BY SEVERITY

	Percentage	Rating Pr	oblem As	Average	Percent	Percent	Percent
	Hajor	Minor	Never	Severity	Fully	Partly	Not
	(3)	(2)	(1)		Solved	Solved	Solved
Obtaining equity financing	40	38	22	2.11	33	44	19
Securing operational financing	39	41	20	2.11	36	45	8
Obtaining debt financing	39	41	21	2.09	34	41	10
Managing cash flow	34	52	14	2.16	28	60	15
Finding competitive advantages	32	54	14	2.19	24	64	13
Effective selling techniques	31	52	16	2.11	17	68	10
Obtaining liability insurance	28	47	25	2.01	45	32	26
Developing new products/services	27	57	16	2.06	- 14	67	5
Pricing products/services	27	56	17	2.05	28	64	10
Finding qualified managers/executives	27	41	32	1.99	27	55	9
Developing accounting/control systems	25	60	15	2.04	46	45	13
Finding qualified employees	25	55	21	2.03	30	55	23
Finding qualified technical staff	25	51	24	2.01	26	60	18
Establish a banking relationship	25	49	26	1.92	50	37	19
Analyzing competition	24	63	13	2.10	23	67	18
Collecting accounts receivable	24	57	19	2.06	23	59	9
Coping with government regulations	24	57	19	2.05	37	46	5
Writing ad copy, selecting media	22	56	22	2.02	21	61	19
Obtaining real estate financing	22	33	45	1.81	41	36	8
Delivering on time/within budget	21	57	23	1.92	28	64	14
Understanding industry trends	20	58	23	1.97	24	67	23
Assessing customer needs	19	60	21	1.96	29	66	8
Managing capital	18	60	22	1.94	31	61	13
Locating suitable rental space	18	56	26	1.89	73	22	13
Preparing a business plan	16	63	21	1.94	36	54	23
Implementing plans/objectives	16	67	17	1 93	19	71	17
Identifying/selecting a suitable site	16	53	31	1 82	72	22	1
Using/updating a business plan	15	63	22	1 92	25	62	10
Measuring performance against plans	15	64	22	1 91	20	65	10
Motivating/compensating personnel	15	61	24	1 88	29	63	14
Providing after-sale service	15	59	26	1 87	32	59	10
Minimizing start-up team conflict	15	41	44	1 70	39	53	6
Setting goals for personnel	14	68	18	1 93	23	83	15
Clarifying goals/objectives	13	67	19	1 90	20	00 66	10
Access to customers	13	51	37	1 74	46	00 40	5
Coordinating tasks among units	12	64	24	1 87	32	40 62	8
Providing customer service	12	64	24	1 83	30	67	6
Selecting accountant/lawyer	10	61	30	1 76	00 03	27	23
Access to suppliers	7	52	41	1 62	52	44	20 5
Access to employees	, ,	47	45	1 59	52 4.4	77 51	5 5
Appropriate transportation structure	6	19	52	1 59	17 13	35	5
Appropriate infrastructure	6	37	58	1 47	50 50	00 / 1	U K
Developing relationships with unions	6	26	. 68	1.44	39	39	5 1
• • • • • • • • • • • • • • • • • • •	•	20		1.11		00	Ĩ
Average	20	54	26	1.91	35	53	12

The details of the items, along with each dimension, are presented in Appendix E, as are the characteristics of the distribution and reliability of each dimension.

The range of problem severity and solved (or unsolved) problems is much less among the dimensions than among individual items. In particular, the relationship between "severity" and "solved" is reduced substantially. This suggests considerable variation in the "solution status" of problems in each dimension among the firms in the sample.

# VARIATION AMONG MINNESOTA REGIONS

The average start-up problem severity is presented for each Minnesota region in Figure 4.1. Considered as a group, the severity of start-up problems is lower for the Greater Minnesota regions than in the Twin Cities metropolitan region. This may reflect, as much as anything else, the lack of high-growth new firms in the Greater Minnesota regions. High-growth new firms tend to report more start-up problems. It could also reflect the strikingly low severity for the East Central region (7E), although this region has a small number of firms.



Figure 4.1. START-UP PROBLEM SEVERITY FOR MINNESOTA REGIONS

# START-UP PROBLEMS AND SOCIETAL CONTRIBUTIONS

A key issue is, of course, the relationship between the severity of start-up problems and the societal contributions (jobs, sales, and exports) made by the firms. The simplest measures of association, product moment correlations, are presented in Table 4.2. As major problems were given a rating of 3, and those that never occurred a rating of 1, a positive correlation would suggest that more severe start-up problems are associated with more contributions.

	Problem Severity Correlated			
Start-up Problem	1986 1985		1985	
Dimension <sup>a</sup>	Jobs	Sales	Exports	
Managan	0.1 ( shales to	0 1/	0.07	
Manage cash	0.14***	0.14**	-0.07	
Marketing strategy	0.15***	0.07	0.04	
Financial support	0.19***	0.17**	0.29***	
Government relations	0.23***	0.20***	0.34***	
Develop/implement plan	0.14***	0.11*	0.13	
Personnel/cohesion	0.21***	0.15**	0.27**	
Site	0.07	0.08	0.21**	
Access	0.09*	0.03	0.13	
Infrastructure	0.16**	0.15**	0.20*	

Table 4.2. START-UP PROBLEMS AND SOCIETAL CONTRIBUTIONS

#### Notes:

- Pearson Product Moment correlations computed after the measures of social contributions transformed to log 10 to produce a normal distribution. Since most firms do not have exports, this reduced the sample of these firms to 442. Sample was weighted to reflect the distribution of the population of new firms.
- 2. Statistical significance, one-tailed: \*\*\* p > 0.000; \*\* 0.01 > p > 0.001; \* 0.05 > p > 0.01.

<sup>a</sup> Scale: 3, major; 2, minor; 1, never occurred.

In fact, the major pattern is one of more start-up problems reported by firms making the most significant contributions. And it is almost universal. Nineteen of twenty-seven correlations are statistically significant, and all but one correlation is actually positive. There is no start-up problem dimension that is not statistically significantly related to at least one type of contribution.

In short, new firms making the greatest contributions--those growing the fastest--report more serious problems at start-up.

# SOLUTION OF START-UP PROBLEMS

Given that high-contributing new firms have more start-up problems, are they solving these problems? A preliminary analysis is provided in Table 4.3. It presents the simple correlations between the absolute number of fully solved start-up problems and the contributions (jobs, sales, exports) at the time of the interview.

with Cont:     6   1985     s   Sale:     02   0.02     10*   0.01	ributions 1985 s Exports 0.04
6 1985 s Sales 02 0.02 10* 0.01	1985 <u>s Exports</u> 0.04
s Sale: 02 0.02 10* 0.01	s Exports
02 0.02 10* 0.01	0.04
10* 0.01	
	-0.10
06 0.16	*** 0.08
02 -0.02	-0.02
0.02	0.00
10 0.01	-0.10
08 0.01	0.02
11* 0.14	** 0.10
06 0.20	* 0.06
	*** 0.14**
	10 0.01   08 0.01   11* 0.14   06 0.20   12** 0.16

Table 4.3. START-UP PROBLEM SOLUTION AND CONTRIBUTIONS

# Note:

1. Statistical significance, one-tailed: \*\*\* - p > 0.000; \*\* 0.01 > p > 0.001; \* 0.05 > p > 0.01.

There are fewer statistically significant relationships (five out of twenty-seven correlations), but most are in the expected direction. The negative correlations regarding marketing strategy may actually reflect the importance attached to marketing and, in turn, the number of marketing problems identified as continuing problems. In those firms where marketing problems were considered easily solved, the growth and contributions could have been more modest. Particularly important is the aggregate count across all dimensions (all forty-one problems mentioned). The more problems solved, the greater the contributions.

# SUMMARY

The inventory of start-up problems seems to capture most of the concerns that could be expressed during the initiation of a new firm. The forty-one items can be successfully summarized by nine general dimensions. There is little major variation in start-up problem severity associated with Minnesota regions. A pervasive and significant pattern was found between the severity of specific start-up problem dimensions and the level of sales, jobs, and exports provided to the Minnesota economy. Firms making greater contributions report more problems of every type. Further, firms providing greater contributions reported solving a larger number of start-up problems.

#### Chapter 5

#### MINNESOTA INFRASTRUCTURE

When asked why they started a new firm in Minnesota, the most frequent specific response from those surveyed was:

#### I live here!!!

The most common theme, mentioned by half of the respondents, relates to their knowledge of, or familiarity with, the area as a context for pursuing business. Spontaneous comments on their reasons for locating in Minnesota and at a specific site are presented in Table 5.1. These responses are not affected by the region of Minnesota or industry sector of the new firm. People tend to start new firms where they have established personal, family, and business relationships.

Firms chosen for the survey were selected on the basis of a Minnesota <sup>start-up</sup>, not the residence of those starting the firm. Yet there is virtually no evidence that anyone moved to Minnesota to start a new firm.

Different themes are reflected in comments about the choice of a specific site for the new business. The comments summarized in Table 5.1 indicate that the relative attractiveness of the location is critical.<sup>1</sup> "Live here!" includes the one in twelve (8 percent) that started a business in the home; the other one in twelve (another 8 percent) presumably is referring to the neighborhood or community.

The universal choice of "home" as a place to start a new business does not mean that "home" is perfect. The following will review selected features of the immediate context in which new firms start their business, referred to as local infrastructure. Both the importance and satisfaction with twenty different aspects of infrastructure are examined.

A number of services might be provided to new firms by local, regional, and state governments. Reactions to these services conclude this chapter.

1 Most sites are assumed to be within commuting distance of the homes of the start-up team members.

Reason Given	For Starting in Minnesota	For Choosing Specific Site .
Location specific Central location Access to customers, suppliers, trans- portation, home, business contacts, metropolitan area, important markets	41	62
Live here Work at home	41	16
Lack of (inadequate) competition	15	3
Familiar with area Know markets, business contacts	11	5
Site attractiveness Land, building suitable Cost appropriate Enough space	8	39
Owned property (land/structure)	2	5

#### Table 5.1. REASONS FOR NEW FIRM LOCATION (in percents)

Note: Multiple responses allow for column totals greater than 100 percent.

#### LOCAL INFRASTRUCTURE FEATURES: IMPORTANCE AND SATISFACTION

Respondents were asked about the relative importance of twenty different aspects of the infrastructure in which the new firm operates. The importance assigned to these is presented in Table 5.2.

Access to customers is clearly the most important infrastructure feature; four of five (83 percent) considered it "very" important. Only one of twenty (5 percent) considered it not important. Access to customers is in a class by itself.

Also in a class by itself is access to research and development facilities. Only one in five (18 percent) considered it very important, and over half (57 percent) considered it not important. One other feature is considered almost as unimportant: land availability (for expansion, etc.).

The remaining features fall between these extremes. Five could be considered of high importance: taxes; quality of life; access to suppliers; capital availability; and labor costs. Six are of moderate importance: local government support for business; availability of highly skilled workers; energy reliability; energy costs, transportation; and building space expenditures. Six are of low importance: land purchase, rental costs; local regulations; educational and training opportunities; building space availability; physical infrastructure; and zoning and land use.

	Importance (in percents).			*******
	Some-		Average	
	Very	what	Not	Impor-
	(2)*	(1)*	(0)*	tance .
Access to customers	83	12	5	1 78
Taxes	67	24	q	1 59
Quality of life	64	27	10	1 54
Access to suppliers	62	24	14	1 49
Capital availability	59	28	14	1 45
Labor costs	59	26	16	1 43
Local government support for husiness	54	28	18	1 36
Availability of highly skilled workers	54	23	23	1 31
Energy reliability	47	33	20	1 28
Energy costs	41	40	18	1 23
Transportation (highways railroads)	42	37	21	1 21
Building space expenditures (rent etc.)	43	32	25	1 18
Land purchase reptal costs	40	30	23	1 16
Local regulations	38	39	27	1 16
Educational and training opportunities	36	34	30	1 07
Building space availability	35	37	28	1 07
Physical infrastructure (roads water)	34	38	28	1 06
Zoning and land use	33	35	32	1 01
Land availability (for expansion etc.)	26	32	52 40	0.83
Access to research and development facilities	18	25	57	0.05
to research and development facilities	TO	23	51	0.00
Average	46	30	24	1.21

# Table 5.2. IMPORTANCE OF LOCAL INFRASTRUCTURE FEATURES

\*Importance graded on a scale of 0 to 2 as indicated in parentheses.

Note: Row totals may not equal 100 percent due to rounding off.

у

11

Satisfaction with these same twenty infrastructure features is presented in Table 5.3. Two aspects rate very high on satisfaction: access to customers and quality of life. Over half the respondents indicated they were very satisfied with these features. Seven features had above average satisfaction ratings: energy reliability; access to suppliers; transportation; physical infrastructure; building space availability; labor costs; educational and training opportunities. Eight had below average satisfaction ratings: availability of skilled workers; land purchase, rental costs; land availability, building space expenditures; zoning and land use; capital availability; local regulations; energy costs; and access to research and development facilities.

	Satisfaction (in percents) .			
	Some-			Average
	Very	what	Not	Satis-
	(2)*	(1)*	(0)*	faction.
Ouality of life	52	36	8	2.49
Access to customers	53	39	7	2.46
Energy reliability	47	46	7	2.39
Access to suppliers	47	42	11	2.35
Transportation (highways, railroads)	43	49	8	2.35
Physical infrastructure (roads, water)	41	48	11	2.30
Labor costs	33	59	9	2.24
Building space availability	36	50	15	2.21
Education and training opportunities	32	50	19	2.13
Availability of highly skilled workers	32	48	20	2.11
Land purchase, rental costs	25	59	16	2.10
Land availability (for expansion, etc.)	31	48	21	2.09
Building space expenditures (rent, etc.)	25	58	18	2.07
Zoning and land use	23	57	20	2.04
Capital availability	27	46	27	2.00
Local regulations	24	53	22	1.98
Energy costs	17	62	21	1.96
Access to research and development facilities	20	48	32	1.87
Local government support for business	17	44	39	1.78
Taxes	5	33	62	1.44
Average	33	48	19	2.12

# Table 5.3. SATISFACTION WITH LOCAL INFRASTRUCTURE FEATURES

\*Satisfaction graded on a scale of 0 to 2 as indicated in parentheses.

Note: Row totals may not equal 100 percent due to rounding off.

Two features are quite distinctive for their very low rating: taxes and local government support for business. Reported dissatisfaction with taxes is widespread, with almost two of three indicating "not satisfied."

The relationship between the relative importance and relative satisfaction with these twenty aspects of the local infrastructure is presented in Table 5.4. The overall association is positive; those factors considered more important tend to be those with the highest reported satisfaction. For example, access to customers, distinctive in importance, also leads the list in terms of satisfaction.

But there are two major exceptions. There is strong agreement that both taxes and local government support for business are relatively important, yet satisfaction with both is very low. Low satisfaction with taxes is most notable in this regard.

The reason for this low satisfaction is open to speculation. It may reflect a resentment of taxes or the failure of local government to provide timely, appropriate support for new business firms. It may be that those starting new businesses do not understand the basis for the various taxes or how they might obtain help from the local government.

	Satisfaction			
		Below	Above	
Importance	Very Low	Average	Average	Very High
Very high				Access to customers
High	Taxes	Capital availability	Labor costs	Quality of life
			Access to suppliers	
Medium	Local govern- ment support for business	Building space expenditures	Availability of skilled workers	
		Energy costs	Energy reliability	
			Transporta- tion	
Low		Local regulations	Physical infrastruc- ture	
· · · · · · · · · · · · · · · · · · ·		Land purchase, rental costs	Education, training opportuniti	es
		Zoning and land use	Building space avail ability	-
Very Low		Land availability		
		Access to research and development facilities		

5

е

Table 5.4. SATISFACTION WITH AND IMPORTANCE OF INFRASTRUCTURE FEATURES

-59-

#### MINNESOTA REGION AND INFRASTRUCTURE SATISFACTION

It is always possible that satisfaction with infrastructure features may vary by region. For example, there may be differences in labor costs or road conditions. The average satisfaction with the local infrastructure is presented in Figure 5.1. The most significant pattern is the clear difference between the Twin Cities metropolitan region and all the regions of Greater Minnesota; overall satisfaction with the infrastructure is slightly higher in the Greater Minnesota regions. Because of the small number of firms in the Greater Minnesota regions and the small variation in satisfaction (range of 2.05 to 2.30), differences among the nonmetropolitan regions are not significant.

The most appropriate conclusion may be that there is little effect of the specific region on satisfaction with infrastructure features. The level of satisfaction (or dissatisfaction) is quite uniform across all regions of Minnesota.



Figure 5.1 MINNESOTA REGION AND INFRASTRUCTURE SATISFACTION
### TYPE OF NEW FIRM AND REACTIONS TO INFRASTRUCTURE

It was clear from the analysis in Chapter 3 that some new firms make substantially greater contributions (jobs, sales, exports) than others. New firms with high growth rates and an export orientation were distinctive in their contributions. For this analysis the sample has been divided into four groups: (1) export orientation and high growth; (2) export orientation and low growth; (3) local market and high growth; (4) and local market and low growth. The importance attached to these twenty infrastructure features by those starting these different types of firms is presented in Table 5.5; statistical significance is indicated in the last column.

	Ext	port	Local	Market
	High	Low	High	Low
	Growth	Growth	Growth	Growth
Access to customers	1.62	1.49	1.73	1.87***
Taxes	1.59	1.51	1.73	1.59
Quality of life	1.51	1.56	1.61	1.56
Capital availability	1.60	1.32	1.48	1.48
Access to suppliers	1.40	1.38	1.45	1.56
Labor costs	1.57	1.14	1.69	1.36***
Availability of highly skilled workers	1.38	1.03	1.60	1.26**
Local government support for business	1.19	1.00	1.38	1.46**
Energy reliability	1.28	1.04	1.24	1.34
Building space expenditures (rent, etc.)	1.49	1.10	1.10	1.21*
Transportation (highways, railroads)	1.31	1.22	1.17	1.22
Energy costs	1.18	1.07	1.29	1.28
Land purchase, rental costs	1.32	0.93	1.29	1.13*
Educational and training opportunities	1.18	0.88	1.09	1.04
Local regulations	1.06	0.94	1.33	1.20*
Building space availability	1.28	0.79	1.22	1.12**
Physical infrastructure (roads, water)	1.07	1.01	1.02	1.12**
Zoning and land use	1.00	0.85	0.96	1.03
Land availability (for expansion, etc.)	1.10	0.61	0.90	0.89*
Access to research & development facilities	0.72	0.77	0.62	0.59
Average	1.29	1.09	1.30	1.27*

Table 5.5. EXPORT, GROWTH STATUS, AND INFRASTRUCTURE IMPORTANCE, BY NEW FIRM TYPE

Note: Statistical significance: \*\*\* - p <0.0000; \*\* - 0.0000 p > 0.001. Statistically significant differences in the importance of the infrastructure features are presented for eight of the twenty items.<sup>2</sup> They are summarized in the accompanying tabulation.

	Consider More Important	Consider	Less	Important
High-growth new firms	Labor costs			
(export & local market)	Land purchase, rental costs			
	Building space expenditures			
Local market new firms (high and low growth)	Access to customers Local government support Local government regulations	5		
Exporting, low-growth new firms		Building ability Skilled ability Land ava	space y worker y ilabi:	e avail- r avail- lity

High-growth new firms (exporting and local market) tend to place greater emphasis on labor costs, land purchase and rental costs, and building space expenditures. Those emphasizing a local market (high and low growth) have a major concern with access to customers, local government support, and local government regulations. Perhaps most distinctive are the low-growth, exportoriented new firms: they consider a number of factors less important than other types of new firms, including the availability of building space, skilled workers, and land for expansion.

Average satisfaction with these local infrastructure features for the four types of new firms is presented in Table 5.6. There are statistically significant differences for only two of the twenty features. Both provide a distinctive picture, indicating that low-growth, exporting and high-growth, local market new firms are more satisfied with access to research and development facilities and less satisfied with taxes.

<sup>&</sup>lt;sup>2</sup> The test of significance, a one-way analysis of variance, indicates only that the differences would probably not have occurred by chance. Interpretation is required to determine the basis for the statistically significant differences.

	Export		Local Market		
	High	Low	High	Low	
	Growth	Growth	Growth	Growth	
Quality of life	2.49	2.64	2.55	2.51	
Access to customers	2.44	2.27	2.53	2.49	
Energy reliability	2.32	2.36	2.29	2.37	
Access to suppliers	2.22	2.26	2.38	2.35	
Transportation (highways, railroads)	2.41	2.47	2.31	2.34	
Physical infrastructure (roads, water)	2.33	2.25	2.32	2.32	
Building space availability	2.22	2.20	2.26	2.22	
Labor costs	2.21	2.14	2.16	^ 2.24	
Education and training opportunities	2.19	2.12	2.10	2.11	
Availability of highly skilled workers	2.18	2.17	2.04	2.17	
Land availability (for expansion, etc.)	2.11	2.29	2.22	2.07	
Building space expenditures (rent, etc.)	2.06	2.03	2.07	2.07	
Land purchase, rental cost	2.08	1.88	2.12	2.07	
Zoning and land use	2.03	1.85	1.94	2.02	
Capital availability	2.04	2.05	1.89	2.02	
Local regulations	1.92	1.92	1.97	1.97	
Energy costs	1.92	2.07	1.93	1.95	
Access to research-development facilities	1.81	2.03	2.07	1.84*	
Local government support for business	1 66	1 67	1 71	1 81	
Taxes	1.38	1.27	1.28	1.46*	
	2.00			2	
Average	2.11	2.09	2.11	2.12	

# Table 5.6. EXPORT, GROWTH STATUS, AND INFRASTRUCTURE SATISFACTION, BY NEW FIRM TYPE

# REACTION TO POTENTIAL GOVERNMENT SERVICES

}-1t A range of services and assistance may be provided to new firms by state, regional, and local governments. The reaction to thirty-one such services in terms of interest and value was obtained from most respondents. The average response and proportion showing a high level of interest or perceived value are presented in Table 5.7. They are rank ordered by the average response.

The range of responses is considerable, almost one in three showing a high interest in marketing skills development and working capital financing. About one in sixteen (6 percent) show a high interest in small-business incubator sites, energy development opportunities, infrastructure financing, and energy audits, conservation projects.

	Percent	
	with High	Average
	Interest	Interest
Marketing skills development	26	1.59
General business management development	21	1.51
Working capital financing	29	1.50
Finance skill development	24	1.49
Personnel management skill development	22	1.48
Training new employees	18	1.24
Knowledge of government regulations	14	1.22
AVTI small-business programs <sup>a</sup>	16	1.19
Machinery, equipment financing	17	1.13
Small-business development centers	14	1.08
Community college small-business programs	12	1.08
Entrepreneurial training	13	1.06
Building construction financing	19	1.06
Skills in labor-management relations	10	0.95
Retraining existing employees	11	0.94
New technology for productivity gains	13	0.93
Venture, seed capital financing	16	0.92
Land acquisition financing	13	0.81
Employee ownership financing	9	0.78
New site locations within a county	8	0.71
Engineering, scientific skill development	6	0.70
New site locations within Minnesota	7	0.62
Energy audits, conservation projects	7	0.59
Applying company's research and development in new marke	ets 8	0.57
Federal procurement assistance	9	0.57
Infrastructure financing	6	0.55
Tourism market development	10	0.54
Energy development opportunities	6	0.53
Joint research and development product development	6	0.53
Small-business incubator sites	6	0.55
Develop foreign export markets	6	0.39
	<b>v</b> .	0.00
Average	14	0.93

# Table 5.7. INTEREST IN AND VALUE OF POTENTIAL GOVERNMENT SERVICES

Note: Interest scale: High, 3; Moderate, 2; Low, 1; None, 0.

<sup>a</sup> AVTI refers to area vocational technical institutes, a post-high school training system in operation in Minnesota.

The large number of new firms initiated each year in Minnesota suggests that the absolute numbers of potential clients for these services may be substantial. Assuming that 10,000 new business entities are initiated annually in Minnesota, as many as 600 may be interested in even the least popular service. The more popular would attract the interest of 3,000 each year. Over a single decade, this would be a potential client base of 6,000-30,000. The more popular programs are likely to be related to financial assistance or programs to provide training and development of management, administrative, or supervisory skills.

The relative interest in these thirty-one services by the four types of new firms is presented in Table 5.8. There are statistically significant differences among types of new firms for twenty-one of the thirty-one services.

Thirteen, associated with increased interest among high-growth new firms, are summarized in the accompanying tabulation:

All High Growth	Exporting	Local Market
(export and local market)	(high and low growth)	High Growth
Personnel management skill development Knowledge of government regulations Building construction financing Skills in labor- management relations Retraining existing employees Land acquisition financing Training new employees Engineering, scientific skill development	Applying company's research and de- velopment in new markets Joint research and development product development Develop foreign export markets	New site locations within the county New site locations within Minnesota Federal procurement assistance

This reflects a pattern of interest consistent with the distinctive Problems associated with different forms of growth. For example, local markets high-growth new firms are the only ones with a special interest in site location assistance; exporting firms are interested in assistance with developing foreign export markets.

### Table 5.8. EXPORT, GROWTH STATUS, AND INTEREST IN AND VALUE OF POTENTIAL GOVERNMENT SERVICES, BY NEW FIRM TYPE

	Export		Local Market	
	High	Low	High	Low
· · · · · · · · · · · · · · · · · · ·	Growth	Growth	Growth	Growth
Marketing skills development	1.90	1.76	2.08	1.39***
General business management development	1.71	1.71	1.92	1.41**
Working capital financing	2.00	1.86	1.92	1.31***
Finance skill development	1.66	1.58	1.71	1.39
Personnel management skill development	1.78	1.59	1.87	1.43*
Training new employees	1.70	1.07	1.65	1.07***
Knowledge of government regulations	1.46	1.21	1.63	1.04*
AVTI <sup>a</sup> small-business programs	1.05	1.15	1.25	1.20
Machinery, equipment financing	1.34	1.48	1.27	1.07
Small-business development centers	0.84	1.03	1.12	1.12
Community college small-business programs	1.05	1.10	1.23	1.04
Entrepreneurial training	1.24	1.16	1.39	1.03
Building construction financing	1.33	0.83	1.39	0.95*
Skills in labor-management relations	1.25	0.90	1.32	0.88*
Retraining existing employees	1.31	0.89	1.55	0.78***
New technology for productivity gains	1.40	1.45	1.30	0.77***
Venture, seed capital financing	0.87	1.16	1.18	0.79*
Land acquisition financing	1.00	0.80	1.11	0.70*
Employee ownership financing	1.14	0.50	0.74	0.78*
New site locations within a county	0.43	0.40	1.19	0.59***
Engineering, scientific skill development	0.90	0.78	0.95	0.60*
New site locations within Minnesota	0.69	0.69	1.02	0.51**
Energy audits, conservation projects	0.82	0.65	0.64	0.60
Applying company's research and			-	
development in new markets	0.97	1.17	0.63	0.47***
Federal procurement assistance	0.78	0.69	0.96	0.40***
Infrastructure financing	0.83	0.57	0.71	0.50
Tourism market development	0.24	0.43	0.40	0.62*
Energy development opportunities	0.67	0.40	0.76	0.51
Joint research and development		••••		
product development	0.77	0.84	0.56	0.46*
Small-business incubator sites	0.46	0.37	0.73	0.48
Develop foreign export markets	0.77	0.67	0.29	0.31**
Average	0.85	1.00	1.11	1.11**

Notes: Interest scale: High, 3; Moderate, 2; Low, 1; None, 0.

Statistical significance: \*\*\* - p <0.0000; \*\* - 0.0000 p > 0.001.

<sup>a</sup> AVTI refers to area vocation technical instututes, post-high school training system in operation in Minnesota.

Two other statistically significant differences reflect high interest: a greater interest in employee ownership financing by high growth exporting new firms and a greater interest in tourism market development by low-growth local market new firms. While local market new firms show a generally greater interest in government services than export-oriented new firms, they are significantly lower on five: marketing skills development, general business management development, working capital financing, training new employees, and engineering and scientific skill development.

The interest in specific programs does not reflect the interest of individual firms. Variations among different types of new firms are presented in Table 5.9.

	Average		Percent 1	Intereste	d In	
	Number	None	1-4	5-9	10-14	15-23
All new firms	3.85	30	36	22	9	3
High start-high growth	5.52	17	37	23	12	10
Low start-high growth	4.87	16	42	28	11	3
High start-low growth	2.34	38	39	. 20	3	1
Low start-low growth	3.50	35	34	20	9	2
National exporter	3.41	30	39	21	8	2
Regional exporter	5.54	7	39	30	22	2
Local market	3.84	30	39	21	8	2
Exporter-high growth	4.48	19	39	26	15	
Local market-high growth	n 5.67	14	40	28	9	9
Exporter-low growth	3.81	25	39	22	13	1
Local market-low growth	3.42	36	33	21	8	3
Greater Minnesota	4.45 <sup>a</sup>	25	36	25	10	4
Twin Cities metro regior	n 3.63	30	40	20	8	2

Table 5.9. NEW FIRM INTEREST IN GOVERNMENT SERVICES

Note: All four comparisons have a statistically significant variation, p equal to or smaller than 0.05.

<sup>a</sup> Average of the average for the twelve Greater Minnesota regions.

١g

The typical respondent indicated a high level of interest in almost four of these services. Seven in ten indicated a high level of interest in one or <sup>more</sup> programs and services; one in eight indicated a high level of interest in ten or more.

Interest in help from government services and programs is high for all <sup>major</sup> types of new firms. There are some variations related to the growth <sup>rate</sup> of the new firms, an emphasis on regional exports, and a location outside

the Twin Cities metropolitan region. High-growth new firms indicate more interest in government services, as do firms emphasizing regional markets and new firms located outside the Twin Cities metropolitan area.

#### SUMMARY

Major findings related to the contexts in which new firms operate include the following:

- New firms are started by people that live in the area. There is absolutely no indication that people have moved to Minnesota to start new firms.
- There is a clear difference in importance assigned to different infrastructure features: access to customers dominates in importance; access to research and development facilities is considered unimportant by more than half.
- In general, there is a high level of satisfaction with infrastructure features. Satisfaction is generally the highest with regard to those features considered the most important. Satisfaction with infrastructure is relatively uniform across the thirteen regions of Minnesota.
- Taxes are a distinctive infrastructure feature. Considered of high importance, this factor is associated with the lowest level of satisfaction.
- Interest in potential government assistance varies substantially. Three in ten show high interest in management skill development; one in sixteen indicates an interest in small business incubator sites.
- Interest in government services tends to reflect the market orientation of the new firm. Local market oriented new firms show the greatest general interest in all services; exporting new firms show more interest in services related to exports and research and development; high-growth new firms reflect an interest in services related to growth.
- A typical new firm has a high interest in about four government services; seven in ten are interested in one or more.

There are several implications for economic development policies. Two major policy strategies are general attempts to improve the general business climate or programs designed to assist specific new firms.

Strategies for enhancing economic development through improving the "business climate" must deal with several major issues: What should be "improved"? Can it be "improved" enough to make a difference?

Perhaps the most important finding from this research was that those starting new firms report relatively high levels of satisfaction with most features of the infrastructure. And satisfaction is highest with those considered most important. It is not clear what might be "improved" except taxes, and it is not clear that taxes can be "improved" enough to make a major difference.

Conversely, substantial evidence suggests that programs designed to assist specific new firms may "make a difference." Most important, the analysis in Chapter 3 indicated that a small proportion of new firms (onefourth) are responsible for the majority of the jobs, sales, and exports. The analysis in this chapter suggests that a substantial proportion of new firms were interested in government assistance. Most important, different types of firms showed an interest in different combinations of assistance.

This suggests that a policy of identifying high-potential new firms and providing services tailored to their specific needs could have a substantial payoff. The resulting contributions to jobs, sales, and exports could be more easily measured. This would allow for the modification and correction of such programs to optimize the cost/benefit ratio.

.

and the second second

#### Chapter 6

#### EDUCATIONAL INSTITUTIONS AND NEW FIRMS

Educational programs and institutions in Minnesota occupy a central role in the development of new firms in two ways. They provide the education and training not only of those who initiate new firms but also of those who work in new firms, the individuals who are both the providers of jobs and a critical resource needed to develop goods and services. Both relationships will be reviewed in this chapter.

### EDUCATIONAL ATTAINMENT OF NEW FIRM PRINCIPALS

New firms are one major source of jobs, sales, and exports for Minnesota. The individuals who start new firms have completed different amounts and types of formal education.<sup>1</sup> The relationship between the new firms, characterized by the educational attainment of the principal, and the proportion of total contributions recorded in the sample, is presented in Figure 6.1. As the sample is representative of all new firms, the patterns shown there should reflect those of all Minnesota new firms initiated between 1979 and 1984.



# Figure 6.1. PRINCIPAL'S EDUCATION AND PERCENT OF TOTAL CONTRIBUTIONS

For this analysis, the education completed by the respondent will be considered as an indicator of that of the entire start-up team, and most new firms (60 percent) are started by two or more individuals. Data on educational attainment are available for almost the entire sample. The major results can be considered in terms of four educational attainment categories.

	Total C	ontributions (in	percents)
	Jobs	Sales	Exports
Up to high school degree	19	23	12
Post-high school experience	33	35	35
College (4-year) degree	34	30	36
Post-college (4-year] experience	15	12	17

About one-third of all contributions come from new firms initiated by those with four-year college degrees; about one-third from those with education beyond high school (vocational/technical training, associate degree, some college experience); roughly one-sixth are provided by those with high school experience (both with and without a degree); and approximately one-sixth by those with graduate experience (both with and without a graduate degree).

Those with all levels of educational attainment are making contributions to the state by starting new firms. Overall contributions are related to both the number of new firms and their average size, presented in Table 6.1.

	A11		1985	1985	Sales
	Respondents	1986	Sales	Exports	per Job
	(in percents)	Jobs	(\$1,000)	_(\$1,000)	(\$1,000)
Number	1,065	1,065	899	895	893
Some high school	4	7.0	542	16	136
High school degree	19	7.2	638	59	134
Vocational/technical	14	7.4	517	58	77
Associate degree	4	8.4	485	105	68
Some college	21	8.0	517	125	257
Four-year college deg	ree 24	12.5	655	162	77
Some graduate school	4	11.6	680	198	97
Master's degree	6	9.1	496	110	123
LLB, PhD, MD, etc.	3	7.1	227	88	61
Average all new firms		9.0	542	107	76

Table 6.1. PRINCIPAL'S EDUCATION AND AVERAGE NEW FIRM CONTRIBUTIONS

Note: Averages are those of total firms.

Except for slightly more employees for firms started by those with college degrees or some graduate education, there is little variation in the average number of jobs. Average 1985 sales do not vary much, except for new firms started by those who have completed advanced graduate programs; they also have the lowest sales per employee. Exports vary considerably, reflecting the different types of new firms started by those with differing educational attainment.

The industry sector of the new firms is presented by principal's educational attainment in Table 6.2. It is clear that educational attainment does affect the industry in which the new firm is initiated. Those with high school and area vocational technical institute (AVTI) experience tend to start new firms oriented toward local markets--in construction, retail, and consumer services. Those with associate degrees emphasize manufacturing, distributive services, business services and retail. Those with some college, but not a four-year degree, emphasize these four industries and construction. Completion of college and graduate experience leads to a major concentration similar to those with associate degrees: manufacturing, distributive services, business services, and retail. Most distinctive are those who have completed graduate programs, with a substantial emphasis in business services and health, educational, and social services.

The sales development patterns of new firms associated with the principal's educational attainment are presented in Table 6.3. High-growth new firms are associated with all levels of educational attainment, and, ironically, the lowest proportion of high-growth new firms is associated with completion of advanced degree programs. This may reflect the heavy emphasis on business and health, education, and social services among those with advanced graduate degrees.

The market orientation of the new firms--within Minnesota, regional, or national--is associated with the principal's educational attainment, as indicated in Table 6.4. Here there is a relatively clear relationship. With some exceptions, the more education completed by the principal, the greater the tendency to be oriented toward markets outside Minnesota. Particularly striking is the tendency for those with high school and vocational/technical training to be associated with new firms oriented toward local or regional markets.

	Agri	Lcult	ural							
	:	Min	ing							
	:	:	Con	struc	tion					
	:	:	:	Mar	ufact	uring	3			
	:	:	:	:	Dis	trib	utive	Serv	ices	
	:	:	. :	:	:	Bus	siness	s Ser	vices	
	:	:	:	:	:	:	Ret	cail		
	:	:	:	:	:	:	:	Co	nsumer	Services
	:	:	:	:	:	:	:	:	Hea	Lth,
	:	:	:	:	:	:	:	:	Edu	cation,
	:	:	:	:	:	:	:	:	Soc	ial
	•	:	:	:	:	:	:	:	Ser	vices
	:	:	:	:	:	:	:	:	:	
Some high school			22	8	15	9	28	20		102
High school degree	2		16	10	20	7	23	20	1	99
Vocational/technical	1		12	13	14	. 8	31	21	1	101
Associate degree	2		2	15	19	18	36	8	2	102
Some college	*	*	17	16	19	14	24	8	2	100
Four-year college degree	2		4	13	16	20	39	4	2	100
Some graduate school	2			19	22	27	19	6	5	100
Master's degree	3	1	1	13	11	36	26		9	100
LLB, PhD, MD, etc.	2			4	7	47	17		22	99 <sub>5</sub>
All new respondents	2	*	10	13	17	16	29	11	3	101
Some high school			8	2	3	2	4	7		
High school degree	29		30	15	23	9	16	36	5	
Vocational/technical	10		16	14	11	6	15	26	4	
Associate degree	4		1	4	4	4	5	3	3	-
Some college	6	50	36	26	23	17	17	15	14	
Four-year college degree	30		8	24	23	30	33	10	17	
Some graduate school	4			7	6	8	3	2	9	
Master's degree	13	50	1	6	4	14	6		24	
LLB, PhD, MD, etc.	4			1	1	9	2		26	
Column totals	100	100	100	99	98	99	101	99	102	

# Table 6.2. PRINCIPAL'S EDUCATIONAL ATTAINMENT AND NEW FIRM INDUSTRY (in percents)

Note: Asterisks indicate less than 0.6 percent. Both the weighted sample and rounding off of figures lead to a total not equal to 100 percent.

Start	Start			
	JLALL	Start	Start	Total
Low	Low	High	High	High
<u>Growth</u>	Growth	Growth	Growth	Growth
79	2	5	13	19
74	9	10	8	17
74	4	16	6	22
77		19	4	23
69	4	16	10	27
55	13	21	10	22
55	14	15	16	21
58	2	30	10	40
80	4	12	5	16
	Low Growth 79 74 74 77 69 55 55 55 58 80	Start         Start           Low         Low           Growth         Growth           79         2           74         9           74         4           77            69         4           55         13           55         14           58         2           80         4	Start         Start         Start         Start           Low         Low         High           Growth         Growth         Growth           79         2         5           74         9         10           74         4         16           77          19           69         4         16           55         13         21           55         14         15           58         2         30           80         4         12	Start         Growth         Growth <thgrowth< th=""> <thgrowth< th=""> <thgrowth< td=""></thgrowth<></thgrowth<></thgrowth<>

Table 6.3. PRINCIPAL'S EDUCATION AND NEW FIRM DEVELOPMENT (in percents)

Table 6.4. PRINCIPAL'S EDUCATION AND MARKET ORIENTATION (in percents)

Ţ	Within			Total
	State	Regional	National	Exports
Some high school	94	4	2	6
High school degree	92	4	4	8
Vocational/technical	89	8	3	11
Associate degree	80	9	11	20
Some college	82	7	10	17
Four-year college degree	78	8	14	22
Some graduate school	73	10	16	26
Master's degree	80	4	16	20
LLB, PhD, MD, etc.	71	7	23	30

#### EMPLOYEE EDUCATION

Locating suitable personnel is consistently mentioned as a major start-up problem. When asked specifically about difficulty in finding appropriately trained people, one in five of all new firms reports this to be a problem (about 28 percent of new firms with employees). There is, however, no relationship between the sales development patterns or export emphasis of new firms and the tendency to report major problems with finding employees with appropriate skills and training.

The types of training and employment that are difficult to find in the current labor pool, and which the new firms provide themselves, are presented in Table 6.5. A detailed listing of the specific occupational categories is provided in the appendix (see Tables A6.1 and A6.2) to this chapter.

### Table 6.5. EMPLOYEE SKILLS, TRAINING, OR EDUCATION DIFFICULT TO FIND, PROVIDED BY NEW FIRMS (in percents)

	Difficult	Provided by
	to Find	New Firms
Management	2	1 .
Engineers, scientists	2	1
Social scientists, lawyers, education or		
health professionals	1	*
Therapists, health and scientific technicians	8	3
Sales, clerical, service	14	13
Mechanical, construction, precision production	13	. 8
Production, transportation, labor	4	5
Other: general training or not specified	56	68

Note: The asterisk indicates less than 0.6 percent.

It is clear that the major types of problems are not indicated, or are so specific to the new firm and the way it operates that standard job titles do not provide an appropriate label. Most of those that can be labeled are those requiring an intermediate level of commitment. Only a small percentage are in managerial, administrative, or professional categories. In particular, the training provided within the new firm tends to focus on those jobs most easily acquired through on-the-job experiences.

A variety of post-high school educational systems serve the state of Minnesota. The respondents were asked which types of "retraining or additional experience" each type might provide for the firm's "managers, staff, or employees." The responses are summarized in Table 6.6. The details of the occupational categories specific to each type of educational institution are given in Tables A6.3 through A6.6 following this chapter.

The results suggest a systematic conception of research universities and four-year colleges as most suited for additional training related to management, scientific, professional jobs; two-year colleges as most suited to additional training for technicians and sales, clerical, and service jobs; and vocational/technical institutions most suited to craftspersons, production, or transportation-related positions. The high number of responses related to the vocational/technical institutions (equal to the total of responses for the other three alternatives) suggests a well-defined image of the types of educational value that may be provided by such institutions.

	Research University	Four-Year College	Two-Year College	Voc/Tech Schools.
Total responses	107	131	115	350
Management	3	6	2	*
Engineers, scientists Social scientists, lawyers, edu-	12	4	1	2
cation or health professionals Therapists, health and scientific	2		1	*
technicians	12	5	12	11
Sales, clerical, service Mechanical, construction,	18	16	24	19
precision production	1	5	6	24
Production, transportation, labor Other: general training or not	3			6
specified	50	64	54	38
	101	100	100	100

# Table 6.6. EDUCATIONAL INSTITUTIONS AND APPROPRIATE TRAINING OR RETRAINING (in percents)

Note: The asterisk indicates less than 0.6 percent.

### EDUCATIONAL SERVICE USE AND CONTRIBUTIONS

New firms were asked to indicate the extent of use of services that may be available from state, regional, or government agencies or programs. Unfortunately, only a small number completed this item, at the bottom of the fifteenth page of a long questionnaire. However, the use of two educational institutions was indicated by a small number of respondents. The character of the firms they represented is presented in Table 6.7.

Table 6.7. USE OF EDUCATIONAL PROGRAMS AND NEW FIRM CONTRIBUTIONS

-	Number 1985 1985				Percent	
	of	1986	Sales	Exports	Percent	High
	Firms	Jobs	(\$1,000)	(\$1,000)	Exporting	Growth.
۸ <i>۲ г</i> г г г г г г г г г г г г г г г г г		7 0	0.67	50	10	07
AVII	45	7.9	267	53	13	27
Community college	26	12.6	606	257	22	35
No use reported	619	8.8	549	100	18	27
Use not known	440	9.5	541	111	15	25

Although the sample of program users is small, and no information is available for 40 percent of the sample, the strong pattern related to use of community college programs deserves some comment. These are clearly new firms making major contributions; average jobs, sales, and exports are substantially greater than for the typical firm, and a greater percentage are both high growth and exporting. New firms reporting use of community college programs also report substantially greater contributions.

Unfortunately, the nature of the use of these programs is unknown. The community college programs could have been a source of training and education for the members of the start-up team or the employees; they could have been extensive formal programs or weekend workshops. But whatever the program, the community college system may be serving a group of new firms that are making substantial contributions to the economy. In contrast, the AVTI programs appear to serve a larger number of more traditional new firms. Both systems contribute to the success of new firms, but in different ways.

#### SUMMARY

It is clear that those starting new firms report varying degrees of educational attainment; it is equally clear that major contributions come from new firms associated with all levels of educational attainment. Ironically, new firms started by those completing advanced graduate programs make the most modest contributions, both individually and in the aggregate.

Locating suitably trained and experienced personnel is a major problem for new firms, and a continuing problem for growing new firms. New firm principals seem to consider that some of their needs can be met by existing educational institutions; it is clear that much training is provided by the firms themselves. Established occupational categories--which are constantly changing--do not provide adequate descriptions for the majority of the unmet training needs of the new firms. This probably reflects a lag in the revision of these categories as well as the idiosyncratic job descriptions developed within these new firms.

A tentative analysis of a small sample of new firms that reported use of the community college and AVTI systems was possible. Not many had used the community college system, but these firms were making substantial contributions to Minnesota. Almost twice as many new firms reported using the AVTI system; their contributions to Minnesota were typical of all new firms.

There are several implications from this analysis for the policies that may guide the educational systems in Minnesota. First, entrepreneurial training or programs oriented toward starting a new firm are appropriate for educational systems at all levels. Substantial contributions (jobs, sales, and exports) are made by firms initiated by those with all levels of educational attainment. However, these programs might be tailored to the type of new firm most typical of those completing different levels of education. New firms oriented toward local markets (particularly construction, retail, and consumer services) might be emphasized in high school and AVTI programs. This could be expanded to export-oriented industries at the four-year and research

### university institutions.<sup>2</sup>

Second, entrepreneurial training should include a discussion not only of local, state, and federal regulations and assistance but also of current sources of government support. Confusion and complications associated with government regulations are one of the most frequent sources of dissatisfaction among those starting new firms. Stressing techniques for obtaining government assistance in meeting legal and administrative criteria would be of particular value.

Third, educational institutions providing trained, skilled employees should probably approach individual firms and discuss their specific needs. While locating suitable employees is a major problem for new firms, each firm's needs seem unique. More than half the new firms could not describe their employment needs with established job titles. Substantial job training occurs within new firms, and major educational systems might make a significant contribution by developing specialized programs tailored to the job requirements of specific firms. Such efforts may be justified only for those firms with a high potential for job growth.

 $^{\rm 2}$  This is, to some extent, already the case.

			Percent	Percent
			of	of Valid
Category Label	Code	Count	Response	<u>s Cases</u>
Management, not specified	2	1	0.2	0.3
Word processing	6	1	0.2	0.2
Problem, not specified	10	7	0.2	2.3
Product knowledge	12	4	1.1	1.5
Sales, not specified	14	28	6.8	9.3
Telephone experience	15	1	0.0	0.1
Personal qualities	20	6	1.4	1.9
Motivated workers	21	42	10.1	13.9
Willingness to learn	22	9	2.1	2.9
Flexible workers	23	5	1.2	1.7
Good social skills	26	17	4.1	5.6
Other personal qualities	29	19	4.5	6.2
Cannot afford	30	8	1.9	2.6
No one wants to work	40	1	0.2	0.2
Experienced workers, not specified	50	40	9.7	13.3
Well-rounded experience	51	2	0.6	0.8
Reading, writing, math	52	9	2.1	2.9
Experience, other	59	6	1.5	2.1
General managers, top executives	121	4	1.0	1.4
Purchasing managers	124	1	0.3	0.5
Managers: medical, health	131	1	0.2	0.2
Accountants, auditors, financial specialists	141	1	0.2	0.2
Management occupations, not elsewhere classified	149	1	0.2	0.2
Architects	161	1	0.2	0 2
Engineers	162	1	0.2	0.2
Engineers	163	5	1 1	1 6
Computer scientists	171	1	0.0	0 1
Life scientists	185	1	0.0	0.1
Social scientists	191	1	0.0	0.1
Social and recreation	203	2	0.1	0.1
Lawyers	203	1	0.4	0.5
Pre-school and kindergarten	231	1	0.2	0.2
Teachers not alsowhere classified	231	1	0.2	0.3
Health practitioners not elsewhere classified	237	1	0.2	0.5
Registered purces	200	1	0.1	0.1
Pharmanists	301	6	1 4	1 9
Thoranists	303	2	1.4	1.9
	300	2	0.5	0.7
Viewal artista	225	1	0.8	1.1
VISUAL ALLISUS	220	1	0.3	0.4
Fitzers and reporters	223	1	0.1	0.1
Clinical lab tachniciana	260	1	0.2	0.2
Uselth technicians	202	1	0.2	0.2
nearth technicians, not elsewhere classified	202	L C	0.2	0.2
Engineering technicians	2/L 270	0	1.5	2.1
Dialling occupations	3/2	2	0.4	0.5
Hathematical technicians	384	1	0.2	0.2
Legal technicians	396	L	0.2	0.2

# Table A6.1. DIFFICULT-TO-FIND EMPLOYEE SKILLS: DETAILED OCCUPATIONS

J

	· · · · · · · · · · · · · · · · · · ·		Percent	Percent
			of	of Valid
<u>Category</u> Label	Code	Count	Response	s Cases
				0 00000
Programmers	397	7	1.7	2.4
Technicians, not elsewhere classified	399	1	0.0	0.1
Insurance, real estate, sales	412	2	0.6	0.8
Business service sales	415	5	1.3	1.8
Technical sales workers	423	1	0.2	0.2
Sales representatives	424	1	0.2	0.2
Salespersons, retail	434	2	0.4	0.5
Retail sales	435	3	0.6	0.9
Sales occupations, other	436	4	0.9	1.2
Computer equipment operators	461	4	1.1	1.5
Secretary and typist	462	7	1.7	23
Record clerks	469	1	0.0	0 1
Financial record processor	471	2	0.5	0 7
Recording, distributing clerks	475	1	0.1	0.1
Miscellaneous administrative support	479	1	0.2	0.1
Food services	521	17	4 2	5.8
Building services, not household	524	2	4.2 0 4	0.5
Personal/service occupations	525	6	14	1 9
Farmersworking owner	551	1	$0^{-1.4}$	0.2
Farm occupations, except management	561	1	0.2	0.2
Related agricultural occupations	562	4	0.2	1 2
Timber cutting	573	1	0.2	0.2
Vehicle mechanics	611	14	35	47
Electrical equipment repair	615	1	0.2	4.7 0 3
Miscellaneous mechanical and renair	617	3	0.2	1 0
Carpenters and related	642	2	0.7	0.7
Paint, paper and plaster	644	1	0.5	0.7
Plumbing pipe and steam	645	7	1 6	2.5
Other construction	646	,	03	0.5
Precision metal workers	681	4	0.5	1 2
Precision metal workers	682	2	0.5	0.5
Precision woodworkers	683	2	0.7	1 0
Precision printing	684	3	0.8	1 1
Precise textile workers	685	6	1 4	1 9
Precision worker assorted	686	1	0 4	0.5
Precision food production	687	3	07	1 0
Metal fab setup operators	733	1	0.7	0.2
Metal and plastic setup	734	1	0.2	0.2
Metal and plastic operators	751	2	0.2	0.5
Miscellaneous metal/plastic	752	1	0.4	0.5
Metal fab machine operators	752	1	0.2	0.2
Woodworking machine operators	753	1	0.1	0.2
Printing machine operators	705	1	0.2	0.5
Textile machine operators	765	1 3	0.0	1.0
Weldorg and goldorers	705	1	0.7	1.0
Motor vohiolo operators	//⊥ 001	2 T	0.5	0.4
Material moving equipment	120 120	2	0.4	0.5
Helporg mechanica	063 03T	د ۱	0./	1.0
Cannot code	605 000	10	U.Z	0.2
Other	000	19	4./ 2 E	0.J 3 /
- OTICT	900	TO	2.5	5.4

# Table A6.2.TRAINING OR EDUCATION PROVIDED BY NEW FIRMS:<br/>DETAILED OCCUPATIONS

			Percent	Percent
			of	of Valid
Category Label	Code	Count	Response	<u>s Cases</u>
Company management	2	3	0.3	0.4
Supervision, not specified	4	4	0.4	0.6
Word processing	7	1	0.1	0.1
On-the-job training	10	128	12.8	18.0
Informal training	11	82	8.3	11.6
Product/service training	12	68	6.8	9.6
Sales experience	14	118	11.8	16.6
Telephone experience	15	6	0.6	0.9
Time management	17	4	0.4	0.5
New technology	18	2	0.2	0.3
Other training	19	14	1.4	1.9
School, voc/tech, etc.	20	35	3.5	4.9
Company courses	21	23	2.3	3.2
Orientation	22	4	0.4	0.5
Apprenticeships	23	27	2.7	3.8
Internships	24	2	0.2	0.3
Seminars, inservice	25	29	2.9	4.1
Tapes, manuals	26	22	2.2	3.1
Training, not elsewhere classified	29	1	0.0	0.0
Customer service communications	30	23	2.3	3.3
Purchasing managers	124	1	0.1	0.1
Managers, medical and health	131	1	0.1	0.1
Construction managers	133	1	0.1	0.1
Managers: service orgnizations	135	1	0.1	0.1
Accountants, auditors, financial specialists	141	3	0.3	0.4
Management analysts	142	1	0.1	0.1
Inspection compliance officer	147	1	0.1	0.1
Management occupations, not elsewhere classified	149	3	0.3	0.4
Architects	161	· 1	0.1	0.1
Engineers	162	1	0.1	0.2
Engineers	163	4	0.4	0.5
Life scientists	185	1	0.1	0.1
Social scientists	191	. 1	0.0	0.0
Lawyers	211	1	0.1	0.1
Teachers, not elsewhere classified	239	1	0.1	0.1
Health practitioner, not elsewhere classified	289	1	0.0	0.0
Therapists	303	2	0.2	0.3
Physicians' assistants	304	1	0.2	0.2
Designers	322	3	0.3	0.4
Visual artists	325	1	0.0	0.0
Photographers	326	3	0.3	0.5
Editors and reporters	331	2	0.2	0.3
Clinical lab technicians	362	1	0.1	0.1
Radiologic technicians	365	1	0.1	0.1
Health technicians, not elsewhere classified	369	1	0.1	0.2
Engineering technicians	371	4	0.4	0.5

			Percent	Percent
			of	of Valid
<u>Category Label</u>	Code	Count	Responses	s Cases
Drafting occupations	372	2	0.2	0.2
Chemical and nuclear technicians	383	3	0.3	0.4
Legal technicians	396	1	0.0	0.0
Programmers	397	7	0.7	1.0
Supervisor: retail sales	403	3	0.3	0.4
Insurance, real estate, sales	412	5	0.5	0.6
Business service sales	415	2	0.2	0.3
Retail sales	435	3	0.3	0.5
Sales occupations: other	436	9	0.9	1.3
Computer equipment operators	461	17	1.7	0.6
Secretaries and typists	462	-/	0 4	0.6
General office occupations	- 463	3	03	0.4
Information clerks	465	2	0.5	0.4
Record clerks	404	1	0.2	0.5
Financial record processing	405	28	2 9	3.0
Mail and message distribution	471	20	2.0	J. <del>J</del>
Recording distributing clerks	4/4	7	0.1	0.1
Adjusters investigators collectors	475	1	0.7	0.9
Food service	4/0 501	2/	0.0	0.1
Health service occupations	500	54	3.5	4.9
Building service occupations	523	2	0.2	0.3
Personal service, not nodsenoid	524	1	0.3	0.4
Personal service occupacions	525	L (	0.0	0.0
Farm occupations except menocoment	520	4	0.4	0.5
Related agricultural accurations	201	Z	0.2	0.3
Vehicle mechanice	562	5	0.5	0.7
Industrial machine warsin	611	4	0.4	0.5
Machineru machine repair	613	4	0.4	0.5
Flootmonie and manual	614	4	0.4	0.6
Migoallana and it is in the second se	615	3	0.3	0.5
Briel and repair	617	9	0.7	0.9
Come and stone masons	641	3	0.3	0.4
Point and related	642	4	0.4	0.6
Plant, paper and plaster	644	3	0.3	0.4
on the steam	645	8	0.8	1.2
Uner construction	646	6	0.6	0.9
recision metal workers	681	5	0.5	0.7
rrecision metal work	682	2	0.2	0.2
rrecision woodworkers	683	2	0.2	0.3
Precision printing	684	8	0.8	1.1
Precise textile work	685	3	0.3	0.5
Precision workers; assorted	686	5	0.5	0.7
Precision food production	687	6	0.6	0.8
Metal and plastic setup	731	2	0.2	0.3
Metal and plastic setup	734	1	0.1	0.2
woodworking setup	743	2	0.2	0.3
Assorted materials; setup	746	1	0.1	0.1
Metal and plastic operators	751	2	0.2	0.3
Metal fab machine operators	753	1	0.1	0.2
Metal and plastic processors	754	2	0.2	0.2

			Percent	Percent
			of	of Valid
Category Label	Code	Count	Responses	<u>s Cases</u>
			<u> </u>	0.1
Woodworking machine operators	763	T	0.1	0.1
Printing machine operators	764	1	0.1	0.1
Textile machine operators	765	3	0.3	0.4
Welders and solderers	771	3	0.3	0.5
Assemblers	772	- 3	0.3	0.4
Supervisor: motor equipment operators	811	1	0.1	0.1
Motor vehicle operators	821	12	1.2	1.8
Pilots and navigators	825	· 1	0.1	0.1
Material moving equipment	831	7	0.7	0.9
Helpers; machine operators	861	1	0.1	0.1
Freight movers; hand	872	5	0.5	0.7
Garage and service station	873	3	0.3	0.4
Miscellaneous manual occupations	876	1	0.0	0.0
Cannot code	888	41	4.1	5.8
Other	900	21	2.1	3.0
Hire only experienced	910	25	2.5	3.6

			Percent	Percent
17.1		-	of	of Valid
Value Label	Code	Count	Cases	Cases
Training needed, not specified	001	14	1.2	12.6
Company management	002	19	1.7	17.8
Company started class	003	1	1	6
Supervision, not specified	004	1	.1	. 7
Reading, writing, math	005	1	.1	.9
Sales experience	012	2	.2	1.6
Telephone experience	014	1	.1	.7
Seminars, inservice	023	. 7	.6	6.1
Financial managers	134	1	.1	.6
Managers; marketing, advertizing and PR	135	2	.1	1.4
Managers: mining. etc.	133	1	.1	.7
Architects	161	1	.1	. 8
Engineers	162	9	. 8	7.8
Life scientists	185	3	.3	3.0
Social scientists	191	1	.1	.9
Lawyers	211	1	.1	.6
Health practitioners, not elsewhere classified	289	1	.1	.9
Physicians' assistants	304	1	.1	.7
Designers	322	3	.3	2.7
Visual artists	325	3	.3	2.7
Clinical lab technicians	362	1	.1	.7
Dental hygienists	363	1	.1	.7
Programmers	397	4	.4	4.0
Insurance, real estate, sales	412	1	.1	1.3
Business service sales	415	11	1.0	10.2
Sales representatives	424	1	.1	.7
Computer equipment operators	461	1	.1	1.4
Financial record processing	471	5	.4	4.3
Farm managers	501	1	.1	.7
Textile machinery operators	765	3	.3	2.7
Cannot code	888	7	.6	6.8
Other	900	2	.2	2.2
No training	000	819	73.2	Missing
No employees	777	151	13.5	Missing
Refused to answer	999	42	3.7	Missing

# Table A6.3. RETRAINING OR ADDITIONAL EDUCATION APPROPRIATE FOR RESEARCH UNIVERSITY: DETAILED OCCUPATIONS

I

			Percent	Percent
Value Label	Codo	Count	OI Cagag	or valid
	Coue		Jases	Uases_
Training needed, not specified	001	13	1.2	9.9
Company management	002	37	3.3	28.0
Company started class	003	0	.0	.2
Supervision not specified	004	1	.1	.7
Word processing	011	1	.1	.5
Product/service training	012	1	.1	.6
Sales experience	014	14	1.2	10.4
Company courses	021	1	.1	.6
Apprenticeships	023	1	.1	.6
Seminars, inservice	025	- 7	. 6	5.0
General managers and top executives	130	1	.1	.6
Managers: marketing, advertising, and PR	131	1	.1	1.1
Managers: medical and health	133	1	.1	1.0
Management analysts	134	1	.1	6
Management occupations, not elsewhere classified	139	3	3	2.4
Engineers	162	1	.5	1 1
Engineers	163	2	.1	1.6
Physical scientists	175	1	.1	- 6
Life scientists	185	1	.1	.6
Registered nurses	290	1	.1	.6
Authors	321	1	.1	· · 6
Designers	322	2	.1	1.1
Editors and reporters	331	1	.1	.6
Health technicians, not elsewhere classified	362	1	.1	.6
Programmers	397	3	.2	2.0
Insurance, real estate, sales	412	2	.2	1.6
Business service sales	415	10	.9	7.5
Computer equipment operators	461	1	.1	.6
Financial record processing	471	7	.6	5.3
Personal services	525	1	.1	.5
Farmersworking owners	551	3	.3	2.2
Farm managers	552	2	.1	1.2
Farm occupations, except management	561	1	.1	.6
Other construction	646	1	.1	.5
Precision worker: assorted	686	1	.1	. 6
Cannot code	888	- 7	.6	5.0
Other	900	. 3	.3	2.3
No training	000	795	71.0	Missing
No employees	777	151	13.5	Missing
Refused to answer	999	42	3.7	Missing
		. –	- • •	0

# Table A6.4. RETRAINING OR ADDITIONAL EDUCATION APPROPRIATE FOR FOUR-YEAR COLLEGES: DETAILED OCCUPATIONS

0

			Percent	Percent
			of	of Valid
Value Label	Code	Count	Cases	Cases
Training needed, not specified	001	19	1.7	16.8
Company management	002	19	1.7	16.1
Company started class	003	1	.1	.8
Supervision, not specified	004	2	.1	1.3
Reading, writing, math	005	2	.2	1.5
On-the-job	010	1	.1	.7
Product/service training	012	3	.2	2.4
Sales experience	014	2	.2	1.5
Seminars, inservice	025	6	.5	5.0
Customer service communications	030	3	.3	2.5
General managers and top executives	031	1	.1	1.3
Accountants, auditors, financial specialists	041	1	.1	.8
Engineers	162	1	.1	.7
Life scientists	185	1	.1	.7
Pre-school and kindergarten	231	1	.1	.7
Therapists	303	1	.1	. 6
Physicians assistants	304	1	.1	.7
Editors and reporters	331	1	.1	.7
Public relations specialists	333	1	.1	1 2
Licensed practical nurses	341	1	.1	. 6
Health technicians	369	1	.1	.0
Engineering technicians	371	2	2	1 5
Programmers	397	5		44
Technicians, not elsewhere classified	399	1	.5	
Business service sales	415	4	. <u>-</u> 4	37
Sales representatives	423	1	1	6
Sales occupations other	435	1	.1	.0
Computer equipment operator	461	3	3	25
Secretary and typists	462	1	.5	9
Information clerks	464	1	.1	.,
Financial record processing	404	12	1 0	10 2
Miscellaneous administrative support	479	1	1	9
Food service	521	1		5
Personal service occupations	525	3	.3	25
Personal services	529	1	.5	5
Farm manager	560	2	1	13
Related agricultural occupations	562	. 1	.1	±.5 6
Brick and stone masons	461	1	1	.0
Carpenters and related	462	3	.1	25
Precision metal workers	681	1		2.5
Cannot code	888	- 6	5	., 4 9
Other	900	1	.5	4.) 7
No training	000	812	72 5	., Missing
No employees	777	151	13 5	Missing
Refused to answer	999	42	3 7	Missing
			J. /	TTOOTING

# Table A6.5. RETRAINING OR ADDITIONAL EDUCATION APPROPRIATE FOR COMMUNITY COLLEGES: DETAILED OCCUPATIONS

F7 F7 F7

-

-87-

Table A6.6.	RETRAINING OR ADDITIONAL	EDUCATION APPROPRIATE FOR VOCATIONAL/	
	TECHNICAL INSTITUTIONS:	DETAILED OCCUPATIONS	

			Percent	Percent
			of	of Valid
Value Label	Code	Count	Cases	Cases
Training needed, not specified	001	66	5.9	18.8
Company management	002	7	.7	2.1
Company started class	003	0	.0	.1
Supervision, not specified	004	1	.1	.2
Reading, writing, math	005	0	.0	.1
Word processing	006	1	.1	.4
On-the-job, not specified	010	1 ·	.1	.4
Informal training	011	1	.1	.2
Travel experience	012	1	.1	.2
Sales experience	014	12	1.1	3.5
Seminars, inservice	023	4	.4	1.2
Training, not elsewhere classified	029	3	.2	.8
Customer service communication	030	1	.1	.3
Reading, writing	039	1	.1	.2
Managers, service organizations	132	1	.1	.3
Engineers	162	4	.4	1.2
Surveyors and map scientists	172	1	.1	.2
Life scientists	185	2	.2	. 6
Pre-school and kindergarten	231	1	.1	.2
Physicians' assistants	281	1	.1	.2
Designers	322	2	2	.=
Visual artists	325	2	.1	.4
Photographers	326	3		8
Public relations specialists	327	1		
Health technicians not elsewhere classified	369	1	.1	2
Engineering technicians	371	10	9	3 0
Drafting occupations	372	6	.,	1 8
Chemical and nuclear technicians	372	4	. U . A	1 0
Legal technicians	396	1	.5	2.0
Programmers	307	8	.1	23
Supervision: insurance etc	410	1	./	2.5
Insurance, real estate sales	410	0	.1	.4
Business service sales	412	1	.0	.1
Salas representatives	415	1	. 1	.2
Potoil galog	424	· 2	. 1	.2
Relati Sales	435	1	. 2	. /
Sales occupacions, other	430	10	.1	. 2
Computer equipment operators	461	10	.9	3.0
Constal office constitute	402	2	. /	2.1
Jufermenties alerely	403	0	. 5	. 0
Discusiel and an	404	11	.1	.4
Financial record processor	4/1	11	1.0	3.2
Mall and message distribution	4/4	1	. L	.2
Recording, distribution clerks	4/5	L	.1	.2
Aajusters, investigators, collectors	4/8	0	.0	.1
Miscellaneous administrative support	479	1	.1	.3

			Percent	Percent of Valid
<u>Value Label</u>	Code	Count	Cases	<u>Cases</u>
Food service	501	10	0	2 0
Health service occupations	523	10	.9	2.9
Building services not household	525	2	.1	.4
Personal service occupations	524	12	1.0	• L • • •
Personal services	525	1	1.0	3.3
Farmersworking owners	551	2	.1	. 2
Farm managers	557	2	. 5	.8
Farm occupations, except manangement	561	2	.1	.4
Related agricultural occupations	562	1	.0	.1
Vehicle mechanics	502	17	.1	.4
Industrial machine renair	613	1/	1.5	4.9
Electrical equipment repair	615	15	.4	1.2
Miscellaneous mechanical and repair	615	12	1.4	4.4
Carpenters and related	64.2	4	. 3	1.1
Electricians and nower trans installors	642	4	.4	1.3
Paint paper and plaster	643	3	.3	.8
Plumbers nine and steam	644	3	.3	.8
Other construction	645	1 /	• 1	.2
Construction trades	646	4	.4	1.2
Precision metal workers	649	L	.1	.2
Precision metal work	681	8	./	2.4
Precision woodworking	682	L	.1	.3
Precision printing	683	2	.2	.5
Precision textile work	684	3	.3	1.0
Precision work: accorted	685	5	.5	1.6
Precise food production	000	1	• 1	.4
Metal and plastic setup	00/	1	.1	.2
Metal and plastic setup	731	1	.1	.4
Woodworking setup	734	1	.0	.1
Metal and plastic operators	743	L	.1	.2
Miscellaneous motol and plastic	744	3	.2	.8
Metal fab machine energiators	746	1	.1	.2
Printing machine operators	751	1	.1	.2
Textile machine operators	764	1	.1	.2
Operators: assorted materials	765	3	.2	.8
Welders and colderers	/00	1	.1	.2
Motor vohiala aparatara	//1	3	.3	1.0
Matorial moving aggingent	821	2	.2	.7
Supervisores bendlement	831	3	.3	.8
Cappet and	840	0	.0	.1
Other	888	24	2.2	7.0
No training	900	7	.6	2.0
No employee	000	5/5	51.4	Missing
Refuged to emprove	///	151	13.5	Missing
Actused to answer	999	42	3.7	Missing



#### Chapter 7

#### SUMMARY AND IMPLICATIONS

Policymakers concerned with new firms and economic development are confronted with several decisions regarding appropriate strategies. The first decision is, Should public resources be devoted to enhancing new firms?

If the answer is yes, three strategies are available:

- I. Improve the regional and institutional features important to new firms. An improved "business climate" will benefit all new firms and may, in turn, facilitate economic development.
- II. Provide assistance to specific high-potential new firms, those that may make the greatest contributions (direct and indirect) to economic development and regional well-being.
- III. Implement a combination of these two strategies.

This study of the new firms established in Minnesota between 1979 and 1984 and surviving into 1986 provides information relevant to these issues.

#### MAJOR FINDINGS

Analysis of the contributions of new firms suggested the following:<sup>1</sup>

- New Minnesota firms established between 1979 and 1984 provided 6-14 percent of the 1986 jobs and 5-12 percent of the 1985 personal income. Their contributions were equal to 42-99 percent of the net growth in jobs between 1978 and 1986 and 12-29 percent of the net growth in personal income (p. 1).
- 2. New firms in all major industries were important sources of jobs and sales (p. 6).
- 3. New firms in manufacturing, distributive services (wholesale), and business services were important as sources of out-of-state exports (p. 6).
- 4. Most new firms' contributions occur in the Minneapolis-St. Paul region, but new firm jobs are a larger proportion of existing jobs in the Greater Minnesota regions (pp. 7-8).

<sup>1</sup> Report page numbers are given in parentheses.

From five to ten thousand new firms are founded in Minnesota each year (p. 16). Examination of the patterns of new firm birth rates across the thirteen regions of Minnesota indicated that:

- 5. Birth rates of new firms in export-potential industries have a modest relationship to the birth rates of new firms in local-market industries (p. 19).
- 6. The founding of new firms in export-potential industries--manufacturing, distributive services, business services--tends to be highest in regions with higher proportions of college graduates, higher per capita income, and more midcareer adults (p. 20).
- 7. The emergence of new firms in local-market industries--agriculture, mining, construction, retail, consumer services--is not significantly related to the major region characteristics. It may be more uniform, reflecting a pervasive turnover of firms in these industries (p. 21).

Analysis of those factors related to variation in contributions (particularly jobs and sales) provided by all new firms indicated that:

- 8. The major factors that accounted for 55 percent of the variation in the 1986 employment of new firms were the absence of managers and professionals in the start-up year, the number of start-up problems, total informal financial resources, total employees in the first year, proportion of unskilled employees, current management focus on marketing, importance of local labor markets, sales emphasis on out-of-state exports, average sales growth in first years, and average initial working capital loan (p. 29).
- 9. The major factors accounting for 42 percent of the variation in 1985 sales were sales growth in the early years, number of start-up problems with employment, total informal financial resources, percentage of start-up team working forty to sixty hours per week, percentage of start-up financial problems solved, percentage of males on start-up team, percentage of costs for materials, importance of local labor markets, number of start-up problems with financing, and sales in the first year (p. 29).

New firms were classified in several different ways to determine their relative level of contributions:

- 10. New firms were classified into four types based on developmental patterns. Those with high annual sales growth (over \$100,000 per year) were 24 percent of the new firms and provided 60 percent of the jobs and 80 percent of both sales and exports. The low-growth firms were the dominant type in every industry sector. High-growth firms are more prevalent in urbanized regions (pp. 31-34).
- 11. New firms were classified in terms of their export orientation: 80 percent emphasized intrastate sales, 12 percent to adjoining states, and 8 percent national. Four of five export-oriented firms were from manufacturing, distributive services, or business services. They were 20 percent of all firms and provided 31 percent of the jobs, 45 percent of

the sales, and 96 percent of all exports. They tended to be located in major urban areas (pp. 36-38).

12. High-growth firms with an export emphasis were 13 percent of the sample and accounted for 24 percent of the jobs, 38 percent of the sales, and 81 percent of the exports provided by the sample of new firms (pp. 38-39).

Analysis of the start-up problems reported by those initiating new firms indicated that:

- 13. Start-up problems fell into several major categories: financial support, marketing strategy, cash management, government relations, development and implementation of planning, acquiring personnel and maintaining cohesion, site factors, access (to customers, clients, suppliers), and features of the infrastructure (p. 49).
- 14. There is a consistent positive relationship between the number and severity of start-up problems and contributions (jobs, sales, and exports); the greater the contributions, the more start-up problems are reported (p. 52).
- 15. The more start-up problems reported as solved, the greater the current contributions of the new firm (p. 53).

Consideration of the reaction of those responsible for new firms to their local context and potential government programs and services indicated that:

- 16. Virtually all new Minnesota firms are started by established residents. They suggest knowledge of the area and local industry and well-developed family and business relations are of major importance. <u>No one</u> in the sample moved to Minnesota to start a new business (p. 55).
- 17. The most important features of the local context are access to customers, least important was land availability and access to research and development facilities (p. 57).
- 18. Satisfaction with most infrastructure features tends to be high (higher for the more important features). The only major exception is taxes, considered both important and universally unsatisfactory (pp.58-59).
- 19. The typical firm indicated high interest in four of thirty-one services that might be provided by state, regional, or local governments. Each new firm had a distinctive set in which they were interested, and all potential services received high interest from a substantial number of new firms. Two types of services--financial support and management/ entrepreneurial training--received the largest percentage of high interest (p. 66-67).

Special attention was given to the relationship between education and the initiation of new firms. The major patterns found in the analysis indicated that:

20. About one-third of all new firm contributions come from firms initiated by those with education beyond high school (vocational/technical training,

associate degrees, some college experience); about one-third by those with college degrees; one-sixth by those with high school experience (with and without degrees); and about one-sixth with graduate experience (with and without degrees) (p. 71).

- 21. Those responsible for new firms consider a variety of educational institutions as suitable sources of retraining or additional experience for their employees; most frequently mentioned are the AVTI schools as a source of training for craft, technical, and service skills (p. 77).
- 22. Those new firms reporting use of community colleges also report somewhat greater jobs, sales, and exports and are more likely to report high growth and an export emphasis (p. 77).

#### IMPLICATIONS

There seems to be little doubt that new firms are <u>one</u> major source of jobs, sales, and exports for Minnesota. Attention to the problems and needs of new firms could provide benefits to the state.

General programs to assist all new firms may have a positive impact, but given the large number of new firms initiated annually (up to 10,000) and the multitude of factors affecting their contributions, the impact or efficiency of such programs may be difficult to determine.

A small proportion of new firms (less than one-third) account for the majority of sales and jobs. An even smaller proportion (about one-sixth) account for almost all out-of-state exports. This suggests that targeting programs to these 1,500-3,000 high-potential new firms may be a cost-effective strategy for improving new firm contributions and, in turn, Minnesota's economic development.

Almost two-thirds of new firms have a modest development profile. Starting small and staying small, they are quite numerous: as many as 7,000 a year in Minnesota. A politically potent force, they may react negatively to programs geared toward high-growth new firms.

Services and programs that might be provided by state, regional, and local governments would be likely to receive a favorable reception--if the right set of services were provided for the distinctive needs of each firm. High-growth, export-oriented new firms seem to be particularly receptive to these forms of assistance.

#### UNRESOLVED ISSUES

In developing a full understanding of the initiation, development, and contributions of new firms, there are several unresolved issues:

I. The ability to devise models of the initiation process in more detail, specifically the characteristics of regions that lead to variation in new firm birth rates must be developed. Such research is possible using existing sources of data on new firm births by county, region, state, etc. Unfortunately, no data on new firm births--even for the entire United States--are available for past years.

- II. Techniques for identifying firms with high potential for growth and/or exports must be refined. Procedures that would be useful for public programs designed to locate and assist new firms with a high potential for contributions. (The success at locating them for this research project is encouraging.)
- III. This analysis has been based on data collected at one point in time, and development patterns were inferred from historical reports from the new firms. The accuracy of the predictions and the future survival of new firms can best be explored with a longitudinal study design. Perhaps involving follow-up surveys at two-year intervals.
  - IV. Two special categories of new firms may bear closer examination, with detailed case studies or special additional surveys.
    - A number of high-growth and export-oriented new firms are emerging in nonmetropolitan regions. Although small in number, their contributions may be quite significant. Careful attention to these unique firms may lead to the design of strategies (for government policy or entrepreneurs) that may increase their frequency and success.
    - Firms emphasizing the use or development of new or high technology are included in the sample. Small in number, they are considered important as a source of new industries and, it is suspected, out of state exports. Careful attention to these new firms may highlight the extent of their contributions, special start-up problems, and any distinctive infrastructure (contextual) needs.

-95-

١g

•
### Appendix A

### DEFINITIONS

### ESTABLISHMENTS VS. ENTERPRISES (FIRMS)

There is great diversity in business firms, both in type and location throughout Minnesota. The description of different types of firms is complicated by a conceptual problem. In this report, a distinction has been made between:

ESTABLISHMENT: A single physical location where business is conducted or where services or industrial operations are performed (e.g., a drug store, manufacturing plant, or warehouse).

ENTERPRISE (or FIRM): A legal entity consisting of one or more establishments.

Differences between the two can be substantial. Four major manufacturing enterprises (firms) control hundreds of establishments in the production of automobiles. Thousands of individual establishments, each a separate legal entity, make up the auto repair industry.

Establishments are the major source of data on businesses. This includes the count of business entities and employment, provided as an annual census by the Bureau of the Census and published in *County Business Patterns*, as well as the listings supplied by Dun and Bradstreet in Dun's Marketing Identifier file.

For this reason, the analyses in these reports are based on establishments. This is not critical when dealing with new firms, most of which <sup>Consist</sup> of a single establishment. But it can be misleading when considering <sup>existing</sup> business entities; the average enterprise (firm) consists of two <sup>establishments</sup>.

### INDUSTRY CLASSIFICATION

The most widely used procedure for classifying business firms on the basis of economic activity is the Standard Industrial Classification (SIC) developed by the Bureau of Labor Statistics.<sup>1</sup> Unfortunately, the categories emphasized by the SIC are losing their value for describing the structure of the economy.<sup>2</sup> For example, the largest number of specific codes is found in <sup>manufacturing</sup>, a source of no more than 20 percent of all jobs. Even more

<sup>1</sup> The basic document is the Standard Industrial Classification Manual: 1972, a
slightly revised version of the 1965 manual. A modest set of changes was
2 published in a 1977 supplement.

<sup>&</sup>lt;sup>4</sup> A major revision of the SIC codes is now being implemented in all federal data-collection systems; *Federal Register*, October 1, 1986, pp. 35170-188.

problematic is that quite diverse and rapidly growing sectors of the economy are lumped under one heading. "Services" includes not only business and personal services but also health, education, and social services.

The classifications used in the analysis for this project are based on aggregations of SIC codes presented in Table A.1. Most categories--such as construction, manufacturing, and retail--are adopted without change from the most widely used descriptive schemes. Three are modified for this analysis. One is a minor consolidation. Industry sectors with a focus on distribution-of goods, people, information, energy--are classified under distributive services, combining wholesale and public utilities and transportation; wholesale establishments constitute 80 percent of this category.

The distinction between producer (or business) and consumer services is greater. The former are those industry sectors in which the major customer or client is another business--a legal person; the latter are those in which the major client is a human being--a natural person. The assignment of some specific industry sectors to producer or consumer services is a "close call." Membership organizations are placed in business services because they include professional societies and unions--but they also include churches. Hotels and lodging are placed in consumer services, but a major portion of their revenue is derived from business travelers. This also true for restaurants, which are included in the retail sector.

This is not the perfect, ultimate classification scheme, but it does provide more useful descriptions of economic activity than the categories now widely used.

# Table A.1. CLASSIFICATION OF FIRM ECONOMIC EMPHASIS

V

Agriculture	0100-0999	Agriculture production and service, forestry, fishing
Mining	1000-1499	Mining, mining services
Construction	1500-1799	Construction of all kinds
Manufacturing	2000-3999	Manufacturing of all kinds
Distributive services	4000-4999 5000-5199	Transportation, other public utilities Wholesale of all kinds
Business services	6000-6799 7300-7399 8100-8199 8600-8699 8900-8999	Fire, insurance, and real estate Business services Legal services Membership organizations Miscellaneous professional services
Retail	5200-5999	Retail of all kinds
Consumer services	7000-7099 7200-7299 7500-7599 7600-7699 7800-7899 7900-7999 8400-8499	Hotel and lodging Personal services Auto repair, service, garages Miscellaneous repair services Motion pictures Amusement and recreation Museums, gardens, etc.
Medical, educational social services	8000-8099 8200-8299 8300-8399	Health and medical services Educational services Social services

#### Appendix B

### ESTIMATING NEW FIRMS

The basic source of all estimates of new firms is the data obtained from Dun's Marketing Identifier (DMI) file provided by Dun and Bradstreet's Marketing Services. The design of the sample is presented in Table B.1. For each of six "year starts" (as indicated in the DMI file), 750 firms were selected at random from the DMI file. The sample was stratified, within each year, on two variables. For each year, 167 were to be sampled from the city of St. Paul and 583 from the rest of the state. Second, 75 percent were to be sampled from five industry sectors (either rare as a source of new firms or distinctive in their out-of-state exports) and 25 percent from three that dominate the total population of new firms. These sectors and the corresponding two-digit SIC codes are presented in Tables B.1 (Minnesota excluding St. Paul) and B.2 (St. Paul only).

Dun's Marketing Services provided a count of the total establishments listed for each "year start" for St. Paul and the rest of Minnesota, as well as for the two sample populations--twenty-four numbers in all. This is referred to as the DMI population. It is used to compute the sampling ratio: the ratio of the DMI population to the number in the sample purchased for the 1986 Minnesota new firm survey.

Table B.1 indicates the sample and sampling ratios by industry sector and DMI "year start" for Minnesota excluding St. Paul; the DMI file is estimated to have 22,692 listings that were the basis for the sample provided by Dun's Marketing Services. The same procedure, with different sampling ratios, is presented for St. Paul in Table B.2; it is estimated that 1,601 listings were in the DMI file when the sample was drawn. The total 4,497 establishment sample<sup>1</sup> is considered representative of the 24,293 establishments in the DMI Population as of April 1986.

Estimates of the total number of new firms involve several changes in this basis data; they are represented in Table B.3. The first is an enhancement for the new firms not identified by the Dun and Bradstreet field staff. Estimates of the proportion of new firms missed, by age of firm and industry sector, were used to adjust for the missed new firms. These figures were derived by estimating the annual "capture" rate of new firms by the Dun and Bradstreet field staff from the rate at which new firms of different ages are

<sup>&</sup>lt;sup>1</sup> This was to have been 4,500, but after dividing the sample into twenty-four cells, there was some loss due to fractional numbers. Further, the population in St. Paul for some cells was less than the sample requirement. In these cells, the entire population was included in the "sample." This is reflected in the St. Paul sampling ratios that are equal to 1.00.

added to the DMI file.<sup>2</sup> It does not, unfortunately, reflect a comparison of DMI data with a complete census of business entities--such a census does not exist. These estimates appear as the "Birch correction" in each of the six years of corrections in Table B.3.

The second problem, identified in surveys of new firms, is that approximately one-half do not qualify as new firms: about 20 percent are not new firms (representing changes in ownership); about 6 percent are failed, inactive, or not businesses; about 12 percent cannot be contacted by phone after *repeated* attempts (more than six); about 2 percent are not businesses, duplicates, mergers; and for 6 percent no successful phone screen has been possible (refusals) (details are provided in the Methodological Appendix). Overall, it is conservative to assume that only one-half of the DMI listings are viable, autonomous, and new firms. Hence, all estimates are adjusted downward to account for this finding.

The result is the range of estimates of the number of new business entities:

	St. Paul	Rest of Minnesota	Minnesota 6-Year Total	Annual Estimates
DMI sample (April 1986)	999	3,498	4,497	750
Estimate of DMI population as of April 1986	1,601	22,692	24,293	4,048
After "Birch correction" for missed firms			56,999	9,500
After correction for DMI listings not currently viable, autonomous, and new			28,499	4,750

The most conservative estimate of new firm foundings is 4,048 per year, and the highest estimate is 9,500 per year; 4,750 per year is a reasonable modal estimate for the entire state. Systematic studies of founding rates, based on state unemployment insurance records, indicate that annual new establishment foundings of 100 per 1,000 are not unreasonable (see Chapter 2). This would be 10,000 per year for Minnesota as of 1987; the estimates above are thus conservative.

Estimates of the number of new firms in each of the thirteen regions of Minnesota follow the same procedure, with one change. Rather than compute the

<sup>&</sup>lt;sup>2</sup> David L. Birch and Susan MacCracken, "The Small Business Share of Job Creation: Lessons Learned from the Use of a Longitudinal File" (Cambridge, Mass.: MIT Program on Neighborhood and Regional Change, mimeo., March 1983).

Birch correction, which enhances the count from the DMI file for each of six "year start dates," the average of the third- and fourth- year correction factor is used for these estimates. The small proportion of new firms in the nonmetropolitan regions of Minnesota leads to very small counts when both DMI "year start" and industry sector are being considered. The state totals resulting from the two procedures suggest a rather high level of correspondence.

	Statewide Estimate by Industry and Year Start	Regional Estimate by Industry
Minimum state total	12,140	12,059
Maximum state total	28,499	26,371

The estimates of new firm foundings in Table B.4 are used for the analysis in Chapter 1.

For the analysis in Chapter 2, estimates of new firm births for each region are based on the uncorrected estimates of the DMI population (a total of 46,400) and then divided by 6 to get an estimated annual number of new firm births. Existing establishments, used to compute new firm birth rates (births/1,000 establishments), are based on the census provided for each of Minnesota's counties in *County Business Patterns* for 1983.

# Table B.1. INITIAL DMI SAMPLE AND ESTIMATES OF DMI POPULATION: MINNESOTA EXCLUDING ST. PAUL

DHI SAMPLE LIST	SIC		Year	Start in	DMI File			Total
	Codes	1979	1980	1981	1982	1983	1984	
SUBSET I								
Agriculture	01-09							
Mining	10-14							
Manufacturing	20-39							
Distributive	40-51							
Services								
Business	60-67.73.							
Services	86.89							
H. Ed. Soc Services	80.82.83							
Total	,,	437	437	437	437	437	437	2 622
				101	101	101	101	6,066
SUBSRT II								
Construction	15-17							
Retail	52-59							
Consumer	70 72 75 76							
Services	78 79 84							
Total	10,10,04	146	146	146	146	146	140	070
10041		140	140	140	140	140	140	010
	F OF SAMPLING							
Subest I	s or philling	1 704	1 200	1 600	1 000	1 795	1 447	0.040
Subcot II		1,104	1,000	1,000	1,000	1,100	1,447	3,340 10 740
		2,229	2,010	2,100	2,111 2 704	2,193	1,903	12,140
CANDITNE THERDUALC (D	171001	3,920	3,100	3,014	3,194	3,928	3,410	22,692
Shirbing Inisavabo (a. Subast I	A1105)	2 00	2 00	2 00	0.05	9 07	0 04	0 70
Subset I		J.30	J.00	0.00 14.07	3.83	3.31	3.31	5.19
SUBSEC II		10.20	14.10	14.97	14.40	15.02	13.45	14.00
DWT CANDLE DV TUDDCTD	V							
Admiculture	T	26	95	96	01	10	10	100
Mining	1 T	20 2	23	20 0	21 0	12	10	120
Construction	1 11	27	4	U 20	4	U 05	3	11
Veryfraturing	11 T	) ( 111	19	23 00	10	23 110	104	140
Distributing	1	111	100	92 150	110	110	104	000
Distributive Services	I T	101	164	100	138	100	162	801
BUSINESS SERVICES	1	152	108	143	146	134	141	874
n, Mu, Soc Services	1	10	20	20	17	23	11	112
		11	90	90	98	100	101	561
Consumer Services	11	32 500	32	21	33	21	25	170
Totals		283	283	583	583	583	583	3,498
PARAT DUT DADIIIARTAN	(12-12-1-3)							
IUIAL DAI POPULATION	(Estimated)	101	0.7	100				
Agriculture		101	97	100	81	48	53	480
Bining	1	8	15	U	8	U	10	41
Construction	11	564	269	434	217	376	269	2,128
Manufacturing	1	433	386	356	435	468	344	2,423
Distributive Services	1 I	511	479	603	531	596	536	3,256
Business Services	1	593	610	553	562	532	467	3,317
H, Ed, Soc Services	1	58	100	77	65	91	36	429
Ketail	11	1,173	1,347	1,347	1,417	1,502	1,358	8,144
Consumer Services	11	487	454	404	477	315	336	2,474
Totals		3,928	3,758	3,874	3,794	3,928	3,410	22,692

-104-

Table B.2. INITIAL DMI SAMPLE AND ESTIMATES OF DMI POPULATION: ST. PAUL ONLY

DMI Sample List	SIC		Year	Start in	DMI File			Total
	Codes	1979	1980	1981	1982	1983	1984	
SUBSET I								
Agriculture	01-09							
Mining	10-14							
Manufacturing	20-39							
Distributive Services	40-51							
Business	60-67,73,							
Services	86,89							
H, Ed, Soc Services	80,82,83							
Total		125	123	125	125	125	124	747
SUBSET II								
Construction	15-17							
Retail	52-59							
Consumer	70.72.75.76.							
Services	78.79.84							
Total	,.,.,	42	42	42	42	42	12	252
		10	14	16	76	76	76	232
DHI POPULATION AT TIM	R OF SAMPLING							
Subset I		125	123	159	160	131	124	822
Subset II		116	113	132	132	142	144	770
		241	236	201	202	142 979	269	1 601
SAMPLING INTERVALS (R	ATTOS	611	200	601	272	210	200	1,001
Subset I		1 00	1 00	1 27	1 28	1 05	1 00	1 10
Subset II		2 76	2 60	3 14	2 14	1.00	3 43	2 00
		2.10	2.03	0.14	0.14	0.00	0.40	0.05
DHI SAMPLE BY INDUSTR	Y							
Agriculture	I	3	3	3	6	0	2	17
Mining	I	Ō	Û	0	Ō	1	0	1
Construction	ĪI	8	Ì	5	10	14	14	58
Manufacturing	I	31	23	24	31	30	27	166
Distributive Services	Ī	24	40	38	33	32	36	203
Producer Services	Ī	61	43	47	52	52	51	306
H. Rd. Soc Services	Ī	6	14	13	3	10	8	54
Retail	ĪI	23	30	29	23	20	21	146
Consumer Services	II	11	5	8	9	8	7	48
Totals		167	165	167	167	167	166	999
							100	
TOTAL DHI POPULATION	(Estimated)							
Agriculture	I	3	3	4	8	0	2	19
Mining	Ī	0	Û	0	0	1	Ū.	1
Construction	TT	22	19	16	31	47	48	183
Manufacturing	T	31	23	31	40	31	27	183
Distributive Services	I I	24	40	48	42	34	36	200
Producer Services	Ī	61	43	10 60	40 67	51	51	336
H Rd Soc Services	ī	ĥ	14	17	Å	10	91 8	500 50
Retail	II	6 Å	<u>81</u>	Q1	7 79	88	.79	53 1 1 1
Consumer Services	II	30	13	51 95	72 28	00 97	16 94	1/9
Totals	**	241	236	201	20	21	27 968	1 601
100410		e 11	200	441	6v6	410	200	11001

Table B.3. DEVELOPING ESTIMATES OF TOTAL NEW FIRM POPULATIONS

Industry Sector	Agri Serv	Mining	Const	Manufac	Dist	Bus	H,Ed,Soc	Retail	Cons	Total
Sampling Subset	I	Ι	II	I	berv I	serv I	Serv I	II	Serv II	
1979										
MN, w/o St.Paul DMI	101	. 8	564	433	511	503	50	1 179	407	0 000
St.Paul DMI	3	0 0	22	31	24	555		1,1()	401	3,928
Total DMI file	104	8	586	464	535	65.4	0 6 A	1 9 9 7	JU 517	241
Birch correction	2.4	1.7	1.7	1.6	1.5	28	28	1,201	) I C 2 Q	4,109
Total population	250	14	996	742	803	1 831	2.0	1.5	4.0 1 449	0 110
1980					000	1,001	110	1,000	1,440	0,110
MN, w/o St.Paul DMI	97	15	269	386	479	610	100	1 347	45.A	3 767
St.Paul DMI	3	0	19	23	40	43	14	81	13	236
Total DMI file	100	15	288	409	519	653	114	1,428	467	3,993
Birch correction	2.4	1.7	1.7	1.6	1.5	2.8	2.8	1.5	2.8	.,
Total population	240	26	490	654	779	1,828	319	2,142	1,308	7,785
1981								·		
MN, w/o St.Paul DMI	100	0	434	356	603	553	77	1,347	404	3.874
St.Paul DMI	4	0	16	31	48	60	17	91	25	292
Total DMI file	104	0	450	387	651	613	94	1,438	429	4,166
Birch correction	2.8	1.8	1.8	1.7	1.5	3.2	3.2	1.5	3.2	-,
Total population	292	0	810	658	977	1,962	301	2,157	1,373	8,529
1982								·		
MN, w/o St.Paul DMI	81	8	217	435	531	562	65	1,417	477	3,793
St.Paul DMI	8	0	31	40	42	67	4	72	- 28	292
Total DMI file	89	8	248	475	573	629	69	1,489	505	4.085
Birch correction	3.4	2.0	2.0	1.8	1.6	3.9	3.9	1.6	3.9	-,
Total population	303	16	496	855	917	2,453	269	2,382	1.970	9,661
1983									•	-,
MN, w/o St.Paul DMI	48	0	376	468	596	532	91	1,502	305	3,918
St.Paul DMI	0	1	47	31	34	54	10	68	27	272
Total DMI file	48	1	423	499	630	586	101	1,570	332	4,190
Birch correction	4.4	2.3	2.3	1.9	1.7	4.9	4.9	1.7	4.9	-,
Total population	211	2	973	948	1,071	2,871	495	2,669	1,627	10,868
1984								·		
MN, w/o St.Paul DMI	53	10	269	344	536	467	36	1,358	336	3,409
St.Paul DMI	2	0	48	27	36	51	8	72	24	268
Total DMI file	55	10	317	371	572	518	44	1,430	360	3,677
Birch correction	6.4	2.6	2.6	2.1	1.8	7.0	7.0	1.8	7.0	•
Total population	352	26	824	779	1,030	3,626	308	2,574	2,520	12,039
Six Year Total										
Total DMI file	500	42	2,312	2,605	3,480	3.653	486	8,592	2,610	24,280
Population estimate	1,647	83	4,589	4,637	5,575	14,572	1,871	13,780	10,244	56,999
Six Year Total Correcte Correction .5	d									
Total DMI file	250	21	1,156	1,303	1,740	1.827	243	4,296	1 305	12 140
Total population	824	42	2,294	2,318	2,787	7,286	936	6,890	5,122	28,499

Table B.4.	DEVE	LOPIN	NG E	STIM	IATES	S OF	NEW	FIR	M P	OPUL	ATIC	N BY	MIN	NESOTA	A REG	SION
Minn Region(a) DHI SAMPLE COUNT	1	2	3	4	5	6W	6 <b>E</b>	7 N	78	8	9	10	11	TC Part S	t.Paul	Total
Agri Services	3	0	2	10	4	1	0	5	1	2	1	7		11	15	0.0
Mining	Ő	Ó	2	1	Ō	Ō	1	ŏ	1	ñ	1	í		1	10	30
Construction	3	3	5	1	į	Ő	- Ā	12	i	3	5	20		75	1 69	202
Manufacturing	11	1	38	27	24	5	17	42	12	13	31	41		370	166	203
Dist Services	22	1	46	39	25	10	19	57	9	21	11	12		490	200	1 064
Bus Services	3	6	39	1	14	3	20	28	8	6	21	53		666	306	1 180
H, Ed, Soc Serv	0	1	8	3	5	1	6	1	Ō	Ō	5	12		64	54	1,100
Retail	19	12	49	. 35	32	9	14	46	23	24	36	60		202	146	707
Consumer Serv	4	2	21	8	1	3	5	14	7	6	7	21		65	48	218
DWT CLUDE CHE CLOSE	65	38	210	137	115	32	86	211	65	-75	154	287		1.980	997	4.452
DAI SAMPLING RATIO	S													-,	•••	1,102
Agri Services	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
dining Constant i	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
Construction	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55		14.55	3.09	
Diet Courting	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
Dist Services	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
Bus Services	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
L, BO, DOC DERV	3.19	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79	3.79		3.79	1.10	
Congran Com	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55		14.55	3.09	
TOTAL DET DELA	14.00 THIPPC	14.00	14.00	14.00	14.55	14.55	14.55	14.55	14.55	14.55	14.55	14.55		14.55	3.09	
Agri Commission	IDAIRS	٨														
Mining	11	U	ō	38	15	4	0	19	4	8	15	27	183	167	17	331
Construction	U AA		0 73	4	0	0	4	0	4	0	4	4	16	15	1	43
Mannfacturing	44	44	13	102	28	0	58	175	58	44	73	291	1,270	1,091	179	2,289
Dist Services	46	. 61	144	102	31	19	64	159	45	49	117	155	1,585	1,402	183	2,601
Bus Services	00	61 92	14	140	30	30	12	216	34	80	167	273	2,080	1,857	223	3,486
H. Rd Soc Serv	11	25	140	41	JJ 10	11	10	100	30	23	00	201	2,861	2,524	337	3,649
Retail	0 976	175	712	11	13	191	23	61	U 295	940	19	40	302	243	59	484
CODEBBER Serv	58	210	306	116	100 102	191	204	204	333	J43 07	324	013	3,390	2,939	451	8,614
Totals	526	107	1 603	1 057	800	250	573	404	104 610	10	1 100	300	1,034	340	148	2,622
CORRECTION(b)	50	421	1,000	1,001	033	200	313	1,014	012	033	1,100	4,114	12,102	11,104	1,398	24,118
Agri Services		0	1	19	8	2	٥	Q	,	1	8	12	92			166
Mining	Ő	Ŏ	i	2	Ő	ñ	2	Ň	2	r A	2	10	J L 9			100
Construction	22	22	36	51	29	ů	29	87	29	22	36	146	253			1 1 4 4
Manufacturing	21	13	72	51	45	9	32	80	23	25	59	78	792			1,111
Dist Services	42	13	87	74	47	19	36	108	17	40	. 83	136	1.040			1 743
Bus Services	6	11	74	13	27	6	38	53	-15	. 11	40	100	1.430			1,825
H, Ed, Soc Serv	0	2	15	6	9	2	11	13	0	0	9	23	151			242
Retail	138	87	356	255	233	65	102	335	167	175	262	437	1.695			4.307
Consumer Serv	29	15	153	58	51	22	36	102	51	- 44	51	153	547			1.311
LOW ESTIMATE	263	163	801	529	449	125	287	787	306	320	550	1.087	6.391			12.059
BIRCH CORRECTION(c	:)											•				,
Agri Services 3	.10 18	0	12	59	23	6	0	29	. 6	12	23	41	284			513
fining 1	.90 0	0	1	4	0	0	4	0	4	0	4	4	15			41
Construction 1	.90 41	41	69	97	55	0	55	166	55	41	69	276	1,207			2,175
danufacturing 1	.75 36	23	126	90	80	17	56	139	40	43	103	136	1,387			2,276
Ulst Services 1	.55 65	21	135	115	-73	29	56	167	26	62	129	211	1,612			2,702
Dus Services 3	.55 20	40	262	47	94	20	135	188	54	40	141	357	5,078			6,477
u, Ed, Soc Serv 3	.55 0	1	54	20	34	1	40	47	0	0	34	81	536			859
actall 1	.55 214	135	553	395	361	101	158	519	259	271	406	677	2,627			6,676
ULBUEER SERV 3	.55 103	52	542	207	181	11	129	362	181	155	181	542	1,942			4,654
ALGH ESTIMATE	498	319	1,760	1,032	901	258	633	1,618	625	624	1,090	2,325	14,689			26,371

(a)See Figure 1.6 for Minnesota region names.
(b)Reduced to compensate for DMI inaccuracies.
(c)Average of third and fourth year corrections of Birch and MacCracken.

. .

### Appendix C

### **RESEARCH PROCEDURE**

This project is a survey of new Minnesota businesses. As with any survey, it has several important stages:

- 1. Definition of the total population, the universe of new firms.
- 2. Selection of a sample to represent the population.
- 3. Determination of information to be gathered on each firm.
- 4. Collection of information from the sample.
- 5. Analysis of the information in relation to the most critical issues or hypotheses.

Although the outline is simple, the execution is somewhat complex, as there are major problems with each aspect of the procedure.

#### DEFINING THE POPULATION OF NEW FIRMS

Defining when a new firm exists is the first major conceptual problem. When is the attempt to initiate an economic enterprise considered to be "more than dreams and talk"? When incorporation occurs? When business cards are printed? When loans are sought? When income is first received? When the first employee is hired? Each criterion has its own set of problems in defining a population of new firms.

Each major source of lists of new firms reflects a different perspective. Legal incorporation is recorded on a regular basis, regardless of whether any economic activities ensue. Yellow page listings are also used in some analyses. Dun and Bradstreet credit ratings, reflecting loan applications, become the Dun's Market Identifier (DMI) file promoted as an "investment" for marketing strategies. Federal and state tax records can be used to identify "business income," often by full-time employees working as consultants or single individuals with several "businesses." The filing of unemployment insurance payments is used to count establishments and employees, but this excludes firms with no employees.

All of these approaches "overlook" businesses providing illegal services (drugs, gambling, sexual services, etc.) and those in the underground economy (legitimate activities performed for cash to avoid taxes, regulations, and legal liability). Further, some of the most complete lists are treated as confidential tax records, for example, those based on unemployment insurance payments or income tax records. The specific identities of these firms are not available for research projects--regardless of the procedures, sponsors, or purpose. Our final choice was to use a relatively complete, though biased, commercial listing and correct, as necessary, for known biases. Specifically, we used the Dun's Market Identifier file, a summary of establishment information provided for marketing purposes by Dun's Marketing Services. Most firms become part of the DMI file when they apply for credit and the lender (bank, supplier, etc.) insists on a Dun and Bradstreet credit report.

One major bias, unfortunately, is in the inclusion of new firms. The DMI file probably includes virtually all established firms, those over ten years old. Firms that require large amounts of credit, in manufacturing and whole-sale, are more likely to be included in the DMI file early in their history. Further, as the file is sold on a "per listing" basis (about sixty-five cents each), there is a vested interest in maintaining a large list of firms; the deletion of inactive firms may be slow and unsystematic.<sup>1</sup>

The DMI file information on each firm includes a "year start date"; initially defined as the date of birth, it is now described as including "changes in control"--changes in firm ownership. This information is the key criterion in locating new Minnesota firms.

The DMI lists about 4,000 Minnesota establishments with a start date for each of the recent years (1979-84). As the Dun and Bradstreet procedures tend to identify firms after several years, the number of firms in the preceding twelve months is usually quite low. For this analysis, new firms were defined as those with a Dun and Bradstreet "year start" date over a six-year period, 1979 through 1984.

To restrict the focus to only those firms that were initiated by private citizens, establishments listed as subsidiaries or branches of existing firms were excluded.

Three relevant descriptions are presented in Figure C.1. One is the distributions of Minnesota establishments for 1985 from *County Business Patterns* (CBP), a complete census. The second is the distribution of establishments with year start dates of 1979-84 estimated for the DMI file (taken from Appendix B, Table B.3). The third is the distribution of new firms represented by the sample that is the basis for the analysis in this report. This last has been adjusted to reflect the different sampling ratios related to the DMI file.

The three descriptions are very similar. There are only two industry sectors with "order of magnitude" differences: manufacturing and health, education, and social services. The new firm sample has about twice the proportion of manufacturing firms as in the population CBP estimate, and about one-third the proportion of health, education, and social services new firms as in the population estimate. This may reflect differential capture rates of new firms by Dun's DMI staff; they may be more likely to include new manufacturing firms that require substantial start-up capital and less likely to include new medical care professionals (the dominant proportion of the

<sup>&</sup>lt;sup>1</sup> Dun's Marketing Services representatives strongly deny this.

entities in the health, education, and social service sector) who may not require major funding to start a business.

These differences may also reflect the slightly lower proportion in business services. The increased DMI proportion in retail may reflect a higher founding rate; the lower sample proportion may reflect a reduced response rate for smaller retail outlets.

It is also possible that these differences may reflect real differences in new firm births, survival, and cooperation related to different industry sectors. For example, there may be a lower founding rate of new entities in health, education, and social services and a higher rate in retail. It is impossible to determine which of these is the most significant in accounting for these differences.

Fortunately, the differences are relatively small. It is appropriate to have confidence that the new firm sample, with appropriate weights, represents new firms that have emerged in Minnesota between 1979 and 1984.





#### SAMPLE: SELECTING RESPONDENTS FROM THE POPULATION

Potential respondents were identified in two steps. The first step was to select firms from the DMI file. This is described in detail in Appendix B. The second was to screen these firms in a brief phone interview and exclude those that were inappropriate.

Prior to the phone screening, the DMI listing was randomly divided into waves, each checked to ensure that all regions and industry sectors were represented. Each wave was then processed through the entire procedure. This technique allowed a complete coverage of the major factors in the event that financial resources prevented implementation of the original research design. This was the case in this project, for only 3,900 of the original 4,497 DMI listings could be processed.

#### QUESTIONNAIRE: INFORMATION OBTAINED ON EACH NEW FIRM

The questionnaire consisted of fifteen major sections:

- I. Products/Services
- II. Choice of Location
- III. Start-up Problems
- IV. Legal Form
- V. Start-up Resource Commitments
- VI. Outside Financial Support
- VII. Current Management Focus
- VIII. Strategic Focus
  - IX. Sales, Export, Asset History
  - X. Costs
  - XI. Employment
- XII. Start-up Team Background
- XIII. Future Plans
  - XIV. Contact with Programs, Services in Minnesota
  - XV. Final Comments

The full questionnaire is included in Appendix D. The codebook is available in the Methodological Appendix, a separate document.

Over 90 percent of all items were answered by the respondents. Many omissions were due to the lack of relevance. Omissions in sections dealing with sales history and production costs may have been due to a concern for confidentiality. In some instances the respondent did not have the time to gather the data requested on costs.

#### GATHERING DATA: PROCEDURE FOR DATA COLLECTION

The procedure for data collection had three major stages and a fourth, minor, one. The major stages were as follows:

1. All DMI listings were screened to identify those firms that were autonomous, new starts, in a business, and still viable. This phone contact was used to identify a suitable respondent--the person who had helped start the business and was still active in the management--as well as his or her correct mailing address.

- Questionnaires were mailed to the respondent: an initial mailing; a follow-up postcard; a second mailing; and, if necessary, a third mailing by certified mail.
- 3. After the three mailings, phone interviews were attempted with firms that had not responded. One-third of the material was deleted for these interviews (sections III, V, VII, and parts of XII). Attempts were made to complete phone interviews until the overall response rate was considered acceptable; some firms were contacted more than ten times by phone.
- 4. After questionnaires were received in the mail or interviews completed over the phone, those editing and coding the questionnaires in preparation for data entry made additional calls to the respondents to obtain critical "missing data," defined as related to sales, exports, or employment.

#### **RESULTS: SAMPLE SELECTION AND RESPONSE RATES**

The results of the data collection procedures are summarized in Table C.1. These operational results are compared with two other similar studies, Pennsylvania 1986 and Minnesota 1984. There is substantial loss between the DMI listings and the final sample, the first and major source of which is the 50 percent dropped because they were not suitable new firms, most because they were not new or could not be contacted by phone.

The overall response rate--the percentage of suitable respondents who provided information--was approximately 75 percent. This was slightly higher than the Pennsylvania study (67 percent) using an almost identical questionnaire and about the same as the earlier Minnesota study (76 percent), the questionnaire for which was four pages shorter. The nonrespondents are almost equally divided between explicit refusals and those failing to cooperate after repeated requests. This response rate can be compared with that obtained by the Bureau of the Census in a recent survey of business owners (*1982 Characteristics of Business Owners*. Washington, D.C.: U.S. Givernment Printing Office, 1987). A two-page form was mailed to 25,000 business owners selected from Internal Review Records and the front page bore the notice that "Response to this inquiry is required by law..." Their overall response rate was 81 percent.

Given the nature of this respondent population--very busy and anxious new firm principals--it is unlikely that a substantially higher response rate can be achieved unless there is a legal obligation to provide information or interviewers make personal visits to each respondent, which would be considerably more expensive.

Only because the interviewing staff was conscientious, gracious, and tenacious was such a high level of response achieved.

	Minne (Summ	sota er/	Pennsyl (Wint	vania er/	Pilot (Su	Study
	Fall 1	986)	Spring	1986)	19	84)
Random sample of firms from files of Dun's Marketing Services	3,000		4,600		1,245	•
Dropped if unable to verify in phone interview as new, autonomous, viable, or a firm	1,516	(51%)	2,289	(50%)	521	(42%)
Received three mailings of self-completion questionnaire and return envelope	1,484		2,311		724	
Returned self-completed questionnaire	665	(45%)	825	(36%)	398	(55%)
Failed to return questionnaire; approached for phone interview	819		1,486		326	
Completed phone interview	454	(55%)	724	(49%)	153	(47%)
Summary of Results						
Total completed interviews	1,119	(75%)	1,549	(67%)	551	(76%)
Total not completed	365	(25%)	762	(33%)	173	(24%)
Total initial sample	1,484	(100%)	2,311	(100%)	724	(100%)

Table C.1. SUMMARY OF RESULTS OF DATA COLLECTION PROCEDURE

### FINAL COMMENTS

It is possible to have high confidence that the sample represents new firms initiated by individuals (alone or as a group) in Minnesota between 1979 and 1984. This confidence is justified by the close match of the sample with a recent census of Minnesota establishments, the careful screening used to exclude ineligible firms (out of business, not new, or not a business), and the relatively high response rate.

Other research may provide different descriptions of the Minnesota economy, but it is unlikely that more representative data could be obtained on a voluntary basis from those initiating new firms. Appendix D

Questionnaire

### 1986 Minnesota New Firm Study



### **University of Minnesota**

This should be completed by a person that:

- Is active in the management of the firm AND
- Had a major responsibility for starting the firm.

This questionnaire is designed for a wide variety of firms. Please answer as many items as you can, skip the others.

### CONFIDENTIALITY AND ANONYMITY

- All information specific to individual firms will be kept confidential.
- The identify of all firms involved in the survey will remain anonymous.

Entry No:\_\_\_\_\_:\_\_\_:

ID No:\_\_\_\_:\_\_:\_\_:\_\_:\_\_:\_\_:\_\_:\_\_:\_\_:

### I — Products/Services

A. What is the major product or service provided by your firm?

B. What is special about your products or services that provides an advantage over your competitors?

C. In terms of current business, what is the mix of sales among principal products, product lines, or services?



A. What were the most important reasons for starting this firm in this city? county?

B. What were the most important reasons for choosing this specific site(s) for the firm?

C. A number of factors are required to produce goods and services. Please evaluate each of the following, as they currently exist, in relation to <u>satisfaction with</u> and <u>importance to</u> your firm.

	CIRCLE TWO NUMBERS FOR EACH FACTOR.	SAT Verv	ISFIED /		IMPORTANT Verv				
		: '	Son	newhat		Son	newhat		
		:	:	Not	:	:	Not		
		:	:	:	:	•	:		
1)	Labor costs	3	2	1	2	1	0		
2)	Availability of highly skilled workers	3	2	1	2	1	0		
3)	Education and training opportunities	3	2	1	2	1	0		
4)	Access to research and development facilities	3	2	1	2	1	0		
5)	Quality of life	3	2	1	2	1	0		
6)	Access to customers	3	2	1	2	1	0		
7)	Access to suppliers	3	2	1	2	1	0		
8)	Capital availability	3	2	1	2	1	0		
9)	Infrastructure (roads, water, sewers)	3	2	1	2	1	0		
10)	Transportation (highways, railroads)	3	2	1	2	1	0		
11)	Energy costs	3	2	1	2	1	0		
12)	Energy reliability	3	2	1	2	1	0		
13)	Taxes	3	2	1	2	1	0		
14)	Zoning and land use	3	2	1	2	1	0		
15)		<b>`</b>	•		0		•		
16)	Land purchase / rental cost	3	2	1	2	1	0		
17)	Building space availability	3	2	1	2	1	0		
18)	Building space expenditures (rent. etc)	3	2	1	2	1	õ		
19)	Local regulations	3	2	1	2	i	ŏ		
20)	Local government support for business	3	2	1	2	1	0		
21)	Other:	3	2	1	2	1	0		

# III — Startup Problems

For each type of problem, please indicate (by circling the correct response):

		(Status a or is, be	L SEVE as the fin eing esta	<b>RITY</b> m was, blished.)	CU (Cu rega	RREI rrent s ards to	NT STA situation this pr	ATUS with oblem.)
		Did Not Apply Never Occurred			Fu	lly S : I : S	olved Partiall Solved	у
		:	:	Minor		:	:	Not
		:	:	Proble	m Maior	:	÷	Sol-
			÷		Prob	•	÷	veu
A.	PRODUCTS AND MARKETS	:	•	÷	:	:	:	:
1)	Understanding industry trends	0	1	2	3	FS	PS	NS
2)	Analyzing competition, competitors	0	1	2	3	FS	PS	NS
3)	Finding competitive advantages	0	1	2	.3	FS	PS	NS
4)	Developing new, follow-on products/services	0	1	2	3	FS	PS	NS
5)	Providing after-sale follow-up or service	0	1	2	3	FS	PS	NS
6)	Understanding and assessing customer needs	0	1	2	3	FS	PS	NS
7)	Effective selling techniques	0	1	2	3	FS	PS	NS
8)	Writing advertising copy, selecting media	0	1	2	3	FS	PS	NS
9)	Providing customer service/follow-up	0	1	2	3	FS	PS	NS
10)	Pricing products/services	0	1	2	3	FS	PS	NS
11)	Delivering on time/within budget	0	1	2	3	FS	PS	NS
B.	FINANCIAL							
1)	Obtaining equity financing	0	1	2	3	FS	PS	NS
2)	Obtaining debt financing	0	1	2	3	FS	PS	NS
3)	Establishing a banking relationship	0	1	2	3	FS	PS	NS
4)	Developing an accounting & control system	0	1	2	3	FS	PS	NS
5)	Managing capital	0	1	2	3	FS	PS	NS
6)	Managing cash flow	0	1	2	3	FS	PS	NS
7)	Collecting accounts receivable; bad checks	0	- 1	2	3	FS	PS	NS
8)	Securing adequate financing to operate the new firm	0	1	2	3	FS	PS	NS
9)	Obtaining suitable real estate financing	0	1	2	3	FS	PS	NS
10)	Obtaining liability insurance	0	1	2	3	FS	PS	S NS

		INITIAL	. SEVE	RITY	CI	URREI	NT ST	ATUS	
	and size the	(Status as	s the firm	1 was,	(Current situation with				
COR	ase circle life rect responses )	Did No	t Anni	isneu.)	189	arus io	unis pro	olem.)	
0011			Never	у	E1	uny S · I	Partial	lv	
		:	Occuri	ed		: :	Solved	'y 	
		:	:	Minor		:	:	Not	
		÷	:	Proble	m	:	:	Sol-	
		÷	÷	÷	Majo Prob	r:	÷	ved	
C.	MANAGEMENT/ORGANIZATIONAL	:	:	:	:	÷	÷	:	
1)	Coping with government regulations	0	1	2	3	FS	PS	NS	
2)	Developing good relationship with union(s)	0	1	2	3	FS	PS	NS	
3)	Selecting a lawyer or accountant	0	1	2	3	FS	PS	NS	
4)	Motivating/compensating personnel	0	1	2	3	FS	PS	NS	
5)	Coordinating tasks among personnel/units	0	1	2	3	FS	PS	NS	
6)	Preparing a business plan	0	1	2	3	FS	PS	NS	
7)	Using/updating a business plan	0	1	2	3	FS	PS	NS	
8)	Setting goals/priorities for personnel	0	1	2	3	FS	PS	NS	
9)	Measuring performance against plans	0	1	2	3	FS	PS	NS	
10)	Clarifying goals/objectives	0	1	2	3	FS	PS	NS	
11)	Implementing plans, strategy	0	1	2	3	FS	PS	NS	
12)	Finding qualified managers, executives	0	1	2	3	FS	PS	NS	
13)	Finding qualified technical or professional staff	0	1	2	3	FS	PS	NS	
14)	Finding qualified employees	0	1	2	3	FS	PS	NS	
15)	Minimizing startup team conflict	0	1	2	3	FS	PS	NS	
D.	SELECTING/DEVELOPING A LOCATION								
1)	Identifying/selecting suitable site	0	_1	2	3	FS	PS	NS	
2)	Locating suitable rental space	0	1	2	3	FS	PS	NS	
3)	Appropriate transportation structure (e.g., rail, highway, public transportation)	0	1	2	3	FS	PS	NS	
4)	Appropriate infrastructure (e.g. sewer, energy sources, water, roads, etc.)	0	1	2	3	FS	PS	NS	
5)	Access to customers, clients	0	1	2	3	FS	PS	NS	
6)	Access to suppliers, vendors	0	1	2	3	FS	PS	NS	
7)	Access to employees	0	1	2	3	FS	PS	NS	

5

.

### A. What was the initial and is the current legal form of this new firm?

	INITIALLY (circle one)	NOW (Sum 86) (circle one)
Sole proprietorship	01	01
Partnership	02	02
Limited partnership	03	03
Subchapter S Corporation	04	04
Corporation	05	05
Franchise	06	06
Other	07	07
Don't know	08	08
Confidential	09	09

### V — Startup Resource Commitments

Month\_\_\_\_

\_ Year\_

\$\_\_\_\_\_,000

A. When did members of the startup team first begin to make major investments — personal time, personal resources — in the new firm?

TOTAL

B. BEFORE formal outside financing (any bank loans, stock placements, venture capital support, etc.) was obtained, how much financial support came from each of the following sources:
1) Personal savings
\$ \_\_\_\_\_\_,000

2)	Spouse or other members of immediate family	\$ 
3)	Relatives and other kin (parents, siblings)	\$ 
4)	Friends or Business Associates (Informally)	\$ 
5)	Salaries foregone by startup team members	\$ 
6)	Salaries foregone by employees	\$ ,000
7)	Credit from suppliers	\$ 
8)	Other (please list):	
		\$ 

Α.	When was the first outside financial support obtained?	None:	Month: .		_ Year:	
Β.	Please estimate the total financial a) when the first outside support	support from was received	each source and b) now.	twice:		
	(Amounts in \$1,000s)		VERY FIRS OUTSIDE S	T UPPOR	NOW T (Sum 86)	
1)	OUTSIDE Loans of any type: (Fror and loans, other regulated financia	m banks, savi al institutions)	ings			
	Working Capital or line of credit		\$	,000	\$	,000
	Term loans: Machinery & Equipme	nt	\$	,000	\$	,000
	Term loans: Inventory		\$	,000	\$	,000
	Term loans: Real estate loans		\$	,000	\$	,000,
	Other:		\$	,000	\$	_,000
2)	EXTERNAL Equity Investments:					
	Private Stock Placement		S	,000,	\$	_,000
	Public Stock Offering		\$	,000,	\$	_,000
	Venture Capital Firm(s)		\$	,000,	\$	_,000
	Other Equity Sources (specify):		\$	,000	\$	_,000
3)	Loans provided by state or local go	overnments?				
	(Specify)		\$	,000	\$	,000
4)	Employment, tax or other subsidies by the state or local governments	s provided				
	(Specify)		\$	,000,	\$	,000
5)	Federal government assistance?					
	SBA Loan or Loan Guarantee		\$	,000	\$	,000
	Other (specify):		\$	,000	\$	.,000
6)	Percentage of control/ownership re by original founders of the firm?	tained		_ %		_ %

VI — Outside Financial Support

C. What have been the major problems in obtaining financial support?

7

-121-

.

# VII — Current Management Focus

		Circ	le One	Number	for Ea	ach lte	em
A.	AT THE PRESENT TIME, TO WHAT DEGREE DO COMPANY MANAGEMENT, YOU AND	VERY :	MUCH Quite	A BIT			
	OTHER TOP EXECUTIVES	:	:	A MOD	ERATE	AMO	UNT
		:	:	:	LIIILE	NUNE	
		÷	:	÷	÷	:	CAN NOT
		:	:	:	:		EVALUATE
41			:	:	:	:	:
1)	clearly know your industry and market?	1	2	3	4	5	0
2)	Have technical experience in key areas?	1	2	3	4	5	0
3)	Have sufficiently well-rounded business experience?	1	2	3	4	5	0
4)	Have willingness to take necessary risks?	1	2	3	4	5	0
5)	Display high levels of energy and motivation?	1	2	3	4	5	0
6)	Have close customer contacts?	1	2	3	4	5	0
7)	Have formal written business and marketing plans?	1	2	3	4	5	0
8)	Regularly use, modify, and update plans?	1	2	3	4	5	0
9)	Set goals, priorities, and follow up to ensure they are attained?	1	2	3	4	5	0
10)	Accurately forecast operational results?	1	2	3	4	5	0
11)	Communicate goals and priorities to all	1	<b>.</b>	2	4	E	0
12)	Work together as a cohesive team?	1	2	ა ი	4	2 E	0
12)		1	2	J	4	5	Ū
B.	IN TERMS OF FINANCE AND BUDGETS, HOW MUCH DOES YOUR COMPANY						
1)	Have a sound financial control system?	1	2.	3	4	5	0
2)	Generate adequate cash flows from sales?	1	2	3	4	5	0
3)	Accurately forecast cash flow requirements?	1	2	3	4	5	0
4)	Have a sound cash flow position?	1	2	3	4	5	0
5)	Have strong support from the financial community of investors?	1	2	3	4	5	0
6)	Face the next few years with high certainty about its ability to survive?	. 1	2	3	4	5	0

Circle One Number for Each Item								
C.	IN TERMS OF PRODUCTS AND MARKETING,	VERY	MUCH QUITE	A BIT A MOD : I	ERATE LITTLE	AMO	UNT CAN NOT EVALUATE :	
	HOW MUCH DUES YOUR COMPANY							
1)	Have a clear market niche for its products and services?	1	2	3	4	5	0	
2)	Provide quality products and services?	1	2	3	4	5	0	
3)	Demonstrate ability to reach markets through its marketing/advertising activities?	1	2	3	4	5	0	
4)	Aggressively sell its products and services?	1	2	3	4	5	0	
5)	Have products or services that have a clear competitive advantage?	1	2	3	4	5	0	
6)	Produce its products/services on time and within budget?	1	2	3	4	5	0	
7)	Have an active program of new product development?	1	2	3	4	5	0	

## VIII — Strategic Focus

Please indicate the most important aspects of your competitive strategy.

(Cir	cle the best response for each.)	CRITICAL : IN			
			: A	AARGINA	L NSIGNIFICANT
1.	Lower orices	1	2	3	4
2.	Better service	- 1	2	3	4
3.	Quality products/services	1	2	3	4
4.	More choices	1	2	3	4
5.	Customize product/service to clients	1	2	3	4
6.	More effective marketing/advertising	1	2	3	4
7.	Fast response to changes in markets	1	2	3	4
8.	Serve those missed by others	1	2	3	4
9.	Superior location/customer convenience	. 1	2	3	4
10.	Distinctive goods/services	1	2	3	4
11.	Better, more attractive facilities	1	2	3	4
12.	More contemporary, attractive products	1	2	3	4
13.	Utilize new/advanced technology	1	2	3	4
14.	Develop new/advanced technology	1	2	3	4

### IX — Sales, Export, Asset History

- A. When did the firm receive its first sales income/revenue? Mth\_\_\_\_\_ Yr\_\_
- B. Please provide a history of sales/revenue and assets -- estimates are acceptable.

	TOTAL			PERCENT	AGE SOLD		YEAR-END			
	(in \$1,	000s)	Within Minn- esota	Adjacent States (IA,ND,SD,WI)	To Rest of U.S.	Outside the U.S.	TOTAL NET ASSET VALUE (in \$1,000s)			
985	\$	,000	%	%	%	%	\$	,000		
984	\$	,000	%	%	%	%	\$	.000		
983	\$	,000	%	%	%	%	\$	,000		
982	\$	,000	%	%	%	%	\$			
981	\$	,000	%	%	%	:5	s	.000		
980	\$	,000	%	%	%	%	3	,000		
979	\$	,000	%	%	%	%	S			
Pre- 1979 Aver	age)	,000	%	%	%	%	\$	,000		
Plea (# PUF	ase estima ALLOCAT RCHASES	te the amo E EACH I OF GOOI	ount spent f E <b>XPENSE</b> DS OR SEF	X CO or each expense TO ONLY ON RVICES:	<b>sts</b> ∋in 1985. IE CATEGO	DRY.)	(in \$	1,000)		
A P A	LL MATEF Sub-Con HYSICAL of physics NY BUSIN	RIAL COST tracts, Sup ASSETS ( al assets: E NESS OR F	TS: Raw Ma plies, Bulk COSTS: Pa Building, Ma PROFESSI	aterial, Parts, Su Purchases, ayments for leas achines, Equipm ONAL SERVIC	ubassemblie: e or purchase nent ES:	s, s e \$		000, _		
DEF	PRECIATI	on of CA	CCOUNTING, PITAL INV	ESTMENTS:		\$ \$		,000 ,000		
SAL T R A	ARIES AN RAINING ESEARCI LL OTHEN Fringe Be	ND WAGE: EXPENSE H AND DE' R COMPEI enefits	S (RELATE ES VELOPME NSATION:	ED TO) ENT Salaries, Wage	s, Bonuses,	\$ \$ \$		000, 000, 000,		
CO: P INT TAX	ST OF SA romotions EREST (ES	LES: Com	missions, C	Distribution, Adv	ertising,	\$ \$ \$		000, 000, 000,		

XI — Emplo	yment
------------	-------

Α.	When did the firm first hire ar — full or part time?	nybody None yet:	Month:	Year:
В.	Has locating and hiring individ or training been a major prob	duals with critical skills lem for the firm?	s NO:	YES:
	If yes, what types of skills or difficult to find?	r training or education	n, critical to the firm	n, are the most
	1	ng		:::
	2			:::
	3			:::
C.	What kinds of on-the-job of employees?	r formal training has	THIS FIRM been	providing new
	1			::_:
	2			:::
	3			:::
D.	For each of the following typ important type of RETRAINII firm's managers, staff, or em	pes of educational ins NG or ADDITIONAL E ployees.	stitutions, please in DUCATION it can	dicate the most provide for your
	Research Universities     (e.g., U.of Minn):			
	• Four Year Colleges		· · · · · · · · · · · · · · · · · · ·	· · ·
	• Two Year Colleges			· · · ·
	• Vocational/Technical Schools	6		::
	(e.g. AVTIs, private): • Other Sources of Training/ Education: (Please list)			:::
	1			:::
	2			:::
E.	List training needed, but not	available.		
	1			::
	2			:::

### Ε.

For each job category, please indicate:
The total involved during the first year of operation.
The total number involved during Summer 1986.
The major source of training, education for the typical employee.

Include all those receiving any cash or in-kind benefits from the company, INCLUDING YOURSELF, OTHER PRINCIPLES, and part time or temporary workers. Please include each employee in only one category.

	NUMBE	R EMPLOYE	D SOURCE OF TRAIL	VING, E		
	1st Yr	Sum 86	Extern External to You	al and Ir Firm	iternal :	:
1. -		E:	xecutives, administrators, and manager naring profit & loss responsibilities	s E	: E/I	:
<b>2</b>	. <u></u>	M Pi	anagers, supervisors without rofit and Loss responsibilities	E	E/I	ł
3. -		ad	perating Staff professionals (engineers, countants, lawyers, programmers, etc.)	E	E/I	ł
4.		R( (S te	esearch and Development Professional cientists, development engineers, lab chnicians, analysts, etc.)	s E	E/I	ł
5		Sa	ales/Marketing Representatives	E	E/I	I
		SKI ONLY S	LLED CATEGORIES SHOULD INCLUI 7 THOSE WITH A FORMAL APPRENT HIP, TRAINING, OR CERTIFICATION.	DE ICE-		
6.		SI of	killed Office Workers (Word processing perators, account clerks, paralegals,)	E	E/I	I
7		SI	killed Service Workers (Hairdresser, che	) E	E/I	I
8.		Si w	killed Craftsperson (Lathe operators, elders, etc.)	Е	E/I	1
9		0	peratives (Truck drivers, fry cook, etc.)	Е	E/I	I
10.		S	UNSKILLED ales Clerks	E	E/I	· Î
11		U	nskilled Office Workers (File clerk, etc.)	Е	E/I	I
12.		U cl	nskilled Service Workers (Waiter/waitre eaner, bellhop, etc.)	ss, E	E/I	I
13.		U ha	nskilled blue collar (Laborer, freight andler, baggers, etc.)	Е	E/I	I
14.		0	ther:	Е	E/I	I
		Т	OTAL EMPLOYEES			

### XII. Startup Team Background

When the first major effort was made to get this firm started, how many people made major commitments and expected to become part of the new firm's management team? Α.

Please select up to three others, in addition to yourself, for the following questions. If more than three others helped start the firm, select those most involved. Identify them by their initials.

	(Use code numbers when	1) You	2)	3)	4)
	appropriate.)				
В.	Gender: Male [01] Female [02]				
C.	Indicate any minority group with which this person might identify:Blacks[01]Hispanic[02]American Indian[03]Southeast Asian[04]Other Asian[05]Other minority[06]None[07]				
D.	Age when major commitment first made to new firm:				
E. Son Higl Tec Son Ass	Education completed when major commitment first made to new firm he Hi Sch [01] College:BA,BS [06] h School [02] Some Grad [07] h, Vocat'l [03] Masters Deg [08] he College [04] LLB, PhD, MD [09] oc. Deg. [05] Don't know [88]				
F.	Years of work between finishing school and starting this firm:		<u> </u>		1. 
G.	Number of positions held before starting this firm:	<u></u>			
H.	Number of years of work in industries similar to that of new firm:				
I.	How many other new firms has this person helped start:				
J.	How many hours per week does each devote to the firm?	<u> </u>			

13 -

	How would you depetibe the		1)	You	2	!)	3)	4)
κ.	career shift of each person to the new firm?							
	School to New firm	[01]						
	to New Firm	[02]						
	One New Firm to another New Firm	[03]						
	Unemployment to New Firm	[04]						
	Other	[05] [06]						
L.	Using the code letters from the list below, indicate the major reasons each me of the team got involved:	mber						
	Most important reason		_					
	Second most important							
	Third most important						<u> </u>	
	Fourth most important						. <u> </u>	
	<ul> <li>[A] The Challenge</li> <li>[B] To pursue an idea</li> <li>[C] Respect, Recognition</li> <li>[D] Build estate for family</li> <li>[E] Income, wealth</li> <li>[F] Self employment, automodia</li> </ul>	omy			(G) (H) (J) (L) (L)	Meet oth Utilize sk No better To live in Build an Contribu	er's expectatior ills, abilities alternatives the area organization te to society	IS
	[M] Other (1):							
	[N] Other (2):					, 		
	XI	11 —	- F	Futi	ıre	Plans	5	
М.	What are your business play	ns for	the	e next	2-3	years? (Cl	HECK ALL THA	t apply)
	1) Get out of the	busin	ess	i.				- · ·
	2) Change mix of	f prod	uct	s or s	ervic	es.		
• •	3) Significant IN(	CREAS	SE i	n emp	love	es.		

- Significant DECREASE in employees. 3)
- 4) Expansion of the firm. TO - City: \_\_\_ 5) \_State: .
- Relocation of the firm. TO City: \_\_\_\_ State: 6)
- 7) Sell the firm.
- No major changes. 8)

# XIV — Contact with Programs, Services in Minnesota

Please evaluate the interest in, or value of, the following existing and proposed services for your firm or business.

τ

	(Please circle your response.)	INTERES	T IN/VALUE	E OF THI	E SERVICE
		High	Moderate	Low	None
1. 2. 3. 4.	Federal procurement assistance Developing foreign export markets Tourism market development Energy development opportunities	3 3 3 3	2 2 2 2	1 1 1	0 0 0 0
5. 6. 7. 8.	New site locations within a county New site locations within Minnesota Small business incubator sites Retraining of existing employees	3 3 3 3	2 2 2 2	1 1 1	0 0 0 0
9. 10. 11. 12.	Training new employees Land acquisition financing Building construction financing Infrastructure (e.g. water, sewer, road) financing	3 3 3 3	2 2 2	1 1 1	0 0 0
13. 14. 15. 16.	Employee ownership financing Machinery, equipment financing Working capital financing Venture, seed capital financing	3 3 3 3	2 2 2 2	1 1 1	0 0 0 0
17. 18. 19. 20.	Joint R&D project development Applying company R&D in new markets New technology for productivity gain Energy audits/conservation projects	3 3 3 3	2 2 2 2	1 1 1	0 0 0 0
21. 22. 23. 24.	Entrepreneurial training General business management development Marketing skills development Engineering, scientific skill development	3 ·3 3 3	2 2 2 2	1 .1 1	0 0 0 0
25. 26. 27. 28.	Finance skill development Personnel management skill development Knowledge of government regulations Skills in labor-management relations	3 3 3 3	2 2 2 2	1 1 1 1	0 0 0 0
29. 30. 31.	Small Business Development Centers AVTI Small Business Programs Community College Small Business Program	3 3 s 3	222	1 1 1	0 0 0

B. PLEASE INDICATE, BY NUMBER, THOSE SERVICES YOUR FIRM HAS USED:

a)\_\_\_\_\_b)\_\_\_\_c)\_\_\_\_d)\_\_\_\_

### **XV — FINAL COMMENTS**

Do you have any comments, suggestions, or questions regarding the research project or Minnesota programs and services that assist business?

### THANK YOU VERY MUCH FOR YOUR HELP!!!

If the prepaid, addressed envelope is misplaced, please return to:

Minnesota New Firm Survey; Center for Urban and Regional Affairs (CURA); 330 Hubert H. Humphrey Center; University of Minnesota; 301 - 19th Avenue South; Minneapolis, MN 55455.

### Appendix E

### GENERAL DIMENSIONS OF NEW FIRMS STUDY: CONSTITUENT ITEMS

Among the data collected from those initiating new firms were five inventories related to an evaluation of or emphasis on different topics or issues. Included were:

9	Infrastructure importance	20	items
0	Start-up problems	43	items
0	Current management focus	25	items
۵	Competitive strategy	14	items
0	Existing, and proposed government services	31	items

It may be that the responses to such inventories reflect a smaller number of more general dimensions. If so, multiple-item indicators may be constructed to represent these dimensions.

A smaller set of more general dimensions not only facilitates analysis, by reducing the number of things for consideration, but may lead to more reliable indicators; more reliable indicators are those for which the same result occurs upon repeated measurements.

Factor analysis is a technique designed to assess the extent to which the same pattern of responses is reflected in two or more items. It is used to determine both the number of general dimensions and the items to be included within each dimension.

The results of the factor analysis for these five inventories are reproduced in Tables E.1-E.5. The general dimensions are ranked (from left to right) by the average value of the constituent items. Standard deviations (dispersion) are provided, as well as are estimated reliabilities; the latter cannot be computed for single-item dimensions, however.

All factor analyses and reliability estimates were computed using the appropriate programs from the SPSS-X omnibus statistical package. (See SPSS-X User's Guide, 2nd ed. [New York: McGraw-Hill, 1986].)

	LAND AL	ND SPA INFRAS : : : : : :	CE TRUCTUI LOCAL : : : : :	RE GOVERNI LABO : : : : :	MENT R ACCES : : :	SS R&D, T AND ED :	RAINING UCATION TAXES :
Land availability Land purchase/rental cost Building space availability Building space expenditures Infrastructure Transportation Energy costs Energy reliability Zoning and land use Local regulations Local govt. support for business Labor costs Availability of skilled workers Access to customers Access to suppliers Capital availability Educ. and training opportunities Access to R & D facilities Taxes	X X X	X X X X	X X X	X X	X X X X	X X	X
Number of items Standard deviation Mean importance (very = 2) Reliability	4 .64 1.07 .78	4 .60 1.20 .80	3 .62 1.17 .64	2 .66 1.37 .55	3 .46 1.58 .46	2 .67 .85 .67	1 .64 1.59

Table E.1. INFRASTRUCTURE IMPORTANCE DIMENSIONS
PR	ODU	CTS A	ND MA	RKETS	5				
	:	EMPLO	YEES						
	:	:	PLANN	ING					
	:	:	:	FINAN	ICING				
	:	:	:	:	ACCO	UNTING	;		
	:	:	:	:	:	ACCES	S		
	:	:	:	•	:	:	INFR/	ASTRUCTU	URE
· · · · ·	:	:	:	:	:	:	:	SITE	
	•	:	:	:	:	:	:	: (	GOVT
	:	•	•		•	•	•	•	•
Understand industry trends	x	•	•	•		•	•	•	•
Analyze competition	x								
Find competitive advantages	x								
Develop follow-on product/service	x								
Provide after-sale follow-up	x								
Assess customer needs	x								
Effective selling techniques	x								
Write ad conv select media	x								
Provide customer service	v								
Pricing products (services	v								
Deliver on time within hudget	x v								
Motivato (componente porconnel	л	v							
Coordinating tasks among units		N V							
Finding custing cases among units		A V							
Finding qualified technical staff		A V							
Finding qualified technical starr		X							
Finding qualified employees		X							
Minimizing start-up team conflict		X	37						
Preparing a business plan			X		`				
Using/updating a business plan			X						
Set goals/priorities for personnel			X						
Measuring performance against plans	5		X						
Clarifying goals			X						
Implementing plans/strategy			X						
Obtaining equity financing				· X					
Obtaining debt financing				Х					
Establishing a banking relationship	þ			X					
Secure financing to operate new fin	cm.			Х					
Develop accounting/control system					X				
Managing capital					X				
Managing cash flow					Х				
Collecting accounts receivable					Х				
Selecting an accountant or lawyer					Х				
Access to customers/clients						Х			
Access to suppliers/vendors						Х			
Access to employees						Х			
Appropriate transportation structur	re						Х		
Appropriate infrastructure							Х		
Identifying/selecting a site								Х	
Locate suitable rental space								Х	
Obtaining liability insurance									Х
Coping with regulations									Х
Number of items	11	6	. 6	4	5	3	2	2	2
Average severity (major = 3) 2	.00	1.93	1.91	2.03	2.00	1.65	1.50	1.85 1	.97
Standard deviation	. 38	.53	.47	.64	.46	.52	.57	.64	.56
Reliability	.77	. 82	. 92	. 83	. 74	.83	.85	.96	.57.

1

	PLAN : : : : : : : : : : : : : : : : : : :	NING FIN : : : : : : : : : : : : : : : : : : :	ANCIAL MARK : : : : : :	CONT ETING PRIN INVO : : :	ROL CIPAL LVEMEJ PROD SERV : : : :	NT UCT/ ICE EXP : :	ERIENCE BUSINESS EXPERIENCE :
Formal business and marketing plans Use, modify, and update plans Set goals, follow-up to see attained Accurate forecasts of operations Communicate goals to personnel Have a sound financial control system Generate adequate cash flows Accurately forecast cash flow needs Have a sound cash flow position Have strong support from investors Face the next years with confidence Market niche for products/services Ability to sell through marketing/ads Aggressively sell products/services Program of new product development Willing to take necessary risks High levels of energy & motivation Have close customer contacts Work together as a cohesive team Provide quality products and services Prod/Serv with competitive advantage Produce prod/serv on time, w/in budget Clearly know your industry and market Have technical experience in key areas Have well-rounded business experience		X X X X X X X	X X X	X X X X	X X X X	XXX	X
Number of variables Average emphasis (none = 5) Standard deviation Reliability	5 2.96 .97 .87	6 2.47 .88 .81	4 2.45 1 .83 .75	4 .88 1 .65 .73	3 .82 1 .66 .66	2 .86 .77 .75	1 2.16 .96

# Table E.3. CURRENT MANAGEMENT FOCUS DIMENSIONS

Table	E.4.	COMPETITIVE	STRATEGY	DIMENSIONS
-------	------	-------------	----------	------------

	Technology					
	: : May			rketing		
	:	:	:	Service		
	•	:	:	:		
	:	:	:	•		
More contemporary, attractive products	x					
Utilize new/advanced technology	Х					
Develop new/advanced technology	Х					
Better service		Х				
Quality products/services		Х				
Customize product/service to clients		Х				
Lower prices			Х			
More effective marketing/advertising			Х			
Fast response to changes in markets			Х			
Serve those missed by others			Х			
Superior location/customer convenience				X		
better, more attractive facilities				X		
Number of variables	З	3	4	2		
Average emphasis (high = 4)	2.69	1.63	2.36	2.58		
Standard deviation	.83	.61	.64	.91		
Reliability	.68	.64	.56	.72		

-135-

	TRAIN : : : : : :	VING A TECH : : : : :	ND EI INOLO( BUS	DUCAT GY AN SINES : C : :	ION D GR S DE APIT. : :	OWTH VELOPM AL FIN SITE L : :	ENT ANCING OCATIONS ENERGY :
Retraining of existing employees Training new employees Entrepreneurial training General business management development Marketing skills development Finance skill development Personnel management skill development Knowledge of government regulations Skills in labor-management relations Federal procurement assistance Developing foreign export markets Working capital financing Venture, seed capital financing Joint R&D project development Applying company R&D in new markets New technology for productivity gain Engineering, scientific skill developmen Small-business development centers AVTI small-business programs Community College Small Business Program Land acquisition financing Building construction financing Infrastructure financing Machinery, equipment financing Tourism market development New site locations within a county New site locations within Ainnesota Small business incubator sites Energy development opportunities Energy audits/conservation projects	X X X X X X X X X X X X X X X X X X X	X X X X X X X X X X	X X X	X X X X X	X X X X X	X X	
Number of variables Average interest (high = 2) Standard deviation Reliability	9 1.27 .83 .90	8 .77 1 .70 .84	3 1.12 .98 .89	4 . 89 . 85 . 77	4 .59 .67 .63	2 .56 .83 .75	

## Table E.5. EXISTING AND PROPOSED GOVERNMENT SERVICES DIMENSIONS

ŧ

#### Appendix F

### NEW FIRM CONTRIBUTIONS: ASSOCIATIONS WITH SELECTED FACTORS

The first stage of the analysis of factors associated with and, perhaps, preceding new firm contributions (jobs, sales, and out-of-state exports) was to consider the relationship of potential independent variables and new firm contributions. A total of 157 such variables were identified for preliminary analysis, the relationships are presented in the following two tables.

Two types of association are reported. If the potential independent variable was a nominal category (e.g., eight types of industry sectors, initial legal form), the differences in average contributions for different firms are reported in Table F.1. At the top of the table the characteristics of the frequency distributions of jobs, sales, and exports are described. It is clear they are all highly skewed distributions with a very substantial proportion of low values.

If the potential independent variable was interval or ratio, simple Pearson product moment correlations were computed. However, because the distribution of contributions (jobs, sales, and exports) was highly skewed toward the low end, the log transform was computed. The top of Table F.2 presents the characteristics of the frequency distributions of the transformed measures of contributions. As both kurtosis (peakedness) and skewness (asymmetry) for all three variables are close to zero, it is assumed the transformed frequencies are normally distributed. The number of new firms included in the 1985 export distributions is substantially reduced, to 377, because those with no exports were not included in the transformed distribution.

All correlations in Table F.2 utilize the log transform of the contributions. The statistical significance of measures of association is indicated as follows:

All analysis was done with a weighted sample to ensure the sample was representative of the population of new firms in Minnesota (as reflected in Dun's Market Identifier files). Hence, the number of new firms in the "number of cases" column represent this weighted value. When less than 1,119, it indicates an absence of data on the variables, usually the independent variable. Those cases with a dramatic drop in sample size usually reflect that portion of the questionnaire not completed in the phone interview, the source of data for about one-third of the unweighted sample or about one-half of the weighted sample.

The "variable names" are the identifiers used in the computer files and are included for the convenience of the authors.

# Table F.1. NEW FIRM CHARACTERISTICS AND AVERAGE CONTRIBUTIONS

TYPE OF CONTRIBUTION Units of Measure Mean (Average) Standard Deviation Minimum Maximum Kurtosis (Peakedness) Skewdness (Asymmetry) Valid Observations Missing Observations		•	1986 Jobs 9.13 13.95 .00 303.00 105.25 8.14 1,102 17	1985 Sales (\$1,000) \$450 1,353 \$0 \$25,000 130.25 9.59 921 198	1985 Exports (\$1,000) \$106 541 \$0 \$12,000 147.17 10.53 936 183
NEW FIRM CHARACTERISTICS [Average Contributions]	Variable Labels	Number of Cases	1986 Jobs	1985 Sales	1985 Exports
INDUSTRY SECTOR Agriculture Mining Construction Manufacturing Distributive Services Business Services Retail Consumer Services Health, Education, Social Services	DUNSIC2	18 1 109 141 185 176 321 125 30	3.70 2.00 8.60 12.30 7.80 9.80 9.00 6.90 14.70	87 65 445 703 1,055 467 419 193 339	2 22 24 254 259 121 15 6 46
LEGAL FORM, INITIAL Sole Proprietor Partnership Limited Partnership Subchapter S Corporation Corporation Franchise Other	IVA	429 150 18 168 317 1 3	4.70 7.80 15.10 13.80 12.80 60.00 9.00	224 294 225 764 976  8,704	22 26 57 146 258  0
LEGAL FORM, 1986 Sole Proprietor Partnership Limited Partnership Subchapter S Corporation Corporation Franchise Other	IVB	368 104 12 155 440 4 6	4.10 8.20 21.30 13.60 18.10 22.80 9.70	180 269 265 484 904 120 3,351	15 19 88 76 216 0
SALES DEVELOPMENT PROFILE Low Start, Low Growth High Start, Low Growth Low Start, High Growth High Start, High Growth	SDBVTYP	447 50 117 64	5.30 17.80 14.50 21.20	150 637 1,145 2,507	17 44 238 716
EXPORT ORIENTATION State Based Regional Exporter Wational Exporter	EXPSTAT	763 60 90	7.60 13.90 14.50	361 1,463 1,452	13 320 1,502

# Table F.2. CORRELATIONS BASED ON TRANSFORMED CONTRIBUTIONS

	1986	1985	1985
	Jobs	Sales	Exports
VARIABLES TRANSFORMED TO LOG (BASE 10)			
Variable Mames	Emplog	SALLOG	OUTLOG
Hean (Average)	.707	2.290	1.730
Standard Deviation	. 454	.619	.890
Minisus	.000	.000	.000
Maximum	2.480	4.450	4.080
Kurtosis (Peakedness)	214	.575	648
Skewdness (Asymmetry)	. 354	149	053
Valid Observations	1,094	919	377
Missing Observations	25	200	742

		CORRELATIONS WITH CO			
	Variable	Number	1986	1985	1985
	Label	of Cases	JOBS	SALES	EXPORTS
INFORMAL FINANCIAL RESOURCES					
Personal savings	VB1	607	.26***	.23***	.23**
Spouse/innediate family	VB2	609	.10**	.06	.07
Relatives or other kin	VB3	609	.04	.06	06
Friends/business associates	VB4	609	.11**	.09*	.16**
Start team salaries foregone	VB5	604	.17***	.11*	.15*
Employee salary foregone	<b>V</b> B6	609	.04	.03	.11
Credit from suppliers	VB7	608	.13***	.14**	.16*
Other Informal Support	<b>VB8</b>	609	.09	.10	.12
Total informal	AB3	608	.32***	.30***	.30***
START-UP PROBLEMS					
Products and markets	SUP1	550	.15***	.07	.04
Варіочев	SUP2	412	.21***	.15***	. 27**
Planning	SUP3	497	.14***	.11*	.13
Financing	SUP4	511	.19***	.17***	.29***
Accounting	SUP5	559	.14***	.14***	07
Ассевв	SUP6	434	.09***	.03	.13
Infrastructure	SUP7	312	.16**	.15**	.20*
Site	SUP8	429	.07	.08	.21**
Local government	SUP9	449	.23***	.20***	.34***
INFRASTRUCTURE INPORTANCE					·
Land and space	IMP1	1,061	.17***	.12***	.08
Infrastructure	IBP2	1,062	.10**	.04	03
Local government importance	IMP3	1,053	.14***	.06*	.00
Labor importance	IHP4	1,055	. 32	.21***	. 02
Access importance	IMP5	1,073	.04	01	10
R&D, quality of life	IMP6	1,066	.08**	.06*	.00
EMPLOYMENT, 1ST YEAR	IIF15	1,086	. 33***	.30***	.28***

# Table F.2./continued

			RRELATIO	NS WITH	CONTRIBUTION
	Variable	Number	1986	1985	1985
	Label	of Cases	JOBS	SALES	EXPORTS
RMPI.OYMRNT					
Professional & administrative	PROFFP	1 101	11***	01	07
Skilled percent initial	STILLEP	1 101	11***	.01	.07
Inskilled percent initial	NACTIP	1 101	11+++	.01	.07
Professional & administrative, nercent 1986	PROFEP	1 101	.11+++	.01	.01
Skilled percent, 1986	STILLCP	1 101	.04	.04	.00
Unskilled percent, 1986	NOSKLGP	1,101	.11***	.04	.07
SALRS GROWTH					
Years since 1st sales	SALAGE	1.029	06*	12***	. 02
Average sales growth, all years	ASALGRT	673	.30***	47***	45***
Average sales growth, year 1	ASALGR1	664	.23***	39***	41***
Average sales growth, year 2	ASALGR2	536	.38***	.54***	47***
Average sales growth, year 3	ASALGR3	394	.41***	.54***	48***
Average sales growth, year 4	ASALGR4	287	40***	.53***	48***
Average sales growth, year 5	ASALGR5	175	.39***	.55***	43***
Sales, 1st operational year	YRIR	858	20***	.24	.08
Sales, 1st operational year	YR4R	497	.45***	.60***	.50***
SALES					
Sales within MN. 1985	HRANT	854	34***	.50***	23***
Sales adjoining MN, 1985	NERAHT	894	23***	33***	47***
Sales rest of U.S., 1985	USANT	900	20***	30***	50***
Sales abroad, 1985	NLDANT	908	10**	11222	19###
Sales, percent in MN	P1	912	- 13***	- 19111	- 66111
Sales, percent in near states	P2	912	10**	12111	31###
Sales, percent in rest of U.S.	P3	910	1011	16111	50***
Sales, percent abroad	P4	910	02	.03	.12*
HANAGEMENT FOCUS					
Planning	NF1	596	- 21***	- 18***	21**
Financial control	NF2	614	- 14***	- 13##	- 24**
Marketing	NF3	596	- 15***	- 17***	- 24**
Principal involvement/commitment	HF4	612	09*	12**	.20**
Product/service	NF5	599	.02	.03	20**
Experience	HF6	612	06	15***	19**
STRATEGIC FOCUS					
Technology	SF1	1,072	.12***	.12***	07
Products	SF2	1,081	00	.05	09
Marketing	SF3	1,077	06	07*	07
Service	SP4	1,077	13***	04	.21***
RESPONDENTS MAJOR MOTIVES					
Challenge	GOAL1P	1,041	.07*	.03	.09
Income, wealth	GOAL5P	1,040	.10**	.06*	.13*
Autonomy	GOAL6P	1,038	12***	06*	.07
No alternatives	GOAL9P	1,041	09**	06*	01
Build an organization	GOAL11P	1,041	.14***	.12***	.15**

# Table F.2./continued

		CORRELATIONS WITH CONTRIBUT					
	Variable	Rusher	1986	1985	1985		
	Label	of Cases	JOBS	SALES	BIPORTS		
CRIDE TO BDIN ATIDIAEDDIAETAA							
START-UP TRAM CHARACTERISTICS							
Size of start team	XIIA	1,060	. 27***	.21***	.15**		
Average age	AGENEAN	1,046 ·	08**	.01	.11**		
Average years between school and work	AVWORK	1,050 -	09**	.01	.08		
Average number of previous positions	AVPOS	1,038	.14**	.07*	.04		
Average years in industry	AVININD	1,058	.00	.14***	.13*		
Average number of firms started	AVSTART	1,047	.14**	.09**	.17**		
Average number of hours per week	AVHOUR	1,052	.09**	.21***	.04		
Age range of principals	AGEGAP	1,045	.16***	.21***	.16***		
Percent of males on start team	BOYPC	1.060	.08**	.21***	23***		
Percent with high school education	EDHSPC	1.061	- 14**	- 07*	- 11#		
Percent with VoTech education	EDVOCPC	1,061 -	03	03	07		
Percent with 4-year degree	EDCOGPC	1,061	.16***	.11***	.14**		
Percent going from school to new firm	SCHNPPC	987 -	05	08**	01		
Percent established firm to new firm	ORGNPPC	987	.06	17***	09		
Percent unemployment to new firm	UNEFPC	987	- 07#	- 08##	- 12#		
Percent other to new firm	OTHNPPC	987	- 09##	- 17***	01		
Percent working 50+ hours/week	HARDWK	1 052	.00++ 09±±	24111	.01		
Percent working 60+ hours/week	OVPDWY	1 052	.03++	14+++	.00		
Percent 10+ years work experience	WPPVUT	1,002	.01+	.14+++	. 03		
Percent 10+ years experience in same industry	WERE 101	1,000 -	00+	.00	.11+		
referre in years experience in same industry	NEBAIDI	1,058	.00	.09**	.07		
FORMAL FINANCIAL SUPPORT, INITIAL							
Working capital/line of credit	VIF1	1.005	17***	18***	17**		
Term loans: machinery and equipment	VIF2	1 005	15222	17±±±	15##		
Term loans: inventory	VIR3	1 006	19***	20***	14+		
Term loans: real estate	VIPA	1 010	.14+++	. 20+++	.13+		
Private stock placement	VIPS	1 009	19+++	.01+	.00		
Public stock offering	VIP7	1 010	.14+++	.03++	.10		
Venture canital sources	VIPO	1,010	.03	. VI	.02		
Athar equity cources	VIED	1,010	. 04	.02	.01		
State/local dovergent loans	V123 V1210	1,010	.00++	.01	.03		
Coversent subsidies	7121V VTD44	1,010	.04	.05	.01		
	VIFII	1,007	.00	00	.01		
rederal Rovernment approximete	11212	1,009	.03	.10**	. 07		
FORMAL FINANCIAL SUPPORT, 1986							
Working capital/line of credit	VTW1	1 010	25***	25+++	75+++		
Tare loane: saching and equipment	VINI	1,010	. 20+++	. 40+++	. 40+++		
Tare loans: inventory	VIN2	1 014	.00	.10++	.114		
Torn loons, nool astate	V105 V104	1,014	.10+++	.13+++	.1044		
Deizete eteck placement	T185 .	1,014	. UD###	.014	.10*		
Private Block placement	TIND	1,012	.11***	.08¥	.10*		
rubiic stock oliering	VIR/	1,015	.07#	.02	.06		
Venture capital sources	VIAS	1,015	.04	.02	.07		
Uther equity sources	¥1 <b>N</b> 9	1,015	.05*	.01	.07		
State/local government loans	VINIO	1,015	.07*	.07*	.07		
Government subsidies	VIN11	1,015	.04	.01	.04		
Federal government assistance	VIN12	1,015	.03	.07*	.08		
Percent control retained, initial	VIP1	961 -	.12***	03	.03		
Percent control retained, 1986	VIP2	971 -	.03	06*	.02		

		CO	I CONTRIBUTI			
	Variable	Number	1986	1985	1985	
	Label	of Cases	JOBS	SALES	BIPORTS	
INITIAL START-UP PROBLEMS COUNT						
Products and markets	I SUP1	587	19111	17***	13±	
	ISUD2	509	37***	39***	37***	
Planning	TCUDS	500	.01+++	10+	11+	
Pipapoing	TCUDA	551		.10+	.11+	
Accounting	TCUDE	201	. 66+++	. 64444	. 61444	
Accounting	ISUPS	202	. 14***	.10***	00	
	ISUPO	484	.11***	.06	.16¥	
	150P7	312	.15***	.16**	.21*	
Site	ISUP8	429	.05	.20*	.15*	
Government	ISUP9	550	.26***	.23***	. 28***	
lotal	ISUPT	611	.35***	.33***	.30***	
* START-UP PROBLEMS, FULLY SOLVED						
Products and markets	PESI	547	- 10 <b>x</b>	01	- 10	
Reployees	PPC?	199	- 10	01	- 10	
Planning	PRCA	171	.10	.01	10	
r ranning Pinanoing	DRCA	111	.05	.UZ 16+++		
Accounting	DPCS	400 500	.00	.10+++	.00	
Accounting	PP 55	220	02	.02	.04	
ACCESS	PIS6	349	.11*	.14**	.10	
Infrastructure	PPS7	147	.06	.20*	.06	
Site	PFS8	306	.08	.01	. 02	
Government	PFS9	459	02	02	02	
Total	PIST	371	03	.06	.08	
THILLY CALVED STADT-TO PDADLENS CONNT						
Products and markets	PCCND1	567	03	0.5	07	
	PCCND2	J01 446	.03	.05	.01	
Dinning	PCCUDA	110	.00	.034	.03	
	FODULA	409	. 03	. UD	.00	
financing	FODULA	\$10	.12**	. 41+++	.05	
Accounting	FSSUPS	562	.00	.01	02	
Access	FSSUP6	364	.13**	.16**	.09	
Infrastructure	FSSUP7	157	.08	.15*	.07	
Site	FSSUP8	320	.09	.04	09	
Government	<b>F</b> SSUP9	477	. 07	.05	.04	
Total	FSSUPT	597	.12**	.16***	.14*	
INTEREST IN COVERNMENT PROCESS					-	
Training and education	GP1	655	.31***	. 18***	.12*	
Technology and growth	GP2	655	74111	19111	31***	
Rusiness development	CP3	654	07	- 03	- 03	
Capital fipancing	CPA	655	.01	00	00	
Cite leasting	202	000	12+++	.10+++	.13++	
Site locations	973 (DC	000	.13444	.00+	.00	
Energy	GP6	655	.13***	.06	.02	
COSTS, PERCENT OF TOTAL						
Baterials	PCHAT	588	09*	.16***	.06	
Physical assets	PCPHY	583	04	- 16***	. 02	
Professional services	PCPROF	580	05	- 1411	00	
Training	PCTDATE	500 690	00	- 04	18++	
persarch and development	DUDIN	503 Kac	.00	- NE	.10++	
	TURAU Drannd	030 E4V	.U6 15+++	03	. V I 19 -	
	ruuuar Doolt P	5/0	.13444	01	13*	
Jales	PUSALE	582	.07	.09	.01	
Interest	PUINT	573	10**	12**	19**	
Taxes	PCTAX	549	15***	21***	12*	

-142-



## **CENTER FOR URBAN AND REGIONAL AFFAIRS** 330 Hubert H. Humphrey Center 301 19th Avenue South Minneapolis, Minnesota 55455